# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

**WASHINGTON, DC 20549** 

# FORM 8-K

#### **CURRENT REPORT**

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of report (Date of earliest event reported): March 6, 2013

# ENCORE CAPITAL GROUP, INC.

(Exact Name of Registrant as Specified in Charter)

**Delaware** (State or Other Jurisdiction of Incorporation) 000-26489 (Commission File Number) 48-1090909 (IRS Employer Identification No.)

3111 Camino Del Rio North, Suite 1300, San Diego, California (Address of Principal Executive Offices)

92108 (Zip Code)

 $(877)\ 445\text{-}4581$  (Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of			
the following provisions ( see General Instruction A.2. below):			
X	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)		
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)		
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))		
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))		

#### Item 1.01. Entry into a Material Definitive Agreement.

Merger Agreement

On March 6, 2013, Encore Capital Group, Inc. (the "Company") entered into an Agreement and Plan of Merger (the "Merger Agreement") by and among the Company, Pinnacle Sub, Inc., a Delaware corporation and wholly owned subsidiary of the Company ("Merger Sub"), and Asset Acceptance Capital Corp., a Delaware corporation ("Asset"), pursuant to which Merger Sub will merge with and into Asset, and the separate corporate existence of Merger Sub will cease, and Asset will continue as the surviving corporation and wholly owned subsidiary of the Company (the "Merger"). The Merger Agreement has been approved by the board of directors of each of the Company and Asset.

Subject to the terms and conditions of the Merger Agreement, at the effective time of the Merger, each issued and outstanding share of Asset common stock will be converted automatically into, and will represent the right to receive, the following consideration:

Each share of Asset common stock with respect to which an election to receive cash (a "Cash Election") has been made will be converted, subject to adjustment in accordance with the terms of the Merger Agreement, into the right to receive an amount in cash equal to \$6.50 without interest (the "Cash Consideration").

Each share of Asset common stock with respect to which an election to receive stock consideration (a "Stock Election") has been made will be converted, subject to adjustment in accordance with the terms of the Merger Agreement, into the right to receive 0.2162 validly issued, fully paid and nonassessable shares of the common stock of the Company (together with any cash in lieu of fractional shares of Company common stock to be paid pursuant to the Merger Agreement, the "Stock Consideration"); provided, however, that no more than 25% of the issued and outstanding shares of Asset common stock may be exchanged for shares of the common stock of the Company (the "Maximum Stock Election") and any shares of Asset common stock elected to be exchanged for shares of the common stock of the Company above the Maximum Stock Election will be subject to proration as provided in the Merger Agreement.

Each share of Company Common Stock with respect to which neither a Cash Election nor a Stock Election has been effectively made, will be converted, subject to adjustment in accordance with the terms of the Merger Agreement, into the right to receive the Cash Consideration.

Pursuant to the terms of the Merger Agreement, each option to acquire shares of Asset common stock outstanding at the effective time of the Merger (whether or not then vested or exercisable) will be cancelled and terminated at the effective time of the Merger in exchange for the right to receive a cash amount equal to the product of (i) the total number of shares of Asset common stock subject to the option immediately prior to the effective time of the Merger *multiplied* by (ii) the excess, if any, of the Cash Consideration over the exercise price per share of Asset common stock underlying such option; provided, that if the exercise price per share of any such option is equal to or greater than the Cash Consideration, such option will be cancelled without any payment or other consideration being made in respect of such option. Each restricted stock unit and each deferred stock unit, with respect to shares of Asset common stock and outstanding at the effective time of the Merger, will be cancelled and entitle the holder thereof to receive from Asset, in full settlement of such equity award, a cash amount equal to the product of (i) the Cash Consideration and (ii) the total number of shares of Asset common stock subject to such restricted stock unit or deferred stock unit, as the case may be.

Consummation of the Merger is subject to the approval of Asset's stockholders. In addition, the Merger is subject to other customary closing conditions, including expiration or termination of the applicable waiting period under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended. Each party's obligations to complete the Merger is also subject to the accuracy of

representations and warranties of, and compliance with covenants by, the other party, in each case as set forth in the Merger Agreement.

The Company and Asset have made customary representations, warranties and covenants in the Merger Agreement, including, among other things, covenants by Asset (i) with respect to the conduct of its business during the interim period between the execution of the Merger Agreement and consummation of the Merger and (ii) not to engage in certain kinds of transactions during such period. Until March 27, 2013, the Company has agreed that Asset has the right to solicit, initiate, encourage and facilitate, whether publicly or otherwise, takeover proposals or to enter into discussions concerning, or provide confidential information in connection with, any such takeover proposal, in each case with certain previously identified parties; provided that following such date, Asset has agreed to not directly or indirectly solicit takeover proposals or, subject to certain exceptions with respect to any takeover proposal that is, or is reasonably likely to be, a superior proposal, to enter into discussions concerning, or provide confidential information in connection with, any alternative takeover proposal.

The Merger Agreement contains certain termination rights for both the Company and Asset, including, but not limited to, in the event that (i) the Merger is not consummated by August 7, 2013 (subject to an automatic extension to October 7, 2013 if all conditions other than those related to regulatory approvals and the absence of injunctions have been satisfied), (ii) Asset's stockholders do not adopt the Merger Agreement, (iii) an injunction permanently restraining, enjoining or otherwise prohibiting the consummation of the Merger becomes final and nonappealable, (iv) either party breaches its obligations or representations in a manner that cannot be cured by the termination date of the Merger Agreement, (v) Asset changes its recommendation to stockholders to adopt the Merger Agreement or breaches its agreement with respect to the solicitation of takeover proposals, or (vi) Asset grants a waiver or release, or determines not to enforce certain standstill agreements. The Merger Agreement further provides that, upon termination of the Merger Agreement under specified circumstances, Asset would be required to pay the Company a termination fee of up to approximately \$7.0 million.

The Merger Agreement contains representations and warranties that the parties have made to each other as of specific dates. Except for their status as contractual documents that establish and govern the legal relations among the parties to the Merger Agreement described therein, the Merger Agreement is not intended to be a source of factual, business or operational information about any of the parties thereto. The representations and warranties contained in the Merger Agreement were made only for purposes of such agreement and as of specific dates, were solely for the benefit of the parties to such agreement, and may be subject to limitations agreed between the parties, including being qualified by disclosures between those parties. The representations and warranties in the Merger Agreement may have been made to allocate risks among the parties thereto, including where the parties do not have complete knowledge of all facts, instead of establishing matters as facts. Furthermore, the representations and warranties may be subject to standards of materiality applicable to the contracting parties that differ from those applicable to investors. The assertions embodied in such representations and warranties are qualified by information contained in disclosure schedules to the Merger Agreement that the parties exchanged in connection with the execution of the Merger Agreement. Accordingly, investors and security holders should not rely on such representations and warranties as characterizations of the actual state of facts or circumstances, since they were only made as of the date of the Merger Agreement and are modified in important part by the underlying disclosure schedules. Moreover, information concerning the subject matter of such representations and warranties may change after the date of the Merger Agreement, which subsequent information may or may not be fully reflected in the Company's or Asset's public disclosures.

#### Voting Agreement

On March 6, 2013, AAC Quad-C Investors LLC (the "Supporting Stockholder") and the Company entered into a Voting Agreement (the "Voting Agreement") with respect to the Merger, which generally requires that the Supporting Stockholder, in its capacity as stockholder of Asset, vote all of its shares of Asset common stock in favor of the Merger and against any alternative transactions and generally prohibits the Supporting Stockholder from transferring its shares of Asset common stock prior to the consummation of the Merger. The Voting Agreement will terminate upon the earlier of (i) the consummation of the Merger, (ii) the termination of the Merger Agreement in accordance with its terms, or (iii) the making of any change, by amendment, waiver or other modification to any provision of the Merger Agreement that (x) decreases the amount or changes the form of the consideration or imposes any restrictions or additional conditions on the receipt of the consideration to the stockholders of the Company or (y) is otherwise materially adverse to the Supporting Stockholder, and which change, amendment, waiver or modification is not revised to the reasonable satisfaction of the Supporting Stockholder.

The Merger Agreement and Voting Agreement are attached hereto as Exhibit 2.1 and Exhibit 10.1, respectively, and are incorporated herein by reference. The descriptions of the Merger Agreement and the Voting Agreement set forth above do not purport to be complete and each is qualified in its entirety by reference to the provisions of the Merger Agreement and the Voting Agreement as applicable.

#### Item 8.01. Other Events.

On March 6, 2013, the Company issued a joint press release announcing the Merger. The press release is attached hereto as Exhibit 99.1. In addition, the Company hosted a conference call to discuss the Merger. A transcript of the conference call is attached hereto as Exhibit 99.2.

#### Item 9.01. Financial Statements and Exhibits.

#### (d) Exhibits.

Exhibit Number	<u>Description</u>
2.1	Agreement and Plan of Merger dated March 6, 2013, by and among Encore Capital Group, Inc., Pinnacle Sub, Inc. and Asset Acceptance Capital Corp.
10.1	Voting Agreement dated March 6, 2013, by and between Encore Capital Group, Inc. and AAC Quad-C Investors LLC.
99.1	Press Release issued by Encore Capital Group, Inc. on March 6, 2013.
99.2	Transcript of Conference Call held on March 6, 2013.

#### **Cautionary Statement Regarding Forward-Looking Statements**

This document contains forward-looking statements concerning the proposed transactions with Asset, the financial and business impact of such transactions, management's beliefs and objectives with respect thereto, and management's current expectations for future operating and financial performance, based on assumptions currently believed to be valid. Forward-looking statements are all statements other than statements of historical facts. The words "anticipates," "ran," "plans," "believes," "estimates," "expects," "projects," "intends," "likely," "will," "should," "to be," and any similar expressions or other words of similar meaning are intended to identify those assertions as forward-looking statements. It is uncertain whether the events anticipated will transpire, or if they do occur what impact they will have on the results of operations and financial condition of the Company, Asset or of the combined company. These forward-looking statements involve significant risks and uncertainties that could cause actual results to differ materially from those anticipated, including but not limited to the ability of the parties to satisfy the conditions precedent and consummate the proposed transactions, the timing of consummation of the proposed transactions, the ability of the parties to secure regulatory approvals in a timely manner or on the terms desired or anticipated, the ability of the Company to integrate Asset, the ability to implement the anticipated business plans following closing and achieve anticipated benefits and savings, and the ability to realize opportunities for growth. Other important economic, political, regulatory, legal, technological, competitive and other uncertainties are identified in the documents filed with the Securities and Exchange Commission (the "SEC") by the Company and Asset from time to time, including their respective Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, and Current Reports on Form 8-K. The forward-looking statements included herein are made only as of the date hereof. Neither the Company nor Asset undertakes any obligation to update the forward-looking statements included herein to reflect subsequent events or circumstances.

#### Additional Information about the Proposed Transactions and Where to Find It

This communication does not constitute an offer to sell or the solicitation of an offer to buy any securities or a solicitation of any vote or approval. In connection with the proposed transaction, the Company and Asset will file with the SEC a registration statement on Form S-4 that will include a proxy statement of Asset and a prospectus of the Company, as well as other relevant documents concerning the proposed transaction. STOCKHOLDERS OF ASSET ARE URGED TO READ THE REGISTRATION STATEMENT AND THE PROXY STATEMENT/PROSPECTUS REGARDING THE MERGER WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT THE COMPANY, ASSET AND THE PROPOSED TRANSACTION.

A free copy of the proxy statement/prospectus, as well as other filings containing information about the Company and Asset, may be obtained at the SEC's internet site (http://www.sec.gov). You will also be able to obtain these documents, free of charge, from the Company at www.encorecapital.com under the tab "Investors," and then under the heading "SEC Filings" or from Asset at www.assetacceptance.com under the tab "Investors," and then under the heading "SEC Filings." Copies of the proxy statement/prospectus can also be obtained, free of charge, by directing a request to the investor relations department of the Company or Asset at the following addresses:

Encore Capital Group, Inc. 3111 Camino Del Rio North, Suite 1300 San Diego, California 92108

Attention: Investor Relations/Director Finance & Treasury

Phone: (858) 560-2600

Email: adam.sragovicz@encorecapital.com

Asset Acceptance Capital Corp. 28405 Van Dyke Ave. Warren, Michigan 48093 Attention: Investor Relations Phone: (586) 983-7087

Email: marraf@assetacceptance.com

The Company, Asset and their respective directors and executive officers may be deemed to be participants in the solicitation of proxies in respect of the proposed transaction. Information about the Company's directors and executive officers is available in the Company's proxy statement dated April 27, 2012, for its June 6, 2012 Annual Meeting of Stockholders. Information about Asset's directors and executive officers is available in Asset's proxy statement dated March 28, 2012, for its May 10, 2012 Annual Meeting of Stockholders. Other information regarding the participants in the proxy solicitation and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the proxy statement/prospectus and other relevant materials to be filed with the SEC regarding the Merger when they become available. Investors should read the proxy statement/prospectus carefully when it becomes available before making any voting or investment decisions. You may obtain free copies of these documents from the Company or Asset using the sources indicated above.

This document shall not constitute an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the U.S. Securities Act of 1933, as amended.

#### **SIGNATURE**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ENCORE CAPITAL GROUP, INC.

Date: March 6, 2013

/s/ Paul Grinberg

Paul Grinberg

Executive Vice President, Chief Financial Officer and Treasurer

#### **EXHIBIT INDEX**

Exhibit Number	<u>Description</u>
2.1	Agreement and Plan of Merger dated March 6, 2013, by and among Encore Capital Group, Inc., Pinnacle Sub, Inc. and Asset Acceptance Capital Corp.
10.1	Voting Agreement dated March 6, 2013, by and between Encore Capital Group, Inc. and AAC Quad-C Investors LLC.
99.1	Press Release issued by Encore Capital Group, Inc. on March 6, 2013.
99.2	Transcript of Conference Call held on March 6, 2013.

#### AGREEMENT AND PLAN OF MERGER

Dated as of March 6, 2013

by and among

ENCORE CAPITAL GROUP, INC.,

PINNACLE SUB, INC.

and

ASSET ACCEPTANCE CAPITAL CORP.

## **TABLE OF CONTENTS**

Article   The Merger			Page
Section 1.1         The Merger         2           Section 1.2         Closing         2           Section 1.3         Effective Time         2           Section 1.4         Effects of the Merger         2           Section 1.5         Certificate of Incorporation and By-laws of the Surviving Corporation         3           Section 1.6         Directors and Officers of the Surviving Corporation         3           Section 1.7         Further Assurances         3           Article II Effect of the Merger on Capital Stock         3           Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 3.1         Organization, Standing and Corporate Power         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Auth	Article I The Merg	er	2
Section 1.2         Closing         2           Section 1.3         Effective Time         2           Section 1.4         Effects of the Merger         2           Section 1.5         Certificate of Incorporation and By-laws of the Surviving Corporation         3           Section 1.6         Directors and Officers of the Surviving Corporation         3           Section 1.7         Further Assurances         3           Article II Effect of the Merger on Capital Stock         3           Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 3.1         Organization, Standing and Corporate Power         11           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6			2
Section 1.3         Effective Time         2           Section 1.4         Effects of the Merger         2           Section 1.5         Certificate of Incorporation and By-laws of the Surviving Corporation         3           Section 1.6         Directors and Officers of the Surviving Corporation         3           Section 1.7         Further Assurances         3           Article II Effect of the Merger on Capital Stock         3           Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5	Section 1.2	Closing	2
Section 1.4         Effects of the Merger         2           Section 1.5         Certificate of Incorporation and By-laws of the Surviving Corporation         3           Section 1.6         Directors and Officers of the Surviving Corporation         3           Section 1.7         Further Assurances         3           Article II Effect of the Merger on Capital Stock         3           Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15	Section 1.3		2
Section 1.5         Certificate of Incorporation and By-laws of the Surviving Corporation         2           Section 1.6         Directors and Officers of the Surviving Corporation         3           Section 1.7         Further Assurances         3           Article II Effect of the Merger on Capital Stock         3           Section 2.1         Effect on Capital Stock         3           Section 2.3         Exchange of Certificates         6           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.7         Legal Proceedings         17	Section 1.4	Effects of the Merger	2
Section 1.6         Directors and Officers of the Surviving Corporation         3           Section 1.7         Further Assurances         3           Article II Effect of the Merger on Capital Stock         3           Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.7         Withholding Taxes         10           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.11         Enployee Benefits Matters         20           Section 3.11 </td <td>Section 1.5</td> <td></td> <td></td>	Section 1.5		
Section 1.7         Further Assurances         3           Article II Effect of the Merger on Capital Stock         3           Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits	Section 1.6		
Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.4         Governmental Approvals         15           Section 3.6         Absence of Certain Changes         17           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.10         Employee Benefits Matters         20           Section 3.12         Intell			
Section 2.1         Effect on Capital Stock         3           Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.4         Governmental Approvals         15           Section 3.6         Absence of Certain Changes         17           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.10         Employee Benefits Matters         20           Section 3.12         Intell	Article II Effect of	the Merger on Capital Stock	3
Section 2.2         Election Procedures         5           Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.6         Absence of Certain Changes         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.13         Absence of Right			
Section 2.3         Exchange of Certificates         6           Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.4         Governmental Approvals         15           Section 3.6         Absence of Certain Changes         17           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreem			
Section 2.4         Treatment of Equity Awards         9           Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         17           Section 3.6         Absence of Certain Changes         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.14         Property         23           Section 3.15         Contra			
Section 2.5         Adjustments         10           Section 2.6         Appraisal Rights         10           Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         11           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.4         Governmental Approvals         15           Section 3.6         Absence of Certain Changes         17           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.14         Property         23           Section 3.15         Contracts         23			
Section 2.6 Section 2.7         Appraisal Rights Withholding Taxes         10           Article III Representations and Warranties of the Company         11           Section 3.1 Organization, Standing and Corporate Power         12           Section 3.2 Capitalization         13           Section 3.3 Authority; Noncontravention         14           Section 3.4 Governmental Approvals         15           Section 3.5 Company SEC Documents; Undisclosed Liabilities         15           Section 3.6 Absence of Certain Changes         17           Section 3.7 Legal Proceedings         17           Section 3.8 Compliance With Laws; Permits         18           Section 3.9 Tax Matters         19           Section 3.10 Employee Benefits Matters         20           Section 3.11 Environmental Matters         21           Section 3.12 Intellectual Property         22           Section 3.13 Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14 Property         23           Section 3.15 Contracts         23           Section 3.16 Opinion of Financial Advisor         25           Section 3.17 Brokers and Other Advisors         25           Section 3.19 Share Ownership         25           Section 3.20 Employment Law Matters         26			
Section 2.7         Withholding Taxes         11           Article III Representations and Warranties of the Company         12           Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section			
Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.19         Share Ownership         25           Section 3.20			
Section 3.1         Organization, Standing and Corporate Power         12           Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.19         Share Ownership         25           Section 3.20	Article III Represei	ntations and Warranties of the Company	11
Section 3.2         Capitalization         13           Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.18         Company Stockholder Approval         25           Section 3.20         Employment Law Matters         26           Section 3.21         In			
Section 3.3         Authority; Noncontravention         14           Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.19         Share Ownership         25           Section 3.20         Employment Law Matters         26           Section 3.21         Insurance         26			
Section 3.4         Governmental Approvals         15           Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.18         Company Stockholder Approval         25           Section 3.20         Employment Law Matters         26           Section 3.21         Insurance         26			
Section 3.5         Company SEC Documents; Undisclosed Liabilities         15           Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.18         Company Stockholder Approval         25           Section 3.19         Share Ownership         25           Section 3.20         Employment Law Matters         26           Section 3.21         Insurance         26			
Section 3.6         Absence of Certain Changes         17           Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.18         Company Stockholder Approval         25           Section 3.19         Share Ownership         25           Section 3.20         Employment Law Matters         26           Section 3.21         Insurance         26			
Section 3.7         Legal Proceedings         17           Section 3.8         Compliance With Laws; Permits         18           Section 3.9         Tax Matters         19           Section 3.10         Employee Benefits Matters         20           Section 3.11         Environmental Matters         21           Section 3.12         Intellectual Property         22           Section 3.13         Absence of Rights Agreement; Anti-Takeover Provisions         23           Section 3.14         Property         23           Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.18         Company Stockholder Approval         25           Section 3.19         Share Ownership         25           Section 3.20         Employment Law Matters         26           Section 3.21         Insurance         26			
Section 3.8Compliance With Laws; Permits18Section 3.9Tax Matters19Section 3.10Employee Benefits Matters20Section 3.11Environmental Matters21Section 3.12Intellectual Property22Section 3.13Absence of Rights Agreement; Anti-Takeover Provisions23Section 3.14Property23Section 3.15Contracts23Section 3.16Opinion of Financial Advisor25Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.9Tax Matters19Section 3.10Employee Benefits Matters20Section 3.11Environmental Matters21Section 3.12Intellectual Property22Section 3.13Absence of Rights Agreement; Anti-Takeover Provisions23Section 3.14Property23Section 3.15Contracts23Section 3.16Opinion of Financial Advisor25Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.10Employee Benefits Matters20Section 3.11Environmental Matters21Section 3.12Intellectual Property22Section 3.13Absence of Rights Agreement; Anti-Takeover Provisions23Section 3.14Property23Section 3.15Contracts23Section 3.16Opinion of Financial Advisor25Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.11Environmental Matters21Section 3.12Intellectual Property22Section 3.13Absence of Rights Agreement; Anti-Takeover Provisions23Section 3.14Property23Section 3.15Contracts23Section 3.16Opinion of Financial Advisor25Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.12Intellectual Property22Section 3.13Absence of Rights Agreement; Anti-Takeover Provisions23Section 3.14Property23Section 3.15Contracts23Section 3.16Opinion of Financial Advisor25Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.13Absence of Rights Agreement; Anti-Takeover Provisions23Section 3.14Property23Section 3.15Contracts23Section 3.16Opinion of Financial Advisor25Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.14       Property       23         Section 3.15       Contracts       23         Section 3.16       Opinion of Financial Advisor       25         Section 3.17       Brokers and Other Advisors       25         Section 3.18       Company Stockholder Approval       25         Section 3.19       Share Ownership       25         Section 3.20       Employment Law Matters       26         Section 3.21       Insurance       26			
Section 3.15         Contracts         23           Section 3.16         Opinion of Financial Advisor         25           Section 3.17         Brokers and Other Advisors         25           Section 3.18         Company Stockholder Approval         25           Section 3.19         Share Ownership         25           Section 3.20         Employment Law Matters         26           Section 3.21         Insurance         26			
Section 3.16Opinion of Financial Advisor25Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.17Brokers and Other Advisors25Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.18Company Stockholder Approval25Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.19Share Ownership25Section 3.20Employment Law Matters26Section 3.21Insurance26			
Section 3.20 Employment Law Matters 26 Section 3.21 Insurance 26			
Section 3.21 Insurance 26			

## **TABLE OF CONTENTS (CONT'D)**

Page

A rti	cle IV Renresentat	tions and Warranties of Parent and Merger Sub	27
an u	Section 4.1	Organization, Standing and Corporate Power	27
	Section 4.2	Capitalization	28
	Section 4.3	Authority; Noncontravention	29
	Section 4.4	Governmental Approvals	29
	Section 4.5	Parent SEC Documents; Undisclosed Liabilities	30
	Section 4.6	Absence of Certain Changes	32
	Section 4.7	Legal Proceedings	32
	Section 4.8	Compliance With Laws; Permits	32
	Section 4.9	Contracts	33
	Section 4.10	Parent Common Stock	33
	Section 4.11	Brokers and Other Advisors	33
	Section 4.12	Ownership and Operations of Merger Sub	33
	Section 4.13	Sufficient Funds	33
	Section 4.14	Share Ownership	34
	Section 4.15	No Other Representations or Warranties	34
Arti	cle V Covenants		34
	Section 5.1	Conduct of Business	34
	Section 5.2	Preparation of the Proxy Statement/Prospectus; Stockholders Meeting	38
	Section 5.3	Solicitation; Change in Recommendation	39
	Section 5.4	Reasonable Best Efforts	44
	Section 5.5	Public Announcements	46
	Section 5.6	Access to Information; Confidentiality	46
	Section 5.7	Notification of Certain Matters	47
	Section 5.8	Indemnification and Insurance	47
	Section 5.9	Transaction Litigation	49
	Section 5.10	Section 16	49
	Section 5.11	Employee Matters	50
	Section 5.12	Merger Sub and Surviving Corporation	51
	Section 5.13	No Control of Other Party's Business	51
	Section 5.14	Takeover Laws	52
	Section 5.15	Reservation of Shares; Stock Exchange Listings	52
Arti	cle VI Conditions l		52
	Section 6.1	Conditions to Each Party's Obligation to Effect the Merger	52
	Section 6.2	Conditions to Obligations of Parent and Merger Sub	53
	Section 6.3	Conditions to Obligations of the Company	53
	Section 6.4	Frustration of Closing Conditions	54
Arti	cle VII Terminatio	on	54
	Section 7.1	Termination	54
	Section 7.2	Effect of Termination	56
	Section 7.3	Termination Fee	56

# TABLE OF CONTENTS (CONT'D)

			Page
rti	cle VIII Miscellane	eous	58
	Section 8.1	No Survival of Representations and Warranties	58
	Section 8.2	Fees and Expenses	58
	Section 8.3	Amendment or Supplement	58
	Section 8.4	Waiver	58
	Section 8.5	Assignment	59
	Section 8.6	Counterparts	59
	Section 8.7	Entire Agreement; Third-Party Beneficiaries	59
	Section 8.8	Governing Law; Jurisdiction	59
	Section 8.9	Specific Enforcement	60
	Section 8.10	WAIVER OF JURY TRIAL	60
	Section 8.11	Notices	60
	Section 8.12	Severability	61
	Section 8.13	Definitions	62
	Section 8.14	Interpretation	71

# **EXHIBITS**

Exhibit A - Certificate of Incorporation & By-laws of Merger Sub

#### AGREEMENT AND PLAN OF MERGER

This AGREEMENT AND PLAN OF MERGER, dated as of March 6, 2013 (this "<u>Agreement</u>"), is entered into by and among Encore Capital Group, Inc., a Delaware corporation ("<u>Parent</u>"), Pinnacle Sub, Inc., a Delaware corporation and a wholly-owned Subsidiary of Parent ("<u>Merger Sub</u>"), and Asset Acceptance Capital Corp., a Delaware corporation (the "<u>Company</u>"). All capitalized terms used in the Agreement shall have the respective meanings set forth in <u>Section 8.13</u>.

#### WITNESSETH

WHEREAS, it is proposed that Merger Sub be merged with and into the Company, in accordance with the General Corporation Law of the State of Delaware (the "<u>DGCL</u>"), with the Company continuing as the surviving corporation and a wholly-owned Subsidiary of Parent (the "<u>Merger</u>"), on the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Company Board, acting upon the unanimous recommendation of a review committee of independent directors of the Company (the "Review Committee"), has, on the terms and subject to the conditions set forth in this Agreement, unanimously (with one director not present) (a) approved the execution and delivery of and performance under this Agreement and the Voting Agreement, (b) resolved that the transactions contemplated by this Agreement and the Voting Agreement (including the consummation of the Merger upon the terms and subject to the conditions set forth in this Agreement and in accordance with the relevant provisions of the DGCL) (the "Transactions"), are advisable, fair to and in the best interests of its stockholders, (c) resolved to recommend that the Company stockholders adopt this Agreement and (d) directed that this Agreement be submitted to the Company's stockholders for their approval and adoption in accordance with applicable law;

WHEREAS, the respective boards of directors of Parent and Merger Sub have each, on the terms and subject to the conditions set forth in this Agreement, unanimously (a) resolved that the Transactions are advisable, fair to and in the best interests of their respective stockholders and (b) approved the execution and delivery of and performance under this Agreement, the Voting Agreement and the Transactions; and

WHEREAS, Parent, Merger Sub and the Company desire to make certain representations, warranties, covenants and agreements specified herein in connection with this Agreement; and

WHEREAS, immediately prior to the execution and delivery of this Agreement, and as a condition and inducement to the willingness of Parent and Merger Sub to enter into this Agreement, AAC Quad-C Investors LLC (the "Supporting Stockholder") has delivered to the Company, Parent and Merger Sub a voting agreement (the "Voting Agreement"), dated as of the date hereof, providing that the Supporting Stockholder, among other things, agrees to vote in favor of the Merger, each on the terms and subject to the conditions set forth in the Voting Agreement.

NOW, THEREFORE, in consideration of the representations, warranties, covenants and agreements contained in this Agreement and other good and valuable consideration, the receipt

and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, Parent, Merger Sub and the Company hereby agree as follows:

#### ARTICLE I

#### THE MERGER

**Section 1.1** <u>The Merger</u>. Upon the terms and subject to the conditions set forth in this Agreement, and in accordance with the DGCL, at the Effective Time, Merger Sub shall be merged with and into the Company, and the separate corporate existence of Merger Sub shall thereupon cease, and the Company shall be the surviving corporation in the Merger (the "<u>Surviving Corporation</u>").

**Section 1.2 Closing**. The closing of the Merger (the "Closing") shall take place at the offices of Kirkland & Ellis LLP, 601 Lexington Avenue, New York, New York 10022 at 10:00 a.m. (local time) on the date that is two (2) Business Days following the satisfaction or waiver (to the extent permitted by applicable Law) of the conditions set forth in Article VI (other than those conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of those conditions at such time), or such other date, time or place as is agreed to in writing by the parties hereto. The date on which the Closing occurs is referred to in this Agreement as the "Closing Date."

Section 1.3 Effective Time. Subject to the provisions of this Agreement, at the Closing the parties hereto shall file with the Secretary of State of the State of Delaware a certificate of merger, executed in accordance with, and in such form as is required by, the relevant provisions of the DGCL with respect to the Merger (the "Certificate of Merger"). The Merger shall become effective upon the acceptance of the Certificate of Merger by the Secretary of State of the State of Delaware or at such later time as is agreed to in writing by the parties hereto and specified in the Certificate of Merger (the time at which the Merger becomes effective is herein referred to as the "Effective Time").

**Section 1.4** Effects of the Merger . The Merger shall have the effects set forth in the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time, all the properties, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Corporation, and all debts, liabilities and duties of the Company and Merger Sub shall become the debts, liabilities and duties of the Surviving Corporation.

Section 1.5 <u>Certificate of Incorporation and By-laws of the Surviving Corporation</u>. At the Effective Time, the certificate of incorporation and by-laws of the Surviving Corporation shall be amended and restated in their entirety to be in the form of the certificate of incorporation and by-laws of Merger Sub, attached hereto as Exhibit A, except that the name of the Surviving Corporation shall at the Effective Time be changed to "Asset Acceptance Capital Corp." and as so amended shall be the certificate of incorporation and by-laws of Surviving Corporation until thereafter amended as provided therein or by applicable Law (and subject to <u>Section 5.8</u>).

Section 1.6 <u>Directors and Officers of the Surviving Corporation</u>. Each of the parties hereto shall take all necessary action to cause the directors and officers of Merger Sub immediately prior to the Effective Time to be the directors and officers of the Surviving Corporation immediately following the Effective Time, until their respective successors are duly elected or appointed and qualified or their earlier death, resignation or removal in accordance with the certificate of incorporation and by-laws of the Surviving Corporation.

Section 1.7 Further Assurances. After the Effective Time, if the Surviving Corporation shall determine or shall be advised that any deeds, bills of sale, instruments of conveyance, assignments, assurances or any other actions are necessary or desirable to vest, perfect or confirm of record in the Surviving Corporation its right, title or interest in, to or under any of the rights, properties or assets of either of the Company or Merger Sub acquired or to be acquired by the Surviving Corporation as a result of, or in connection with, the Merger or otherwise to carry out this Agreement, in each case, in accordance with the terms hereof, then the officers and directors of the Surviving Corporation shall be authorized to execute and deliver, in the name and on behalf of either the Company or Merger Sub, all such deeds, bills of sale, instruments of conveyance, assignments and assurances and to take and do, in the name and on behalf of each of such corporations or otherwise, all such other actions as may be necessary or desirable to vest, perfect or confirm any and all right, title or interest in, to and under such rights, properties or assets in the Surviving Corporation or otherwise to carry out the terms of this Agreement.

#### ARTICLE II

#### EFFECT OF THE MERGER ON CAPITAL STOCK

**Section 2.1** Effect on Capital Stock. At the Effective Time, by virtue of the Merger and without any action on the part of the holder of any shares of common stock, par value \$0.01 per share, of the Company ("Company Common Stock"), any other securities of the Company or any shares of capital stock or other securities of Merger Sub or, except as expressly set forth herein, on the part of the Company or Merger Sub:

(a) <u>Capital Stock of Merger Sub</u>. Each issued and outstanding share of capital stock of Merger Sub shall be converted into and become one validly issued, fully paid and nonassessable share of common stock, par value \$0.01 per share, of the Surviving Corporation, and shall thereupon constitute the only outstanding shares of capital stock of the Surviving Corporation. Each certificate evidencing ownership of such shares of common stock of Merger Sub shall thereafter evidence ownership of shares of common stock of the Surviving Corporation.

(b) <u>Cancellation of Treasury Stock and Parent-Owned Stock</u>. Any shares of Company Common Stock that are owned by the Company as treasury stock, and any shares of Company Common Stock owned by Parent or Merger Sub or any of their respective Subsidiaries, shall be automatically cancelled and shall cease to exist and no consideration shall be delivered in exchange therefor.

- (c) <u>Conversion of Company Common Stock</u>. Each issued and outstanding share of Company Common Stock (other than shares to be cancelled in accordance with <u>Section 2.1(b)</u> and Dissenting Shares) shall thereupon be converted automatically into, and shall thereafter represent the right to receive, the following consideration (collectively, the "<u>Merger Consideration</u>"):
  - (i) Each share of Company Common Stock with respect to which an election to receive cash (a "<u>Cash Election</u>") has been effectively made and not revoked pursuant to <u>Section 2.2</u> shall be converted, subject to <u>Section 2.5</u>, into the right to receive an amount in cash equal to \$6.50 without interest (the "<u>Cash Consideration</u>").
  - (ii) Each share of Company Common Stock with respect to which an election to receive stock consideration (a "Stock Election") is properly made and not revoked pursuant to Section 2.2 (each, a "Stock Electing Company Share") shall be converted, subject to Section 2.3(e) and Section 2.5, into the right to receive 0.2162 validly issued, fully paid and nonassessable shares of Parent Common Stock (together with any cash in lieu of fractional shares of Parent Common Stock to be paid pursuant to Section 2.3 (e), the "Stock Consideration"); provided, however, that no more than twenty-five percent (25%) of the issued and outstanding shares of Company Common Stock may be exchanged for shares of Parent Common Stock (the "Maximum Stock Election") and any shares of Company Common Stock elected to be exchanged for shares of Parent Common Stock above the Maximum Stock Election shall be subject to proration as provided in Section 2.1(d) below.
  - (iii) Each share of Company Common Stock with respect to which neither a Cash Election nor a Stock Election has been effectively made (including withdrawn elections pursuant to <u>Section 2.2</u>) (each, a "<u>Non-Electing Company Share</u>"), shall be converted, subject to Section 2.5, into the right to receive the Cash Consideration.
- (d) <u>Proration</u>. If holders of Company Common Stock elect to receive Stock Consideration in excess of the Maximum Stock Election, then the number of shares exchanged for Stock Consideration by each stockholder so electing will be reduced, such that the aggregate number of shares of Company Common Stock exchanged by such stockholder for shares of Parent Common Stock equals the product determined by multiplying (i) the Maximum Stock Election by (ii) the ratio that the number of shares elected to be exchanged by such stockholder bears to the total number of shares elected to be exchanged for shares of Parent Company Stock by all Company stockholders. To the extent a Company stockholder does not receive Stock Consideration for each share of Company Common Stock that such stockholder elected to be exchanged for Stock Consideration due to the proration provided in this <u>Section 2.1(d)</u>, such stockholder shall be entitled to receive the Cash Consideration for each such remaining share of Company Common Stock not exchanged for Stock Consideration.
- (e) As of the Effective Time, all such shares of Company Common Stock converted into the Merger Consideration pursuant to this <u>Section 2.1</u> shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and the holders immediately prior to the Effective Time of shares of Company Common Stock not represented

by certificates ("<u>Book-Entry Shares</u>") and the holders of certificates, which immediately prior to the Effective Time represented any such shares of Company Common Stock (each, a "<u>Certificate</u>"), shall cease to have any rights with respect thereto, except the right to receive (i) the applicable Merger Consideration to be paid in consideration therefor upon surrender of such Book-Entry Share or Certificate in accordance with <u>Section 2.3</u> or <u>Section 2.3</u>, (ii) cash in lieu of any fractional shares of Parent Common Stock to which such holder is entitled pursuant to <u>Section 2.3(e)</u> and (iii) any dividends or other distributions to which such holder is entitled pursuant to <u>Section 2.3(c)</u>, in each case without interest and subject to any applicable withholding Taxes.

#### Section 2.2 Election Procedures.

- (a) At the time of mailing of the Proxy Statement/Prospectus to holders of record of Company Common Stock entitled to vote at the Company Stockholders Meeting (such date, the "Mailing Date"), (i) an election form, (ii) a letter of transmittal (which, in the case of shares of Company Common Stock represented by Certificates, shall specify that delivery shall be effected, and risk of loss and title to the shares of Company Common Stock represented by such Certificates shall pass, only upon proper delivery of such Certificates to the Exchange Agent, upon adherence to the procedures set forth in the letter of transmittal, and shall be in such form and have such other provisions as Parent and the Company may reasonably agree) and (iii) instructions for use in effecting the surrender of the Certificates or Book-Entry Shares in exchange for payment of the applicable Merger Consideration (the material described in clauses (i) through (iii), collectively, the "Election Form") shall be mailed to each holder of record of shares of Company Common Stock as of the record date for the Company Stockholders Meeting.
- (b) Each Election Form shall permit the holder to specify (i) the number of shares of such holder's Company Common Stock with respect to which such holder makes a Cash Election, and (ii) the number of shares of such holder's Company Common Stock with respect to which such holder elects to make a Stock Election. Any shares of Company Common Stock with respect to which the Exchange Agent has not received an effective, properly completed Election Form on or before 5:00 p.m., New York time, on the Business Day that is four (4) Business Days prior to the Closing Date (which date shall be publicly announced by Parent as soon as reasonably practicable but in no event less than five (5) Business Days prior to the Closing Date) (or such other time and date as the Company and Parent shall agree in writing) (the "Election Deadline") shall be deemed to be Non-Electing Company Shares. If the Effective Time is delayed to a subsequent date, the Election Deadline shall be similarly delayed to a subsequent date, and Parent shall promptly announce any such delay and, when determined, the rescheduled Election Deadline.
- (c) Parent shall direct the Exchange Agent to make Election Forms available as may reasonably be requested from time to time by all Persons who become holders of Company Common Stock between the record date for the Company Stockholders Meeting and the Election Deadline, and the Company shall provide to the Exchange Agent all information reasonably necessary for it to perform as specified herein and as specified in any agreement with the Exchange Agent.

(d) Any election made pursuant to this Section 2.2 shall have been properly made only if the Exchange Agent shall have actually received a properly completed and signed Election Form prior to the Election Deadline accompanied by any Certificates and Book-Entry Shares representing shares of Company Common Stock. Any Election Form may be revoked or changed by the Person submitting such Election Form, by written notice received by the Exchange Agent prior to the Election Deadline. In the event an Election Form is revoked prior to the Election Deadline, the shares of Company Common Stock represented by such Election Form shall become Non-Electing Company Shares; provided that a subsequent election may be made with respect to any or all of such shares of Company Common Stock prior to the Election Deadline pursuant to this Section 2.2. In addition, all Cash Elections and Stock Elections shall automatically be revoked and all Certificates and Book-Entry Shares representing shares of Company Common Stock in the custody of the Exchange Agent shall be promptly returned without charge if this Agreement is terminated in accordance with Article VII of this Agreement or otherwise upon the written request of the holder who submitted the applicable Election Form and shares of Company Common Stock.

(e) Subject to the terms of this Agreement and of the Election Form, the Exchange Agent, in consultation with both Parent and the Company, shall have reasonable discretion to determine whether any election, revocation or change has been properly or timely made and to disregard immaterial defects in any submitted Election Form, and any good faith decisions of the Exchange Agent regarding such matters shall be binding and conclusive. None of Parent or the Company or the Exchange Agent shall be under any obligation to notify any Person of any defect in an Election Form.

#### Section 2.3 Exchange of Certificates.

(a) Exchange Agent. Prior to the Effective Time, Parent shall designate a bank or trust company mutually acceptable to Parent and the Company (the "Exchange Agent") for the purpose of exchanging shares of Company Common Stock for the Merger Consideration and enter into an agreement reasonably acceptable to the Company with the Exchange Agent relating to the services to be performed by the Exchange Agent. Parent shall deposit, or cause to be deposited, such aggregate Merger Consideration with the Exchange Agent at the Closing. The Cash Consideration portion of such aggregate Merger Consideration deposited with the Exchange Agent shall, pending its disbursement to such holders, be invested by the Exchange Agent in (i) short-term direct obligations of the United States of America or (ii) short-term obligations for which the full faith and credit of the United States of America is pledged to provide for the payment of principal and interest. Any interest and other income from such investments shall become part of the funds held by the Exchange Agent for purposes of paying the Cash Consideration portion of the aggregate Merger Consideration, subject to Section 2.3(g). No investment by the Exchange Agent of the Cash Consideration portion of the aggregate Merger Consideration shall relieve Parent, the Surviving Corporation or the Exchange Agent from making the payments required by this Article II and Parent shall promptly replace any funds deposited with the Exchange Agent lost through any investment made pursuant to this Section 2.3(a). No investment by the Exchange Agent of the Cash Consideration portion of the aggregate Merger Consideration shall have maturities that could prevent or delay payments to be made pursuant to this Agreement. Following the Effective Time, Parent agrees to make available to the Exchange Agent, from time to time as needed, additional cash or additional

shares of Parent Common Stock to pay (A) the Merger Consideration as contemplated by this Article II, (B) cash in lieu of any fractional shares of Parent Common Stock pursuant to Section 2.3(e) and (C) any dividends or other distributions pursuant to Section 2.3(c) without interest.

(b) Payment Procedures. Promptly after the Effective Time (but in no event more than five (5) Business Days thereafter), the Surviving Corporation shall cause the Exchange Agent to mail to each holder of record of Company Common Stock (other than any holder which has previously and properly surrendered all of its Certificates and Book-Entry Shares, as applicable, to the Exchange Agent in accordance with Section 2.2) (i) a letter of transmittal (which, in the case of shares of Company Common Stock represented by Certificates, shall specify that delivery shall be effected, and risk of loss and title to the shares of Company Common Stock represented by such Certificates shall pass, only upon proper delivery of such Certificates to the Exchange Agent, upon adherence to the procedures set forth in the letter of transmittal, and shall be in such form and have such other provisions as Parent and the Company may reasonably agree and shall be prepared prior to Closing) and (ii) instructions for use in effecting the surrender of the Certificates or Book-Entry Shares in exchange for payment of the applicable Merger Consideration. Upon surrender of Certificates for cancellation to the Exchange Agent or receipt of an "agent's message" by the Exchange Agent (or such other evidence, if any, of transfer as the Exchange Agent may reasonably request) in the case of Book-Entry Shares, together with such letter of transmittal, duly completed and validly executed in accordance with the instructions (and such other customary documents as may reasonably be required by the Exchange Agent), the holder of such Certificates or Book-Entry Shares shall be entitled to receive in exchange therefor, subject to any required withholding Taxes, the applicable Merger Consideration, without interest, for each share of Company Common Stock surrendered, and any Certificates surrendered shall forthwith be cancelled. If payment of such Merger Consideration is to be made to a Person other than the Person in whose name the surrendered Certificate or Book-Entry Share is registered, it shall be a condition of payment that (A) the Person requesting such exchange present proper evidence of transfer or shall otherwise be in proper form for transfer and (B) the Person requesting such payment shall have paid any transfer and other Taxes required by reason of the payment of such Merger Consideration to a Person other than the registered holder of such Certificate or Book-Entry Share surrendered or shall have established to the reasonable satisfaction of the Surviving Corporation that such Tax either has been paid or is not applicable. Until surrendered as contemplated by this Section 2.3, each Certificate and Book-Entry Share shall be deemed at any time after the Effective Time to represent only the right to receive (1) the applicable Merger Consideration as contemplated by this Article II, (2) cash in lieu of any fractional shares of Parent Common Stock to which such holder is entitled pursuant to Section 2.3(e) and (3) any dividends or other distributions to which such holder is entitled pursuant to Section 2.3(c), in each case without interest and subject to any applicable withholding Taxes.

(c) <u>Distributions with Respect to Unexchanged Shares of Parent Common Stock</u>. No dividends or other distributions declared or made after the Effective Time with respect to shares of Parent Common Stock with a record date after the Effective Time will be paid to the holder of any unsurrendered Certificate or Book-Entry Share that is a Stock Electing Company Share with respect to the shares of Parent Common Stock represented thereby, and no cash payment in lieu of any fractional shares will be paid to any such holder pursuant to <u>Section 2.3</u> (e) , until the holder of such Certificate or Book-Entry Share surrenders

such Certificate or Book-Entry Share. Subject to the effect of escheat, Tax or other applicable Laws, following surrender of any such Certificate or Book-Entry Share, the holder of the Certificate or Book-Entry Share representing whole shares of Parent Common Stock issued in exchange therefor will be paid, without interest, (i) promptly, the amount of dividends or other distributions with a record date after the Effective Time and theretofore paid with respect to such whole shares of Parent Common Stock and (ii) on the appropriate payment date, the amount of dividends or other distributions, with a record date after the Effective Time but prior to surrender and a payment date occurring after surrender, payable with respect to such whole shares of Parent Common Stock.

- (d) <u>Transfer Books</u>; No Further Ownership Rights in Company Common Stock. The applicable Merger Consideration paid in respect of shares of Company Common Stock upon the surrender for exchange in accordance with the terms of this <u>Article II</u> shall be deemed to have been paid in full satisfaction of all rights pertaining to the shares of Company Common Stock and, at the Effective Time, the stock transfer books of the Company shall be closed and thereafter there shall be no further registration of transfers on the stock transfer books of the Surviving Corporation of the shares of Company Common Stock that were outstanding immediately prior to the Effective Time. From and after the Effective Time, the holders of Certificates or Book-Entry Shares that evidenced ownership of shares of Company Common Stock outstanding immediately prior to the Effective Time shall cease to have any rights with respect to such shares of Company Common Stock other than the right to receive the applicable Merger Consideration, except as otherwise provided for herein or by applicable Law. Subject to the last sentence of <u>Section 2.3(g)</u>, if, at any time after the Effective Time, Certificates are presented to the Surviving Corporation for any reason, they shall be cancelled and exchanged as provided in this Article II.
- (e) No Fractional Shares. No certificate or scrip representing fractional shares of Parent Common Stock will be issued upon the surrender for exchange of Certificates representing, or Book-Entry Shares that are, Stock Electing Company Shares, and such fractional share interests will not entitle the owner thereof to vote or to any other rights of a stockholder of Parent. As soon as reasonably practicable after the Effective Time, each holder of a fractional share interest will be paid an amount in cash (without interest and subject to any withholding Tax) equal to the product obtained by multiplying (i) such fractional share interest to which such holder (after taking into account all fractional share interests then held by such holder) would otherwise be entitled by (ii) \$30.07. As promptly as practicable after the determination of the amount of cash, if any, to be paid to holders of fractional share interests, the Exchange Agent will so notify Parent, and Parent will promptly deposit such amount with the Exchange Agent and will cause the Exchange Agent to forward payments to such holders of fractional share interests in accordance with the provisions of this Section 2.3.
- (f) <u>Lost, Stolen or Destroyed Certificates</u>. If any Certificate shall have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the Person claiming such Certificate to be lost, stolen or destroyed and, if required by the Exchange Agent or the Surviving Corporation, the posting by such Person of a bond, in such reasonable amount as Parent may direct, as indemnity against any claim that may be made against it with respect to such Certificate, the Exchange Agent will pay, in exchange for such lost, stolen or destroyed

Certificate, the applicable Merger Consideration to be paid in respect of the shares of Company Common Stock formerly represented by such Certificate, as contemplated by this Article II.

- (g) <u>Termination of Fund</u>. At any time following the first (1st) anniversary of the Closing Date, the Surviving Corporation shall be entitled to require the Exchange Agent to deliver to it any funds or other property (including any interest received with respect thereto) that had been made available to the Exchange Agent and which have not been disbursed in accordance with this <u>Article II</u>, and thereafter Persons entitled to receive payment pursuant to this <u>Article II</u> shall be entitled to look only to the Surviving Corporation (subject to abandoned property, escheat or other similar Laws) as general creditors thereof with respect to the payment of any Merger Consideration, cash in lieu of any fractional shares of Parent Common Stock to which such holder is entitled pursuant to <u>Section 2.3(e)</u>, and any dividends or other distributions to which such holder is entitled pursuant to <u>Section 2.3(c)</u>, that may be payable upon surrender of any Company Common Stock held by such holders, as determined pursuant to this Agreement, in each case without any interest thereon and subject to any applicable withholding Taxes. Any amounts remaining unclaimed by such holders immediately prior to such time at which such amounts would otherwise escheat to or become property of any Governmental Authority shall become, to the extent permitted by applicable Law, the property of the Surviving Corporation, free and clear of all claims or interest of any Person previously entitled thereto.
- (h) No Liability. Notwithstanding any provision of this Agreement to the contrary, none of Parent, the Merger Sub, the Surviving Corporation, the Company or the Exchange Agent shall be liable to any Person for Merger Consideration or any dividends or other distributions to which such holder is entitled pursuant to Section 2.3(c) delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law.
- (i) <u>Dissenting Shares</u>. Subject to applicable Law, any portion of the Merger Consideration made available to the Exchange Agent in respect of any Dissenting Shares shall be returned to Parent, upon demand.
- (j) <u>Tax Treatment</u>. The parties agree and acknowledge that the Merger will be treated as a taxable purchase of shares of Company Common Stock for the Merger Consideration (and not as a reorganization within the meaning of Section 368(a) of the Code) for United States federal, state and local income Tax purposes.

#### Section 2.4 Treatment of Equity Awards.

(a) Parent shall not assume any Options in connection with the Merger. Each holder of an option (an "Optionholder") that represents the right to acquire shares of Company Common Stock granted under the Company Stock Plans which is outstanding immediately prior to the Effective Time (whether or not then vested or exercisable) (each, an "Option") shall be provided with notice pursuant to which all outstanding Options held by such Optionholder shall become fully vested and may be exercised by such Optionholder for a period of at least fifteen (15) days prior to the Effective Time in accordance with the terms and conditions of the applicable award agreement and Company Stock Plan under which such Option was granted. To the extent that any outstanding Option is not so exercised on or prior to the

Effective Time, such Option shall be cancelled and terminated at the Effective Time in exchange for the right to receive a cash amount equal to the product of (i) the total number of shares of Company Common Stock subject to the Option immediately prior to the Effective Time *multiplied by* (ii) the excess, if any, of the Cash Consideration *over* the exercise price per share of Company Common Stock underlying such Option; <u>provided</u> that if the exercise price per share of any such Option is equal to or greater than the Cash Consideration, such Option shall be canceled without any payment or other consideration being made in respect thereof.

- (b) Immediately prior to the Effective Time, each restricted stock unit with respect to shares of Company Common Stock granted under the Company Stock Plans ("Company RSUs.") that is outstanding immediately prior to the Effective Time and each outstanding deferred stock unit ("Company DSUs" and together with the Company RSUs, collectively, the "Company Stock Units.") shall be cancelled and entitle the holder thereof to receive from the Company, in full settlement of such Company Stock Units, a cash amount equal to the product of (i) the Cash Consideration and (ii) the total number of shares of Company Common Stock subject to such Company Stock Units (using, if applicable, the goal (100%) level of achievement under the respective award agreement to determine such number).
- (c) No later than the Effective Time, Parent shall provide, or shall cause to be provided, to the Surviving Corporation all funds necessary to fulfill the obligations under this <u>Section 2.4</u>. All payments required under this <u>Section 2.4</u> shall be made at the Effective Time.
- (d) At or prior to the Effective Time, the Company, the Company Board (or the appropriate committee thereof), shall adopt any resolutions and shall cause the Company to take any actions that may be necessary to effectuate the provisions of <u>clauses (a)</u> and <u>(b)</u> of this <u>Section 2.4</u> and to terminate, as of the Effective Time, all Company Stock Plans, and no further Options, Company RSUs or Company DSUs or other rights with respect to Company Common Stock shall be granted thereunder.
- **Section 2.5** Adjustments. If at any time during the period between the date of this Agreement and the Effective Time, any change in the outstanding shares of capital stock of the Company or Parent shall occur as a result of any reclassification, stock split (including a reverse stock split) or combination, exchange or readjustment of shares or any stock dividend or stock distribution with a record date during such period, the Merger Consideration, the exchange ratios and any other similarly dependent items shall be equitably adjusted; provided, however, that nothing in this Section 2.5 shall be deemed to permit or authorize any party hereto to effect any such change that it is not otherwise authorized or permitted to undertake pursuant to this Agreement.

**Section 2.6** <u>Appraisal Rights</u>. Notwithstanding anything in this Agreement to the contrary, shares of Company Common Stock that are issued and outstanding immediately prior to the Effective Time which are held by a stockholder who did not vote in favor of the Merger (or consent thereto in writing) and who (a) is entitled to demand and properly demands appraisal of such shares pursuant to and (b) complies in all respects with, the provisions of Section 262 of the DGCL (the "<u>Dissenting Stockholders</u>") shall not be converted into or be exchangeable for the right to receive the applicable Merger Consideration (the "<u>Dissenting</u>

Shares "), but instead such Dissenting Stockholder shall be entitled to receive such consideration as may be determined to be due to such Dissenting Stockholder pursuant to Section 262 of the DGCL (and at the Effective Time, such Dissenting Shares shall no longer be outstanding and shall automatically be cancelled and shall cease to exist, and such Dissenting Stockholder shall cease to have any rights with respect thereto, except the rights set forth in Section 262 of the DGCL), unless and until such Dissenting Stockholder shall have failed to perfect or shall have effectively waived, withdrawn or lost rights to appraisal under the DGCL. If any Dissenting Stockholder shall have failed to perfect or shall have effectively waived, withdrawn or lost such right to appraisal, or if appraisal rights are unavailable to the Company's stockholders pursuant to Section 262 of the DGCL in connection with the Transactions, such Dissenting Stockholder's shares of Company Common Stock shall thereupon be treated as if they had been converted into and become exchangeable for the right to receive, as of the Effective Time, the applicable Merger Consideration for each such share of Company Common Stock, in accordance with Section 2.1, without any interest thereon and less any applicable withholding Taxes. The Company shall give Parent (i) prompt notice of any written demands for appraisal of any shares of Company Common Stock, attempted withdrawals of such demands and any other instruments served pursuant to the DGCL and received by the Company relating to stockholders' rights of appraisal and (ii) the opportunity to participate in and direct, in each case at its own expense, all negotiations and proceedings with respect to demands for appraisal under the DGCL. Except with the prior written consent of Parent, the Company shall not voluntarily make any payment with respect to, or offer to settle or settle, any such demands prior to the Effective Time.

Section 2.7 <u>Withholding Taxes</u>. Parent, the Surviving Corporation and the Exchange Agent shall be entitled to deduct and withhold from any amounts payable pursuant to this Agreement such amounts as may be required to be deducted and withheld with respect to the making of such payment under the Internal Revenue Code of 1986 (the "<u>Code</u>"), or under any applicable provision of state, local or foreign Law related to Taxes. To the extent amounts are so withheld and paid over to the appropriate Taxing authority, the withheld amounts shall be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction and withholding was made. If any withholding obligation may be avoided by such holder providing information or documentation to Parent, the Surviving Corporation or the Exchange Agent, such information shall be requested prior to any such withholding.

#### **ARTICLE III**

#### REPRESENTATIONS AND WARRANTIES OF THE COMPANY

The Company represents and warrants to Parent and Merger Sub that, except as disclosed in the disclosure schedule delivered by the Company to Parent simultaneously with the execution of this Agreement (the "Company Disclosure Schedule") or except as set forth in (or incorporated by reference in) any of the Company SEC Documents filed after December 31, 2011 and prior to the date of this Agreement (other than any non-Company specific disclosures set forth under the heading "Risk Factors" and any non-Company specific disclosures set forth in any "forward-looking statements" disclaimer or any other disclosures that are predictive, cautionary or forward-looking in nature, and provided that no such disclosure set forth in the Company SEC Documents shall be deemed to modify or qualify the representations and

warranties set forth in <u>Section 3.2</u> (Capitalization), <u>Section 3.3(a)</u> (Authority), <u>Section 3.3(b)(i)</u> (Non-Contravention) and <u>Section 3.4</u> (Governmental Approvals)):

#### Section 3.1 Organization, Standing and Corporate Power.

(a) The Company is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware and has all requisite corporate power and authority necessary to own, lease and operate all of its properties and assets and to carry on its business as it is now being conducted. The Company is duly qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such qualification necessary, except where the failure to be so qualified or in good standing would not have a Company Material Adverse Effect.

(b) For each Subsidiary of the Company, Section 3.1(b) of the Company Disclosure Schedule lists the name, jurisdiction of incorporation or organization, each jurisdiction in which such Subsidiary is qualified to do business, the issued and authorized capitalization of such Subsidiary and the record holders of such Subsidiary's equity securities. Each of the Company's Subsidiaries is duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization and has all requisite limited liability company power and authority necessary to own, lease and operate all of its properties and assets and to carry on its business as it is now being conducted. The Company is duly qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such qualification necessary, except where the failure to be so qualified or in good standing would not have a Company Material Adverse Effect. All the outstanding shares of capital stock of, or other equity interests in, each such Subsidiary (except for directors' qualifying shares or the like) are owned directly or indirectly by the Company free and clear of all liens, pledges, security interests and transfer restrictions, other than the Permitted Liens and such transfer restrictions of general applicability as may be provided under the Securities Act of 1933 (the "Securities Act"), and other applicable securities Laws. There are no outstanding (i) securities of the Company or any of its Subsidiaries convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, any Subsidiary of the Company, (ii) options, stock appreciation rights, warrants, restricted stock units, rights or other commitments or agreements to acquire from the Company or any of its Subsidiaries, or that obligate the Company or any of its Subsidiaries to issue, any capital stock of, or other equity or voting interest in, or any securities convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, any Subsidiary of the Company, (iii) obligations of the Company to grant, extend or enter into any subscription, warrant, right, convertible or exchangeable security or other similar agreement or commitment (whether payable in equity, cash or otherwise) relating to any capital stock of, or other equity or voting interest (including any voting debt) in, any Subsidiary of the Company (the items in clauses (i), (ii) and (iii), together with the capital stock of the Subsidiaries of the Company, being referred to collectively as " Subsidiary Securities") or (iv) other obligations by the Company or any of its Subsidiaries to make any payments based on the price or value of any Subsidiary Securities. There are no Contracts of any kind which obligate the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any outstanding Subsidiary Securities. Neither the Company nor any of its Subsidiaries is a party to

any Contract restricting the transfer of, relating to the voting of, requiring registration of, or granting any preemptive rights, anti-dilution rights or rights of first refusal or similar rights with respect to any Subsidiary Securities.

(c) The Company has made available to Parent complete and correct copies of the certificate of incorporation and by-laws of the Company, in each case as amended to the date of this Agreement (the "Company Charter Documents"), and the certificate of incorporation and by-laws (or similar organizational documents) of each Subsidiary of the Company, in each case as amended to the date of this Agreement.

Section 3.2 Capitalization. The authorized capital stock of the Company consists of 100,000,000 shares of Company Common Stock and 10,000,000 shares of preferred stock, par value \$0.01 per share ("Company Preferred Stock"). At the close of business on March 5, 2013, (a) 30,782,076 shares of Company Common Stock were issued and outstanding, (b) 2,676,980 shares of Company Common Stock were held by the Company in its treasury, (c) 1,221,095 shares of Company Common Stock were issuable upon exercise of outstanding Options under the Company Stock Plans, (d) 406,919 shares of Company Common Stock were subject to the Company RSUs under the Company Stock Plans (including the goal (100%) level of achievement under the respective award agreement for Company RSUs that vest based on performance), (e) 66,359 shares of Company Common Stock were subject to Company DSUs and (f) no shares of Company Preferred Stock were issued or outstanding. All outstanding shares of Company Common Stock have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights. Except as set forth in this Section 3.2, at the close of business on March 5, 2013, there are (i) no outstanding shares of capital stock of, or other equity or voting interest in, the Company, (ii) no outstanding securities of the Company convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, the Company, (iii) no outstanding options, stock appreciation rights, warrants, restricted stock units, rights or other commitments or agreements to acquire from the Company, or that obligates the Company to issue, any capital stock of, or other equity or voting interest in, or any securities convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, the Company, (iv) no obligations of the Company to grant, extend or enter into any subscription, warrant, right, convertible or exchangeable security or other similar agreement or commitment (whether payable in equity, cash or otherwise) relating to any capital stock of, or other equity or voting interest (including any voting debt) in, the Company (the items in clauses (i), (ii), (iii) and (iv), together with the capital stock of the Company, being referred to collectively as "Company Securities") and (v) no other obligations by the Company or any of its Subsidiaries to make any payments based on the price or value of the Company Securities. Since March 5, 2013, the Company has not issued any Company Securities other than shares of Company Common Stock upon the exercise of Options outstanding on March 5, 2013. There are no outstanding Contracts of any kind which obligate the Company or any of its Subsidiaries to repurchase, redeem or otherwise acquire any Company Securities. Neither the Company nor any of its Subsidiaries is a party to any Contract restricting the transfer of, relating to the voting of, requiring registration of, or granting any preemptive rights, anti-dilution rights or rights of first refusal or similar rights with respect to any Company Securities. Section 3.2 of the Company Disclosure Schedule sets forth a list of all outstanding Options and Company Stock Units, the name of each holder of such Options and Company Stock Units and the exercise price of each Option.

#### Section 3.3 Authority; Noncontravention.

- (a) The Company has all necessary corporate power and authority to execute and deliver this Agreement and, subject to obtaining the Company Stockholder Approval, to perform its obligations hereunder and to consummate the Transactions. The execution and delivery of and performance by the Company under this Agreement, and the consummation of the Transactions, have been duly authorized and approved by the Company Board and, except for obtaining the Company Stockholder Approval, no other corporate action on the part of the Company is necessary to authorize the execution and delivery of and performance by the Company under this Agreement and the consummation by it of the Transactions. This Agreement has been duly executed and delivered by the Company and, assuming due authorization, execution and delivery hereof by the other parties hereto, constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except that such enforceability (i) may be limited by bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and other similar Laws of general application affecting or relating to the enforcement of creditors' rights generally and (ii) is subject to general principles of equity, whether considered in a proceeding at law or in equity (the "Bankruptcy and Equity Exception").
- (b) Neither the execution and delivery of this Agreement by the Company, nor the consummation by the Company of the Transactions, nor compliance by the Company with any of the terms or provisions hereof, will (i) assuming the Company Stockholder Approval is obtained, conflict with or violate any provision of the Company Charter Documents, (ii) assuming that each of the consents, authorizations and approvals referred to in Section 3.4 and the Company Stockholder Approval are obtained (and any condition precedent to any such consent, authorization or approval has been satisfied) and each of the filings referred to in Section 3.4 are made and any applicable waiting periods referred to therein have expired, violate any Law applicable to the Company or any of its Subsidiaries or (iii) result in any breach of, constitute a default (with or without notice or lapse of time, or both) under, give rise to any right of termination, amendment, acceleration or cancellation of, or result in loss of any benefit or the imposition of any additional payment or liability under, any Contract to which the Company or any Subsidiary is a party, or result in the creation of a Lien, other than any Permitted Lien, upon any of the properties or assets of the Company or any of its Subsidiaries, other than, in the case of clauses (ii) and (iii), as would not have a Company Material Adverse Effect.
- (c) The Company Board, at a meeting duly called and held and acting upon the unanimous recommendation of the Review Committee, has unanimously (with one director not present) (i) approved and declared advisable this Agreement, the Voting Agreement and the Transactions, including the Merger, (ii) determined that this Agreement, the Voting Agreement and the Transactions, including the Merger, are fair to and in the best interests of the Company and its stockholders, (iii) assuming the accuracy of the representations and warranties of Parent and Merger Sub set forth in Section 4.14, took all actions necessary so that the restrictions on business combinations and stockholder vote requirements contained in Section 203 of the DGCL will not apply with respect to or as a result of the Merger, this Agreement and the Transactions, (iv) directed that the adoption of this Agreement be submitted to a vote of the stockholders of the Company at the Company Stockholders Meeting and (v) resolved, subject to Section 5.3, to recommend that stockholders of the Company adopt this Agreement.

Section 3.4 Governmental Approvals. Except for (a) the filing with the SEC of a proxy statement relating to the Company Stockholders Meeting (as amended or supplemented from time to time, the "Proxy Statement/Prospectus") and of Parent's registration statement on Form S-4 (the "Form S-4"), in which the Proxy Statement/Prospectus will be included as a prospectus, and declaration of effectiveness of the Form S-4 by the SEC, and other filings required under, and compliance with other applicable requirements of, the Securities Exchange Act of 1934 (the "Exchange Act") and the rules of Nasdaq and applicable state securities and "blue sky" laws, (b) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware pursuant to the DGCL and (c) filings required under, and compliance with other applicable requirements of, the HSR Act, no consents or approvals of, or filings, declarations or registrations with, any Governmental Authority are necessary for the execution and delivery of this Agreement by the Company and the consummation by the Company of the Transactions, other than as would not have a Company Material Adverse Effect.

#### Section 3.5 Company SEC Documents; Undisclosed Liabilities.

(a) The Company has filed with or furnished to the SEC, on a timely basis, all registration statements, reports and proxy statements required to be filed or furnished since January 1, 2011 (collectively, and in each case including all exhibits and schedules thereto and documents incorporated by reference therein, as such statements and reports may have been amended since the date of their filing, the " Company SEC Documents "). As of their respective effective dates (in the case of Company SEC Documents that are registration statements filed pursuant to the requirements of the Securities Act) and as of their respective filing dates (in the case of all other Company SEC Documents), or in the case of amendments thereto, as of the last such amendment, the Company SEC Documents complied in all material respects with the requirements of the Exchange Act, the Securities Act and the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), as the case may be, and the rules and regulations of the SEC thereunder, applicable to such Company SEC Documents, and none of the Company SEC Documents as of such respective dates (or, if amended, the date of the filing of such amendment, with respect to the disclosures that are amended) contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Neither the Company nor any of its Subsidiaries has received from the SEC any written comments or questions with respect to any of the Company SEC Documents or any written notice from the SEC that such Company SEC Documents are being reviewed or investigated, and, to the Knowledge of the Company, there is not, as of the date of this Agreement, any investigation or review being conducted by the SEC of any Company SEC Documents. The principal executive officer of the Company and the principal financial officer of the Company (and each former principal executive officer or principal financial officer of the Company) have made the certifications required by Sections 302 and 906 of the Sarbanes-Oxley Act, and the rules and regulations of the SEC promulgated thereunder with respect to the Company SEC Documents that were required to be accompanied by such certifications. For purposes of the preceding sentence, "principal executive officer" and "principal financial officer" shall have the meanings given to such terms in the Sarbanes-Oxley Act. None of the Company's Subsidiaries is required to file any forms, reports, schedules, statements or other documents with the SEC.

- (b) None of the information supplied or to be supplied by the Company for inclusion or incorporation by reference in (i) the Form S-4 or any amendment or supplement thereto will, at the time such Form S-4 or any amendment or supplement thereto is filed with the SEC or at the time such Form S-4 becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein not misleading or (ii) the Proxy Statement/Prospectus will, at the date of mailing to holders of Company Common Stock and at the time of the Company Stockholders Meeting to be held in connection with the Merger, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein, in the light of the circumstances under which they were made, not misleading. When filed, the Proxy Statement/Prospectus will comply as to form in all material respects with the requirements of the Exchange Act and the rules and regulations thereunder. No representation or warranty is made by the Company with respect to statements made or incorporated by reference therein based on information supplied by Parent for inclusion or incorporation by reference in the Proxy Statement/Prospectus or the Form S-4.
- (c) Except to the extent updated, amended, restated or corrected by a subsequent Company SEC Document, as of their respective dates of filing with the SEC, the consolidated financial statements of the Company included in the Company SEC Documents (i) complied as to form in all material respects with all applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto (except, in the case of unaudited statements, as permitted by Form 10-Q of the SEC), (ii) have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except (A) as may be indicated in the notes thereto or (B) as permitted by Regulation S-X) and (iii) present fairly, in all material respects, the consolidated financial position of the Company and its Subsidiaries, and the results of their operations and cash flows, for each of the dates and for the periods shown, in conformity with GAAP (subject to normal year-end adjustments in the case of any unaudited interim financial statements that were not, or are not expected to be, material in amount). Since January 1, 2011, the Company has not made any change in the accounting practices or policies applied in the preparation of its financial statements, except as required by GAAP, SEC rule or policy or applicable Law.
- (d) The Company has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 under the Exchange Act) as required by Rule 13a-15 under the Exchange Act. The Company's disclosure controls and procedures are reasonably designed to ensure that all material information (both financial and nonfinancial) required to be disclosed by the Company in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such material information is accumulated and communicated to the Company's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes-Oxley Act. The principal executive officer and principal financial officer of the Company have evaluated the effectiveness of the Company's disclosure controls and procedures and, to the extent required by applicable Law, presented in any applicable Company SEC Document that is a report on Form 10-K or Form 10-Q, or any amendment thereto, its conclusions about the effectiveness of the disclosure controls and procedures as of the end of the

period covered by such report or amendment based on such evaluation. The Company's management has completed an assessment of the effectiveness of the Company's internal control over financial reporting in compliance with the requirements of Section 404 of the Sarbanes-Oxley Act for the year ended December 31, 2012, and such assessment concluded that such controls were effective.

- (e) To the Knowledge of the Company, there are no (and none of the Company's independent auditors has identified to the Company Board or the Company's senior management any) (i) significant deficiencies or material weaknesses in the system of internal accounting controls utilized by the Company and its Subsidiaries, (ii) instances of fraud, whether or not material, committed by the Company's senior management or other employees who have a role in the preparation of financial statements or the internal accounting controls utilized by the Company and its Subsidiaries or (iii) written claims or allegations regarding any of the foregoing.
- (f) The Company is in compliance in all material respects with the applicable corporate governance rules and regulations of Nasdaq.
- (g) Neither the Company nor any of its Subsidiaries has any liabilities which would be required to be reflected or reserved against on a consolidated balance sheet of the Company prepared in accordance with GAAP or the notes thereto, except for liabilities (i) reflected or reserved against on the balance sheet of the Company and its Subsidiaries as of December 31, 2012 (the "Balance Sheet Date") (including the notes thereto) included in Section 3.5(g) of the Company Disclosure Schedule, (ii) incurred after the Balance Sheet Date in the ordinary course of business and that individually or in the aggregate would not have a Company Material Adverse Effect or (iii) as contemplated by this Agreement or otherwise arising in connection with the Transactions.

Section 3.6 Absence of Certain Changes. Between January 1, 2013 and the date of this Agreement, (a) except in connection with the Transactions, the business of the Company and its Subsidiaries has been conducted in all material respects in the ordinary course of business consistent with past practice, (b) there has not been any (i) fact, change, event, circumstance, occurrence or effect that, individually or in the aggregate, has had or would reasonably be expected to have a Company Material Adverse Effect or (ii) action that, if taken after the date of this Agreement without the prior written consent of Parent, would constitute a breach of clauses (iv), (v), (vii)(B), (xi) or (xv) of Section 5.1(a) and (c) the Company and its Subsidiaries have not acquired any defaulted or charged-off accounts receivable portfolios other than those that have previously disclosed to Parent in the "data room" maintained by or on behalf of the Company in connection with the process leading to the execution of this Agreement.

Section 3.7 <u>Legal Proceedings</u>. There is no pending or, to the Knowledge of the Company, threatened legal or administrative proceeding, claim, suit or action against the Company or any of its Subsidiaries that (a) involves any Governmental Authority as a party thereto, (b) is a consumer class action or (c) if resolved in a manner adverse to the Company and its Subsidiaries, would reasonably be expected to result (based on the advice of the Company's or its Subsidiaries counsel prior to the date hereof) in (x) an aggregate liability in excess of \$1,000,000 or (y) the imposition on the Company or its Subsidiaries of any material (i)

injunctive or other non-monetary relief, or (ii) legal restraint on or limitation on the Company's ability to operate the business of the Company and its Subsidiaries (taken as a whole) in substantially the same manner as operated immediately prior to the date hereof. There is no pending or, to the Knowledge of the Company, threatened legal or administrative proceeding, claim, suit or action against the Company or any of its Subsidiaries that, individually or in the aggregate, has had or would reasonably be expected to have a Company Material Adverse Effect.

#### Section 3.8 Compliance With Laws; Permits .

(a) Except with respect to the matters described in Section 3.5, Section 3.9, Section 3.10 and Section 3.11, which are excluded from the provisions of this Section 3.8, the Company and its Subsidiaries are, and since January 1, 2011 have been, in compliance in all material respects with all laws, statutes, ordinances, codes, rules, regulations, decrees (including the Consent Decree) judgments, injunctions and orders of Governmental Authorities (collectively, "Laws") applicable to the Company or any of its Subsidiaries, including all laws relating to the collection of debts. Since January 1, 2011, neither the Company nor any of its Subsidiaries has (i) received any written notice of any administrative, civil or criminal investigation or audit by any Governmental Authority responsible for consumer protection or licensing, in each case related to debt collection practices, relating to the Company or any of its Subsidiaries or (ii) to the Knowledge of the Company (x) received any written notice from any Governmental Authority responsible for consumer protection or licensing, in each case related to debt collection practices, alleging a violation by the Company or any of its Subsidiaries of any applicable Law, (y) provided any written notice to any covernmental Authority responsible for consumer protection or licensing, in each case related to debt collection practices, regarding any violation by the Company or any of its Subsidiaries of any applicable Law or (z) received from or provided to any other Governmental Authority any written notice regarding an alleged violation by the Company or any of its Subsidiaries of an applicable Law which, in each case, individually or in the aggregate, could be materially adverse to the Company and its Subsidiaries taken as a whole, and no such notice referred to in clauses (i), (ii) or (iii) of this Section 3.8(a) remains outstanding or unresolved as of the date of this Agreement. Neither the Company nor any of its Subsidiaries is subject to any material injunction, order, judgm

(b) The Company and each of its Subsidiaries hold (and are, and since January 1, 2011 have been, in compliance in all material respects with) all licenses, franchises, permits, certificates, approvals and authorizations from Governmental Authorities required by Law for the conduct of their respective businesses as they are now being conducted which are necessary for the Company and its Subsidiaries to operate their respective businesses as currently operated (collectively, the "Company Permits"). There has occurred no material violation of, material default under or event (in each case with or without notice or lapse of time or both) giving to others any right of revocation, non-renewal, adverse modification or cancellation of, with or without notice or lapse of time or both, any such Company Permit, nor, to the Knowledge of the Company, would any such revocation, non-renewal, adverse modification or cancellation result from the consummation of the Transactions, and no suspension or cancellation of any such Company Permits is pending or, to the Knowledge of the Company, threatened. Since January 1, 2011, neither the Company nor any of its Subsidiaries has received any written notice from any Governmental Authority regarding (i) any material violation by the Company or any of its

Subsidiaries of any Company Permits or the failure to have any required Company Permits or (ii) any revocation, cancellation or termination of any Company Permits held by the Company or any of its Subsidiaries, and no such notice in either case remains outstanding or unresolved as of the date of this Agreement.

#### Section 3.9 Tax Matters.

- (a) Each of the Company and its Subsidiaries has (i) timely filed, or has caused to be timely filed on its behalf, taking into account any extension of time within which to file, all Tax Returns required to be filed by it in each jurisdiction in which they are required to be filed, and all such filed Tax Returns are correct and complete in all material respects, and have been completed in accordance with applicable Laws in all material respects, (ii) duly paid all material Taxes that are due (whether or not shown on any Tax Return), (iii) duly withheld and paid all material Taxes required by applicable Law to have been withheld and paid in connection with amounts paid or owing to any employee, independent contractor, creditor, or other third party and (iv) complied with all information reporting and back-up withholding requirements, and have maintained all required records with respect thereto. No deficiency with respect to Taxes has been proposed, asserted or assessed against the Company or any of its Subsidiaries which has not been fully paid or adequately reserved in the Company SEC Documents. No audit or other administrative or court proceedings are pending or being conducted with any Governmental Authority with respect to Taxes of the Company or any of its Subsidiaries, and no written notice indicating an intent to open an audit or otherwise review has been received. Neither the Company nor any Subsidiary has incurred any material liabilities for Taxes since the Balance Sheet Date other than in the ordinary course of business and consistent with past practices. There are no Liens for material Taxes upon any property or asset of either of the Company or any of its Subsidiaries, except for Permitted Liens for Taxes.
- (b) Neither the Company nor any of its Subsidiaries has any request for a ruling in respect of Taxes pending before any Taxing authority.
- (c) No written claim has been made by any Taxing authority in a jurisdiction where the Company or any of its Subsidiaries does not file Tax Returns that it is or may be subject to taxation by that jurisdiction. Neither the Company nor any of its Subsidiaries has waived any statute of limitations in respect of Taxes or agreed to any extension of time with respect to a material Tax assessment or deficiency.
- (d) Neither the Company nor any of its Subsidiaries is a party to or bound by any agreement providing for the allocation or sharing of Taxes or, under Treasury Regulation Section 1.1502-6 (or any similar provision of Law), as a transferee or successor, or otherwise, is liable for the unpaid Taxes of any entity other than the Company or a Subsidiary (excluding customary Tax indemnification provisions in commercial Contracts not primarily relating to Taxes).
- (e) Neither the Company nor any of its Subsidiaries will be required to include any material item of income in, or exclude any material item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of any (i) change in method of accounting for a taxable period ending on or prior to the Closing

Date, (ii) "closing agreement" as described in Section 7121 of the Code (or any corresponding or similar provision of state, local, or foreign Law) executed on or prior to the Closing Date, (iii) intercompany transactions or any excess loss account described in Treasury Regulations under Section 1502 of the Code (or any corresponding or similar provision of state, local, or foreign Law), (iv) installment sale or open transaction disposition made on or prior to the Closing Date, (v) prepaid amount received on or prior to the Closing Date or (vi) election under Section 108(i) of the Code.

- (f) Neither the Company nor any of its Subsidiaries has engaged in a reportable transaction under Treasury Regulation Section 1.6011-4(b), including any transaction that is the same as or substantially similar to one of the types of transactions that the IRS has determined to be a tax avoidance transaction and identified by notice, regulation or other form of published guidance as a listed transaction, as set forth in Treasury Regulation Section 1.6011-4(b)(2).
- (g) Neither the Company nor any of its Subsidiaries has been either a "distributing corporation" or a "controlled corporation" in a transaction during the last two (2) years which the parties treated as a distribution to which Section 355 of the Code applied.
- (h) For purposes of this Agreement, (i) "Taxes" shall mean (A) any and all federal, state, local or foreign taxes, charges, fees, imposts, levies or other assessments, including all net income, gross receipts, capital, sales, use, ad valorem, value added, transfer, franchise, profits, inventory, capital stock, license, withholding, payroll, employment, social security, unemployment, excise, severance, escheat, stamp, occupation, property and estimated taxes, customs duties, fees, assessments and charges of any kind whatsoever and all interest, penalties, fines, additions to tax or additional amounts imposed by any Governmental Authority in connection with any of the foregoing, (B) any liability for the payment of any amounts of the type described in clause (A) as a result of being or having been a member of an affiliated, consolidated, combined or unitary group for any period (including any Liability under Treasury Regulation Section 1.1502-6 or any comparable provision of applicable Laws (including any arrangement for group or consortium relief or similar arrangement)) and (C) any liability for the payment of any amounts of the type described in clause (A) or (B) as a result of any obligation to indemnify any other Person or as a result of any obligations under any agreements or arrangements with any other Person with respect to such amounts and including any Liability for taxes of a predecessor or transferor or by operation of law; and (ii) "Tax Returns" shall mean any return, report, claim for refund, estimate, information return or statement or other similar document relating to or required to be filed with any Governmental Authority with respect to Taxes, including any schedule or attachment thereto, and including any amendment thereof.

Notwithstanding any other provisions of this Agreement, this <u>Section 3.9</u> and <u>Section 3.5</u> and <u>Section 3.10</u> constitutes the sole and exclusive representation or warranty of the Company relating to Tax matters, including compliance with Laws relating thereto.

Section 3.10 <u>Employee Benefits Matters</u>. <u>Section 3.10</u> of the Company Disclosure Schedule lists each Company Plan. The Company has made available to Parent correct and complete copies of (a) the current plan document for each Company Plan, (b) the most recent annual reports on Form 5500 as filed with respect to each Company Plan (if any

such report was required), (c) the most recent summary plan description for each Company Plan for which such summary plan description is required and (d) each trust agreement and material insurance contract relating to any Company Plan. Each Company Plan has been maintained, funded and administered, in form and operation, in compliance in all material respects with its terms and with the applicable provisions of ERISA, the Code and all other applicable Laws. There are no pending or, to the Knowledge of the Company, threatened claims, suits, audits or investigations (other than claims for benefits in the ordinary course) with respect to any Company Plans that would reasonably be expected to result in a liability to the Company and its Subsidiaries (taken as a whole) individually or in the aggregate in excess of \$250,000. Each Company Plan that is a "pension plan" (as defined in Section 3(2) of ERISA) and that is intended to be tax qualified under Section 401(a) of the Code (each, a "Company Pension Plan") is covered by a current favorable determination letter from the IRS or is the subject of a pending, timely filed application therefor, and no circumstance exists that would be reasonably expected to result in the revocation of any such determination letter or the refusal by the IRS to approve any such application. The Company has made available to Parent a correct and complete copy of the most recent determination letter received with respect to each Company Pension Plan. Neither the Company nor any of its Subsidiaries maintains, sponsors, contributes to or has or would reasonably be expected to have any current or contingent liability or obligation (directly or indirectly) under or with respect to a multiemployer plan (as defined in Section 4001(a)(3) of ERISA) or a plan that is or was subject to Section 302 or Title IV of ERISA or section 412 of the Code. Except as set forth on Section 3.10 of the Company Disclosure Schedule, the consummation of the Transactions will not, either alone or in combination with another event, (i) entitle any current or former employee, officer, director or independent contractor of the Company or any of its Subsidiaries to severance pay, or a transaction or retention bonus, unemployment compensation or any other payment or additional compensation, or (ii) accelerate the time of payment or vesting, or increase the amount of compensation due any such employee, officer, director or independent contractor. Except as set forth on Section 3.10 of the Company Disclosure Schedule, no Company Plan provides (or would reasonably be expected to require the Company or any of its Subsidiaries to provide) any postemployment health care coverage or, to the Knowledge of the Company, other welfare benefits except as specifically required by applicable Laws. No amount paid or which may become payable under any Company Plan or otherwise will fail to be deductible for federal income tax purposes by virtue of Section 280G or Section 162(m) of the Code. Neither the Company nor any of its Subsidiaries has or would be reasonably expected to have any indemnity obligation for any Taxes imposed under Section 4999 or 409A of the Code. No amounts that have been paid or that may become payable under any Company Plan would reasonably be expected to become subject to the interest and additional tax set forth under Section 409A(a)(1)(B) of the Code. Notwithstanding any other provisions of this Agreement, this Section 3.10 and Section 3.5 and Section 3.20 constitutes the sole and exclusive representation and warranty of the Company relating to pension and employee benefit or liabilities or obligations, including compliance with Laws relating thereto.

**Section 3.11** Environmental Matters. Except for those matters that, individually or in the aggregate, would not reasonably be expected to result in a Company Material Adverse Effect, (a) each of the Company and its Subsidiaries is, and since January 1, 2011 has been, in compliance in all respects with all applicable Environmental Laws, which compliance includes obtaining, maintaining or complying with all Company Permits required under Environmental Laws for the operation of their respective businesses, and (b) there is no.

and since January 1, 2011 there has not been, any investigation, suit, claim or action relating to or arising under Environmental Laws (including relating to or arising from the Release, threatened Release or exposure to any Hazardous Material) that is pending or, to the Knowledge of the Company, threatened against the Company or any of its Subsidiaries or any real property currently or previously owned, operated or leased by the Company or any of its Subsidiaries. Neither the Company nor any of its Subsidiaries has received any written notice of, or entered into any order, settlement, judgment, injunction or decree involving uncompleted, outstanding or unresolved material liabilities or corrective or remedial obligations relating to or arising under Environmental Laws (including relating to or arising from the Release, threatened Release or exposure to any Hazardous Material). Notwithstanding any other provisions of this Agreement, this Section 3.11 constitutes the sole and exclusive representation and warranty of the Company relating to environmental matters, including compliance with Laws relating thereto.

Section 3.12 Intellectual Property. Section 3.12 of the Company Disclosure Schedule contains a true and complete list, as of the date hereof, of all (i) Intellectual Property owned or licensed by the Company or its Subsidiaries ("Company IP") that is the subject of any issuance, registration, certificate or application by, to or with any Governmental Authority or authorized private registrar, including registered trademarks, registered copyrights, issued patents, domain name registrations and pending applications for any of the foregoing, and (ii) material unregistered Company IP (other than trade secrets). To the Knowledge of the Company, the Company and its Subsidiaries' rights in the Company IP are valid, subsisting and enforceable. The Company and each of its Subsidiaries has taken reasonable steps to maintain the Company IP and to protect and preserve the confidentiality of all trade secrets. Section 3.12 of the Company Disclosure Schedule contains a complete and accurate list of all material Contracts relating to material Company IP (other than licenses for shrinkwrap, clickwrap or other similar commercially available off-the-shelf Software that has not been modified or customized by a third party for the Company or any of its Subsidiaries) ("Company IP Agreements"). Except as may be limited by the Bankruptcy and Equity Exception, all Company IP Agreements are valid and binding against the Company or the applicable Subsidiary party thereto, and to the Company's Knowledge, the counterparties thereto, in accordance with their respective terms, and neither the Company nor any of its Subsidiaries nor, to the Knowledge of the Company, no third party has violated any provision of, or committed or failed to perform any act which, with notice or lapse of time or both, would become a default by the Company or any of its Subsidiaries or such third party under, any Company IP Agreement. The conduct of the Company's business as currently conducted does not infringe or otherwise violate in any material respect any Person's Intellectual Property, (b) there is no pending or, to the Knowledge of the Company, threatened legal or administrative proceeding, claim, suit or action regarding such infringement against the Company or any of its Subsidiaries, (c) to the Knowledge of the Company, no Person is infringing or otherwise violating any Intellectual Property owned by the Company and (d) no legal or administrative proceeding, claim, suit or action regarding such infringement or other violation is pending or, to the Knowledge of the Company, threatened by any Person against the Company or any of its Subsidiaries. Notwithstanding any other provisions of this Agreement, this Section 3.12 constitutes the only representation and warranty of the Company with respect to any actual or alleged infringement or other violation of any Intellectual Property of any other Person.

#### Section 3.13 Absence of Rights Agreement; Anti-Takeover Provisions .

- (a) The Company is not party to a rights agreement, "poison pill" or similar agreement or plan.
- (b) Assuming the representations and warranties of Parent and Merger Sub set forth in <u>Section 4.14</u> are true and correct, no "fair price," "moratorium," "control share acquisition" or other form of antitakeover statute or regulation is applicable to the Agreement, the Voting Agreement or the Transactions.

Section 3.14 Property. The Company does not own any real property. Section 3.14 of the Company Disclosure Schedule sets forth the address of all real property in which the Company or any Subsidiary holds a leasehold or subleasehold estate (the "Leased Real Property", and the leases or subleases for such Leased Real Property being referred to as the "Leases"). The Company has made available to Parent a true and complete copy of each of the Leases. Except as may be limited by the Bankruptcy and Equity Exception, all Leases are valid and binding against the Company or the applicable Subsidiary party thereto, free and clear of all Liens and Encumbrances (except in all cases for Permitted Liens and Permitted Encumbrances) and is in full force and effect. Neither the Company nor any of its Subsidiaries nor, to the Knowledge of the Company, no third party has violated any material provision of, or committed or failed to perform any act which, with notice or lapse of time or both, would become a default by the Company or any of its Subsidiaries or such third party under, such Lease.

#### Section 3.15 Contracts.

- (a) Except as attached as exhibits to the Company SEC Documents, <u>Section 3.15</u> of the Company Disclosure Schedule lists each of the following Contracts (and all material amendments, modifications and supplements thereto and all material side letters to which the Company or any of its Subsidiaries is a party affecting the obligations of any party thereunder) to which the Company or any of its Subsidiaries is a party or by which any of their respective material properties or assets are bound (each such Contract, and each "material contract" filed as an exhibit to the Company SEC Documents, a "<u>Material Contract</u>"):
  - (i) any agreement or indenture relating to the borrowing of money or any guarantee of any such borrowing which involves more than \$1,000,000 or relating to any interest rate, currency or commodity derivatives or hedging transactions;
  - (ii) any partnership, limited liability company agreement, joint venture, profit-sharing or similar agreement entered into with any Person other than the Company or any of its Subsidiaries;
  - (iii) any agreement for the acquisition, merger or sale or similar transaction involving any business of the Company or any of its Subsidiaries or acquisition of defaulted or charged-off accounts receivable portfolios from consumer credit originators by the Company or any Subsidiary (A) for aggregate consideration under such agreement in excess of \$500,000 or (B) that has continuing indemnification, "earn-out" or other contingent payment obligations by the Company;

- (iv) any collective bargaining agreements, memoranda of understanding, settlements or other labor agreements with any union or labor organization applicable to the Company, its affiliates or their employees;
- (v) any Contract that obligates the Company or its Subsidiaries to make any capital commitment or capital expenditure in excess of \$500,000 in the aggregate;
- (vi) any employment, consulting, deferred compensation, change of control, severance or other similar agreement which contains a change of control or "golden parachute" provision or that otherwise provides for base salary or base compensation in any fiscal year that is equal to or greater than \$150,000;
  - (vii) any Contract with any Governmental Authority;
- (viii) any Contract that contains any covenant granting "most favored nation" status that, following the Transactions, would apply to or be affected by actions taken by Parent, the Surviving Corporation or any of their respective Subsidiaries or Affiliates;
- (ix) any Contract that contains any covenant that (x) limits the ability of the Company or any of its Subsidiaries (or, after the Effective Time, Parent, the Surviving Corporation or any of their respective Subsidiaries) to engage in any line of business or to compete with any Person or operate at any location, (y) would require the disposition of any material assets or line of business of the Company or any of its Subsidiaries (or, after the Effective Time, Parent, the Surviving Corporation or any of their respective Subsidiaries), or (z) prohibits or limits in any material respect the right of the Company or any of its Subsidiaries to make, sell or distribute any products or services or use, transfer, license, distribute or enforce any of their respective material Intellectual Property rights;
- (x) any Contract that contains a put, call or similar right pursuant to which the Company or any of its Subsidiaries would be required to purchase or sell, as applicable, any assets that have a fair market value or purchase price of more than \$250,000;
- (xi) any other agreement to which the Company or any of its Subsidiaries is a party or by which they or any of their assets are bound and which involves consideration or other obligations in excess of \$500,000 annually other than any such agreement with any third party debt collection agency or law firm collection agreement;
- (xii) any Contract to which any directors, executive officers (as such term is defined the Exchange Act) or 5% stockholders of the Company or, to the Knowledge of the Company, any of their Affiliates (other than the Company or any Company Subsidiary) or immediate family members is a party;

- (xiii) any Contract that requires a consent to consummate the Transactions, if the failure to obtain such consent would or would reasonably be expected to prevent or materially delay the consummation of the Transactions in accordance with the terms of this Agreement; and
- (xiv) any Contract which is or would be required to be filed by the Company as a "material contract" pursuant to Item 601(b) (10) of Regulation S-K under the Securities Act.
- (b) Each Company Material Contract is valid and binding on the Company and any of its Subsidiaries to the extent the Company or such Subsidiary is a party thereto, as applicable, and to the Knowledge of the Company, each other party thereto, and is in full force and effect and enforceable in accordance with its terms (subject to the Bankruptcy and Equity Exception). The Company and each of its Subsidiaries, and, to the Knowledge of the Company, any other party thereto, has performed in all material respects all obligations required to be performed by it under each Company Material Contract, and no event has occurred that with notice or lapse of time or both would constitute a material breach or material default thereunder by the Company or any of its Subsidiaries, or, to the Knowledge of the Company, any other party thereto. Neither the Company nor any of its Subsidiaries has received any written notice or other written communication regarding any actual or possible material violation or material breach of or material default under, or intention to cancel or modify, any Material Contract.
- Section 3.16 <u>Opinion of Financial Advisor</u>. The Review Committee and the Company Board has received the opinion of William Blair & Company, L.L.C., dated as of the date of this Agreement, to the effect that, as of such date, and subject to the various assumptions and qualifications set forth therein, the Merger Consideration to be received in the Merger by holders of the Company Common Stock is fair from a financial point of view to the holders of the Company Common Stock.
- **Section 3.17 Brokers and Other Advisors**. Except for William Blair & Company, L.L.C., no broker, investment banker, financial advisor or other Person is entitled to any broker's, finder's, financial advisor's or other similar fee, in connection with the Transactions based upon arrangements made by or on behalf of the Company or any of its Subsidiaries. The Company has furnished to Parent a true and complete copy of its agreement with William Blair & Company, L.L.C.
- **Section 3.18** Company Stockholder Approval. The adoption of this Agreement by the affirmative vote (in person or by proxy) of the holders of a majority of the outstanding shares of Company Common Stock entitled to vote at the Company Stockholders Meeting (the "Company Stockholder Approval") is the only vote or approval of the holders of any class or series of capital stock of the Company necessary to adopt this Agreement and approve the Transactions.
- **Section 3.19 Share Ownership**. The Company has not been, at any time during the three (3) years preceding the date hereof, an "interested stockholder" of Parent, as defined in Section 203 of the DGCL. As of the date of this Agreement, neither the Company nor its

Subsidiaries own (directly or indirectly, beneficially or of record) any shares of capital stock of Parent and neither the Company nor its Subsidiaries hold any rights to acquire any shares of capital stock of Parent except pursuant to this Agreement.

Section 3.20 Employment Law Matters. The Company and each of its Subsidiaries is, and since January 1, 2011 has been, in compliance in all material respects with all applicable Laws relating to employment, employment practices, terms and conditions of employment, worker classification (including the proper classification of workers as independent contractors and consultants and for overtime purposes), prohibited discrimination, equal employment, fair employment practices, meal and rest periods, immigration status, employee safety and health, wages (including overtime wages), compensation, and hours of work. Neither the Company nor any of its Subsidiaries has any material liability with respect to any misclassification of: (x) any Person as an independent contractor rather than as an employee; (y) any employee leased from another employer; or (z) any employee currently or formerly classified as exempt from overtime wages. Neither the Company nor any of its Subsidiaries is a party to a conciliation agreement, consent decree or other agreement or order with any Governmental Authority which primarily relates to employment matters. Since January 1, 2011, there have been no strikes, slowdowns, work stoppages, lockouts or other material labor disputes with respect to any employees of the Company or any of its Subsidiaries. Each of the Company and its Subsidiaries is in compliance in all material respects with the Worker Adjustment Retraining Notification Act of 1998, as amended, and similar state Laws (collectively, "WARN"). In the past two years, (i) neither the Company nor any of its Subsidiaries has effectuated a "plant closing" (as defined in WARN) affecting any site of employment or facility of its business, and (ii) there has not occurred a "mass layoff" (as defined in WARN) affecting any site of employment or facility of the Company or any of its Subsidiaries. Except as set forth on Section 3.20 of the Company Disclosure Schedule, none of such employees is a party to a written employment agreement with the Comp

Section 3.21 <u>Insurance</u>. Section 3.21 of the Company Disclosure Schedule sets forth all insurance policies maintained by the Company and its Subsidiaries (the "<u>Company Insurance Policies</u>"). All of the Company Insurance Policies are in full force and effect. Neither the Company nor any of its Subsidiaries is in material breach or default, and neither the Company nor any of its Subsidiaries has taken any action or failed to take any action which, with notice or the lapse of time, would constitute such a breach or default, or permit termination of any of the Company Insurance Policies. No written notice of cancellation or termination has been received by the Company with respect to any such Company Insurance Policy other than as is customary in connection with renewals of existing Company Insurance Policies. Set forth on Section 3.21 of the Company Disclosure Schedule is a true and complete list of all currently outstanding claims made by the Company and its Subsidiaries under the Company Insurance Policies. Neither the Company nor any of its Subsidiaries has failed to timely report any claim under a Company Insurance Policy, except where such failure to timely report would not result in a material liability. There has been no reservation of rights by any insurance carrier, and to the Knowledge of the Company, no such reservation is threatened, concerning the coverage of the Company with respect to any matter required to be disclosed pursuant to this Section 3.21.

Since January 1, 2011 neither the Company nor any Subsidiary has been denied insurance coverage by any insurance carrier.

Section 3.22 No Other Representations or Warranties. Except for the representations and warranties expressly made by the Company in this Article III, neither the Company nor any other Person makes any representation or warranty with respect to the Company or its Subsidiaries or their respective business, operations, assets, liabilities, condition (financial or otherwise) or prospects, notwithstanding the delivery or disclosure to Parent or any of its Affiliates or Representatives of any documentation, forecasts or other information with respect to any one or more of the foregoing.

#### ARTICLE IV

## REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Parent and Merger Sub jointly and severally represent and warrant to the Company, except as disclosed in the disclosure schedule delivered by the Parent to the Company simultaneously with the execution of this Agreement (the "Parent Disclosure Schedule") or except as set forth in (or incorporated by reference in) any of the Parent SEC Documents filed after December 31, 2011 and prior to the date of this Agreement (other than any non-Parent specific disclosures set forth under the heading "Risk Factors" and any non-Parent specific disclosures set forth in any "forward-looking statements" disclaimer or any other disclosures that are predictive, cautionary or forward-looking in nature, and provided that no such disclosure set forth in the Parent SEC Documents shall be deemed to modify or qualify the representations and warranties set forth in Section 4.2 (Capitalization), Section 4.3(a) (Authority), Section 4.3(b)(i) (Non-Contravention) and Section 4.4 (Governmental Approvals)):

# Section 4.1 Organization, Standing and Corporate Power.

(a) Each of Parent and Merger Sub is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware and has all requisite corporate power and authority necessary to own, lease and operate all of its properties and assets and to carry on its business as it is now being conducted. Each of Parent and Merger Sub is duly qualified to do business and is in good standing in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such qualification necessary, except where the failure to be so qualified or in good standing would not have a Parent Material Adverse Effect.

(b) Each of Parent's Subsidiaries is duly organized, validly existing and in good standing under the Laws of the jurisdiction of its organization and has all requisite limited liability company power and authority necessary to own, lease and operate all of its properties and assets and to carry on its business as it is now being conducted. All the outstanding shares of capital stock of, or other equity interests in, each such Subsidiary (except for directors' qualifying shares or the like) are owned directly or indirectly by Parent free and clear of all liens, pledges, security interests and transfer restrictions, other than the Permitted Liens and such transfer restrictions of general applicability as may be provided under the Securities Act, and other applicable securities Laws. There are no outstanding (i) securities of

Parent or any of its Subsidiaries convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, any Subsidiary of Parent, (ii) options, stock appreciation rights, warrants, restricted stock units, rights or other commitments or agreements to acquire from Parent or any of its Subsidiaries, or that obligate Parent or any of its Subsidiaries to issue, any capital stock of, or other equity or voting interest in, or any securities convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, any Subsidiary of Parent, (iii) obligations of Parent to grant, extend or enter into any subscription, warrant, right, convertible or exchangeable security or other similar agreement or commitment (whether payable in equity, cash or otherwise) relating to any capital stock of, or other equity or voting interest (including any voting debt) in, any Subsidiary of Parent (the items in clauses (i), (ii) and (iii), together with the capital stock of the Subsidiaries of Parent, being referred to collectively as "Parent Subsidiary Securities") or (iv) other obligations by Parent or any of its Subsidiaries to make any payments based on the price or value of any Parent Subsidiary Securities. There are no Contracts of any kind which obligate Parent or any of its Subsidiaries to repurchase, redeem or otherwise acquire any outstanding Parent Subsidiary Securities. Neither Parent nor any of its Subsidiaries is a party to any Contract restricting the transfer of, relating to the voting of, requiring registration of, or granting any preemptive rights, anti-dilution rights or rights of first refusal or similar rights with respect to any Parent Subsidiary Securities.

(c) Parent has made available to the Company complete and correct copies of the certificate of incorporation and by-laws of each of Parent and Merger Sub, in each case as amended to the date of this Agreement (the "Parent Charter Documents").

Section 4.2 Capitalization. The authorized capital stock of Parent consists of 50,000,000 shares of Parent Common Stock and 5,000,000 shares of convertible preferred stock, par value \$0.01 per share ("Parent Preferred Stock"). At the close of business on March 5, 2013, (a) 23,447,313 shares of Parent Common Stock were issued and outstanding, (b) no shares of Parent Common Stock were held by Parent in its treasury, (c) 1,927,259 shares of Parent Common Stock were issuable upon exercise of outstanding options to acquire Parent Common Stock under the Parent Stock Plans, (d) 708,163 shares of Parent Common Stock were subject to restricted stock unit awards outstanding under the Parent Stock Plans and (e) no shares of Parent Preferred Stock were issued or outstanding. All outstanding shares of Parent Common Stock have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights. Except as set forth in this Section 4.2, as of the date set forth above, there are (i) no outstanding shares of capital stock of, or other equity or voting interest in, Parent, (ii) no outstanding securities of Parent convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, Parent, (iii) no outstanding options, stock appreciation rights, warrants, restricted stock units, rights or other commitments or agreements to acquire from the Company, or that obligates the Company to issue, any capital stock of, or other equity or voting interest in, or any securities convertible into or exchangeable for shares of capital stock of, or other equity or voting interest in, Parent, (iv) no obligations of Parent to grant, extend or enter into any subscription, warrant, right, convertible or exchangeable security or other similar agreement or commitment (whether payable in equity, cash or otherwise) relating to any capital stock of, or other equity or voting interest (including any voting debt) in, Parent (the items in clauses (i), (iii), (iii) and (iv), together with the capital stock of Parent, being referred to collectively as "Parent Securities") and (v) no other obligations by Parent to make any payments based on the price or value of the Parent Securities. There are no outstanding Contracts of any

kind which obligate Parent or any of its Subsidiaries to repurchase, redeem or otherwise acquire any Parent Securities. Neither Parent nor any of its Subsidiaries is a party to any Contract restricting the transfer of, relating to the voting of, requiring registration of, or granting any preemptive rights, anti-dilution rights or rights of first refusal or similar rights with respect to any Parent Securities.

## Section 4.3 Authority; Noncontravention.

- (a) Each of Parent and Merger Sub has all necessary corporate power and authority to execute and deliver this Agreement, to perform their respective obligations hereunder and to consummate the Transactions. The execution and delivery of and performance by Parent and Merger Sub under this Agreement, and the consummation by Parent and Merger Sub of the Transactions, have been duly authorized and approved by all necessary corporate action by Parent and Merger Sub (including by the Parent Board and the Merger Sub Board) and adopted by Parent as the sole stockholder of Merger Sub, and no other corporate action on the part of Parent and Merger Sub is necessary to authorize the execution and delivery of and performance by Parent and Merger Sub under this Agreement and the consummation by them of the Transactions. This Agreement has been duly executed and delivered by Parent and Merger Sub and, assuming due authorization, execution and delivery hereof by the Company, constitutes a legal, valid and binding obligation of each of Parent and Merger Sub, enforceable against each of them in accordance with its terms, subject to the Bankruptcy and Equity Exception.
- (b) Neither the execution and delivery of this Agreement by Parent or Merger Sub, nor the consummation by Parent or Merger Sub of the Transactions, nor compliance by Parent or Merger Sub with any of the terms or provisions hereof, will (i) conflict with or violate any provision of the Parent Charter Documents or (ii) assuming that each of the consents, authorizations and approvals referred to in Section 4.4 (and any condition precedent to any such consent, authorization or approval has been satisfied) and each of the filings referred to in Section 4.4 are made and any applicable waiting periods referred to therein have expired, violate any Law applicable to Parent or any of its Subsidiaries or (iii) result in any breach of, or constitute a default (with or without notice or lapse of time or both) under, or give rise to any right of termination, amendment, acceleration or cancellation of, or result in loss of any benefit or the imposition of any additional payment or liability under, any Contract to which Parent, Merger Sub or any of their respective Subsidiaries is a party, except, in the case of clauses (ii) and (iii), as would not have a Parent Material Adverse Effect.
- (c) No vote or approval of the holders of any class or series of capital stock of Parent is necessary to adopt this Agreement and approve the Transactions.

Section 4.4 Governmental Approvals. Except for (a) the filing with the SEC of the Proxy Statement/Prospectus and of the Form S-4, and declaration of effectiveness of the Form S-4 by the SEC, and other filings required under, and compliance with other applicable requirements of, the Exchange Act, the rules of Nasdaq and applicable state securities and "blue sky" laws, (b) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware pursuant to the DGCL and (c) filings required under, and compliance with other applicable requirements of, the HSR Act, no consents or approvals of, or filings, declarations or registrations with, any Governmental Authority are necessary for the execution and delivery of

this Agreement by Parent and Merger Sub and the consummation by Parent and Merger Sub of the Transactions, other than as would not have a Parent Material Adverse Effect.

## Section 4.5 Parent SEC Documents; Undisclosed Liabilities .

(a) Parent has filed with or furnished to the SEC, on a timely basis, all registration statements, reports and proxy statements required to be filed or furnished since January 1, 2011, and, after the date of this Agreement and until the Effective Time, Parent will file all forms, reports, schedules, statements and documents with the SEC that are required to be filed by it under applicable Law at or prior to the time so required (collectively, and in each case including all exhibits and schedules thereto and documents incorporated by reference therein, as such statements and reports may have been amended since the date of their filing, the "Parent SEC Documents"). As of their respective effective dates (in the case of Parent SEC Documents that are registration statements filed pursuant to the requirements of the Securities Act) and as of their respective filing dates (in the case of all other Parent SEC Documents), or in the case of amendments thereto, as of the last such amendment, the Parent SEC Documents complied, or will comply, as the case may be, as of its filing date, in all material respects with the requirements of the Exchange Act, the Securities Act and the Sarbanes-Oxley Act, as the case may be, and the rules and regulations of the SEC thereunder, applicable to such Parent SEC Documents, and none of the Parent SEC Documents as of such respective dates (or, if amended, the date of the filing of such amendment, with respect to the disclosures that are amended) contained, or will contain, as the case may be, any untrue statement of a material fact or omitted, or will omit, as the case may be, to state a material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading. Except as set forth in Section 4.5 of the Parent Disclosure Schedule, neither Parent nor any of its Subsidiaries has received from the SEC any written comments or questions with respect to any of the Parent SEC Documents or any written notice from the SEC that such Parent SEC Documents are being reviewed or investigated, and, to the Knowledge of Parent, there is not, as of the date of this Agreement, any investigation or review being conducted by the SEC of any Parent SEC Documents. The principal executive officer of Parent and the principal financial officer of Parent (and each former principal executive officer or principal financial officer of Parent) have made the certifications required by Sections 302 and 906 of the Sarbanes-Oxley Act, and the rules and regulations of the SEC promulgated thereunder with respect to the Parent SEC Documents that were required to be accompanied by such certifications. For purposes of the preceding sentence, "principal executive officer" and "principal financial officer" shall have the meanings given to such terms in the Sarbanes-Oxley Act. None of Parent's Subsidiaries is required to file any forms, reports, schedules, statements or other documents with the SEC.

(b) None of the information supplied or to be supplied by Parent for inclusion or incorporation by reference in (i) the Form S-4 or any amendment or supplement thereto will, at the time such Form S-4 or any amendment or supplement thereto is filed with the SEC or at the time such Form S-4 becomes effective, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements made therein not misleading or (ii) the Proxy Statement/Prospectus will, at the date of mailing to holders of Company Common Stock and at the time of the Company Stockholders Meeting to be held in connection with the Merger, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order

to make the statements made therein, in the light of the circumstances under which they were made, not misleading. When filed, the Form S-4 will comply as to form in all material respects with the requirements of the Securities Act and the rules and regulations thereunder. No representation or warranty is made by Parent with respect to statements made or incorporated by reference therein based on information supplied by the Company for inclusion or incorporation by reference in the Proxy Statement/Prospectus or the Form S-4.

- (c) Except to the extent updated, amended, restated or corrected by a subsequent Parent SEC Document, as of their respective dates of filing with the SEC, the consolidated financial statements of the Parent included in the Parent SEC Documents (i) complied as to form in all material respects with all applicable accounting requirements and with the published rules and regulations of the SEC with respect thereto (except, in the case of unaudited statements, as permitted by Form 10-Q of the SEC), (ii) have been prepared in accordance with GAAP applied on a consistent basis during the periods involved (except (A) as may be indicated in the notes thereto or (B) as permitted by Regulation S-X) and (iii) present fairly in all material respects, the consolidated financial position of the Parent and its Subsidiaries, and the results of their operations and cash flows, for each of the dates and for the periods shown, in conformity with GAAP (subject to normal year-end adjustments in the case of any unaudited interim financial statements that were not, or are not expected to be, material in amount). Since January 1, 2011, Parent has not made any change in the accounting practices or policies applied in the preparation of its financial statements, except as required by GAAP, SEC rule or policy or applicable Law.
- (d) Parent has established and maintains disclosure controls and procedures and internal control over financial reporting (as such terms are defined in paragraphs (e) and (f), respectively, of Rule 13a-15 under the Exchange Act) as required by Rule 13a-15 under the Exchange Act. Parent's disclosure controls and procedures are reasonably designed to ensure that all material information (both financial and nonfinancial) required to be disclosed by Parent in the reports that it files or furnishes under the Exchange Act is recorded, processed, summarized and reported within the time periods specified in the rules and forms of the SEC, and that all such material information is accumulated and communicated to Parent's management as appropriate to allow timely decisions regarding required disclosure and to make the certifications required pursuant to Sections 302 and 906 of the Sarbanes-Oxley Act. The principal executive officer and principal financial officer of Parent have evaluated the effectiveness of Parent's disclosure controls and procedures and, to the extent required by applicable Law, presented in any applicable Parent SEC Document that is a report on Form 10-K or Form 10-Q, or any amendment thereto, its conclusions about the effectiveness of the disclosure controls and procedures as of the end of the period covered by such report or amendment based on such evaluation. Parent's management has completed an assessment of the effectiveness of Parent's internal control over financial reporting in compliance with the requirements of Section 404 of the Sarbanes-Oxley Act for the year ended December 31, 2012, and such assessment concluded that such controls were effective.
  - (e) Parent is in compliance in all material respects with the applicable corporate governance rules and regulations of Nasdaq.

(f) Neither the Parent nor any of its Subsidiaries has any liabilities which would be required to be reflected or reserved against on a consolidated balance sheet of the Parent prepared in accordance with GAAP or the notes thereto, except for liabilities (i) reflected or reserved against on the balance sheet of the Parent and its Subsidiaries as of the Balance Sheet Date (including the notes thereto) included in the Parent SEC Documents, (ii) incurred after the Balance Sheet Date in the ordinary course of business and that individually or in the aggregate would not reasonably be expected to have a Parent Material Adverse Effect or (iii) as contemplated by this Agreement or otherwise arising in connection with the Transactions.

**Section 4.6** <u>Absence of Certain Changes</u>. Between January 1, 2013 and the date of this Agreement, (a) except in connection with the Transactions, the business of the Parent and its Subsidiaries has been conducted in all material respects in the ordinary course of business consistent with past practice and (b) there has not been any fact, change, event, circumstance, occurrence or effect that, individually or in the aggregate, has had or would reasonably be expected to have a Parent Material Adverse Effect.

**Section 4.7** <u>Legal Proceedings</u>. Except for matters that, individually or in the aggregate, have not had a Parent Material Adverse Effect, (a) there is no pending or, to the Knowledge of Parent, threatened, legal or administrative proceeding, claim, suit or action against Parent or any of its Subsidiaries and (b) neither Parent nor any of its Subsidiaries is subject to any injunction, order, judgment, ruling or decree of any Governmental Authority.

## Section 4.8 Compliance With Laws; Permits .

(a) Except for instances of non-compliance that, individually or in the aggregate, would not have a Parent Material Adverse Effect, Parent and its Subsidiaries are, and since January 1, 2011 have been, in compliance with all Laws applicable to Parent or any of its Subsidiaries, including all laws relating to the collection of debts. Except for instances that, individually or in the aggregate, would not have a Parent Material Adverse Effect, since January 1, 2011, neither Parent nor any of its Subsidiaries has (i) received any written notice of any administrative, civil or criminal investigation or audit by any Governmental Authority relating to Parent or any of its Subsidiaries, (ii) received any written notice from any Governmental Authority alleging or relating to a possible violation by Parent or any of its Subsidiaries of any applicable Law, or (iii) provided any written notice to any Governmental Authority regarding any violation by Parent or any of its Subsidiaries of any applicable Law, and no such notice referred to in clause (i), (ii) or (iii) of this Section 4.8 remains outstanding or unresolved as of the date of this Agreement.

(b) Except as would not have a Parent Material Adverse Effect, Parent and each of its Subsidiaries hold, and are, and since January 1, 2011 have been, in compliance with, all licenses, franchises, permits, certificates, approvals and authorizations from Governmental Authorities required by Law for the conduct of their respective businesses as they are now being conducted (collectively, "Parent Permits"). There has occurred no violation of, default (with or without notice or lapse of time or both) under or event giving to others any right of revocation, non-renewal, adverse modification or cancellation of, with or without notice or lapse of time or both, any such Parent Permit, nor would any such revocation, non-renewal, adverse modification or cancellation result from the consummation of the Transactions, and no

suspension or cancellation of any such Parent Permits is pending or, to the Knowledge of Parent, threatened, except for such noncompliance, suspensions or cancellations that would not have a Parent Material Adverse Effect. Except for instances that, individually or in the aggregate, would not have a Parent Material Adverse Effect, neither Parent nor any of its Subsidiaries has received any written notice from any Governmental Authority regarding (a) any violation by Parent or any of its Subsidiaries of any Parent Permits or the failure to have any required Parent Permits, or (b) any revocation, cancellation or termination of any Parent Permits held by Parent or any of its Subsidiaries, and no such notice in either case remains outstanding or unresolved as of the date of this Agreement.

Section 4.9 Contracts. Except for matters that would not reasonably be expected to have a Parent Material Adverse Effect, (i) each Contract to which Parent or any of its Subsidiaries is a party or by which any of their respective properties or assets are bound that is listed as an exhibit to Parent's Annual Report on Form 10-K for the year ended December 31, 2012 (each such Contract a "Parent Material Contract") is valid and binding on Parent and any of its Subsidiaries to the extent Parent or such Subsidiary is a party thereto, as applicable, and to the Knowledge of Parent, each other party thereto, and is in full force and effect and enforceable in accordance with its terms (subject to the Bankruptcy and Equity Exception), (ii) Parent and each of its Subsidiaries, and, to the Knowledge of Parent, any other party thereto, has performed all material obligations required to be performed by it under each Parent Material Contract, (iii) no event has occurred that with notice or lapse of time or both would constitute a breach or default thereunder by Parent or any of its Subsidiaries, or, to the Knowledge of Parent, any other party thereto, except where such breach or default would not have a Parent Material Adverse Effect and (iv) neither Parent nor any of its Subsidiaries has received any written notice or other communication regarding any actual or possible violation or breach of or default under, or intention to cancel or modify, any Material Contract.

**Section 4.10 <u>Parent Common Stock</u>**. The shares of Parent Common Stock to be issued pursuant to the Merger have been duly authorized and upon consummation of the Transactions will be validly issued, fully paid and nonassessable and free and clear of all Liens.

**Section 4.11 <u>Brokers and Other Advisors</u>**. Except for Morgan Stanley & Co. LLC, no broker, investment banker, financial advisor or other Person is entitled to any broker's, financial advisor's or other similar fee, in connection with the Transactions based upon arrangements made by or on behalf of Parent or any of its Subsidiaries.

**Section 4.12** Ownership and Operations of Merger Sub. Parent owns beneficially and of record all of the outstanding capital stock of Merger Sub. Merger Sub was formed on March 4, 2013 solely for the purpose of engaging in the Transactions, has engaged in no other business activities and has conducted its operations only as contemplated hereby.

**Section 4.13 <u>Sufficient Funds</u>**. As and when needed (including at the Closing and all times thereafter), Parent shall have sufficient cash, marketable securities and other sources of immediately available funds necessary to consummate the Transactions on the terms contemplated by this Agreement, including any amounts payable pursuant to the refinancing of any outstanding Indebtedness of the Company and any other amounts incurred or otherwise payable by Parent, Merger Sub or the Surviving Corporation in connection with the

Transactions, and there is no restriction on the use of such cash for such purposes. Parent and Merger Sub acknowledge and agree that their obligations hereunder are not subject to any conditions regarding Parent's, Merger Sub's or any other Person's ability to obtain financing for the consummation of the Transactions.

Section 4.14 Share Ownership. Neither Parent nor Merger Sub has been, at any time during the three (3) years preceding the date hereof, an "interested stockholder" of the Company, as defined in Section 203 of the DGCL. As of the date of this Agreement, none of Parent, Merger Sub or their respective Subsidiaries owns (directly or indirectly, beneficially or of record) any shares of capital stock of the Company and none of Parent, Merger Sub or their respective Subsidiaries holds any rights to acquire directly or indirectly any shares of capital stock of the Company except pursuant to this Agreement and the Voting Agreement.

Section 4.15 No Other Representations or Warranties. Except for the representations and warranties expressly made by Parent in this Article IV, none of Parent, Merger Sub or any other Person makes any representation or warranty with respect to Parent or its Subsidiaries or their respective business, operations, assets, liabilities, condition (financial or otherwise) or prospects, notwithstanding the delivery or disclosure to the Company or any of its Affiliates or Representatives of any documentation, forecasts or other information with respect to any one or more of the foregoing.

### ARTICLE V

# **COVENANTS**

## **Section 5.1 Conduct of Business**.

- (a) Except as expressly contemplated or permitted by this Agreement, as required by applicable Law or as contemplated by Section 5.1(a) of the Company Disclosure Schedule, during the period from the date of this Agreement until the Effective Time, unless Parent otherwise consents (which consent shall not be unreasonably withheld, delayed or conditioned), each of the Company and each of its Subsidiaries shall conduct its business in the usual, regular and ordinary course and in all material respects consistent with past practice and use all reasonable efforts to preserve intact its present lines of business, maintain its rights and franchises and preserve satisfactory relationships with Governmental Authorities, employees, customers and suppliers, and, without limiting the generality of the foregoing, the Company shall not, and shall not permit any of its Subsidiaries to:
  - (i) issue, sell, or grant any shares of its capital stock, or any securities or rights convertible into, exchangeable or exercisable for, or evidencing the right to subscribe for any shares of its capital stock, or any rights, warrants or options to purchase any shares of its capital stock, or any securities or rights convertible into, exchangeable or exercisable for, or evidencing the right to subscribe for, any shares of its capital stock, except for the issuance of shares of Company Common Stock required to be issued upon exercise or settlement of Options outstanding prior to the date hereof;

- (ii) redeem, purchase or otherwise acquire any of its outstanding shares of capital stock, or any rights, warrants or options to acquire any shares of its capital stock, except (A) pursuant to commitments in effect as of the date hereof and set forth on Section 5.1 (a) of the Company Disclosure Schedule or (B) in connection with withholding to satisfy Tax obligations with respect to Options and the Company Stock Units, acquisitions in connection with the forfeiture of equity awards, or acquisitions in connection with the net exercise of Options;
- (iii) (A) declare, authorize, set aside for payment or pay any dividend on, or make any other distribution (whether in cash, stock, property or otherwise) in respect of, any shares of its capital stock, other than dividends paid by any Subsidiary of the Company to the Company or any wholly-owned Subsidiary of the Company or (B) adjust, split, combine, subdivide or reclassify any shares of its capital stock;
- (iv) (A) incur any Indebtedness, except for Indebtedness (1) incurred to replace, renew, extend, refinance or refund any existing Indebtedness the availability of which will expire pursuant to its terms on before the End Date, (2) for borrowed money incurred in the ordinary course of business consistent (including with regard to underwriting targets) with past practice to acquire any defaulted or charged-off accounts receivable portfolios, which, for the avoidance of doubt, shall mean defaulted or charged-off accounts receivable portfolios reasonably expected to have a net, unlevered ten percent (10%) internal rate of return assuming an eleven and four-tenths percent (11.4%) allocation of the Company's overhead costs, in addition to appropriate variable costs, related to such acquisitions (a "Permitted Portfolio Acquisition") and, in any event, not above (x) the limits set forth in the Company's Amended and Restated Credit Agreement dated November 14, 2011, between the Company, JPMorgan Chase Bank, N.A. and the other lenders thereto and (y) the amounts contemplated by the 2013 Company Plan, a copy of which is attached hereto as part of Section 5.1(a) of the Company Disclosure Schedule, or (3) among the Company and any of its wholly-owned Subsidiaries or among any of such Subsidiaries, (B) assume, guarantee, endorse or otherwise become liable or responsible (whether directly, contingently or otherwise) for the obligations of any other Person except with respect to obligations of direct or indirect wholly-owned Subsidiaries of the Company, (C) make any loans, advances or capital contributions to or investments in any other Person except for advances in the ordinary course of business consistent with past practice to employees of the Company or any of its Subsidiaries or (D) except in connection with any Permitted Portfolio Acquisition or pursuant to the terms of any Contract providing for Indebtedness existing as of the date of this Agreement or Indebtedness otherwise permitted to be incurred pursuant to this Section 5.1(a)(iv), mortgage or pledge any of its or its Subsidiaries' assets, tangible or intangible, or create or suffer to exist any Lien thereupon (other than Permitted Liens);
- (v) sell any of its properties or assets that are material to the Company and its Subsidiaries taken as a whole, except (A) sales, leases, rentals and licenses in the ordinary course of business (it being acknowledged that the Company and its Subsidiaries do not sell defaulted or charged-off accounts in the ordinary course and shall not sell any defaulted or charged-off accounts between the date hereof and the Effective Time), (B) pursuant to Contracts in force on the date of this Agreement and set forth on Section

- <u>5.1(a)</u> of the Company Disclosure Schedule, (C) abandonments or other dispositions of inventory, equipment or other assets that are no longer material to the conduct of the business of the Company or any of its Subsidiaries or (D) transfers among the Company and its Subsidiaries:
- (vi) make capital expenditures other than as budgeted (both as to time and amount) in the 2013 Company Plan (except with respect to budgeted amounts for July 2013, which shall not exceed \$568,000);
- (vii) make any acquisition (including by merger) of (A) the capital stock or a material portion of the assets of any other Person, except pursuant to Contracts in force on the date of this Agreement and set forth on Section 5.1(a) of the Company Disclosure Schedule or (B) any defaulted or charged-off accounts receivable portfolios (other than Permitted Portfolio Acquisitions (x) in any calendar month ending after the date hereof in an aggregate amount not in excess of \$23,000,000 per month and (y) in any three-consecutive-calendar month period commencing after the date hereof in an aggregate amount not in excess of \$45,000,000);
- (viii) increase in any respect the compensation of any of its (A) directors or executive officers, except as required pursuant to applicable Law or the terms of Company Plans or other employee benefit plans or arrangements in effect on the date of this Agreement or (B) other employees, other than increases in salaries, wages and benefits of employees (other than executive officer and directors) made in the ordinary course of business consistent with past practice in connection with the Company's annual or quarterly compensation review cycle (provided that payments of cash bonuses and other cash grants and cash awards made in the ordinary course consistent with past practice shall not constitute an increase in compensation);
- (ix) hire employees or independent contractors, in each case, with aggregate annual compensation in excess of \$150,000, except to fill vacancies existing as of the date hereof and to replace departed employees and independent contractors;
- (x) except to the extent required to comply with applicable Law or the existing terms of any Company Plan previously disclosed to Parent: (A) grant any severance, retention or termination pay to, or amend any existing severance, retention or termination arrangement with, any current or former director, officer or employee, (B) increase or accelerate the payment or vesting of, benefits payable under any existing severance, retention or termination pay policies or employment agreements, (C) enter into or amend any Company Plan, (D) establish, adopt or amend any collective bargaining agreement or any Company Plan, (E) pay any special bonus or special remuneration to any director, officer, consultant, independent contractor or employee, or (F) pay any benefit not required by any plan or arrangement as in effect as of the date hereof, other than, with respect to non-officer employees, in the ordinary course of business consistent with past practice;
- (xi) make any change to its methods of accounting, except as required by GAAP (or any interpretation thereof), Regulation S-X of the Exchange Act, as

required by a Governmental Authority or quasi-Governmental Authority (including the Financial Accounting Standards Board or any similar organization) or as required by applicable Law;

- (xii) amend the Company Charter Documents or organizational documents of any Subsidiary of the Company;
- (xiii) adopt a plan or agreement of complete or partial liquidation or dissolution;
- (xiv) take any action that would reasonably be expected to prevent or materially impede, interfere with, hinder or delay the consummation of the Transactions;
- (xv) except as reasonably contemplated by the scope of the 2013 Company Plan, change in any material respect (A) its collection strategy offered to consumers, including discount or settlement rates, (B) the frequency of mail volumes and telephone calls to consumers or change the percentage of legal suits to collect amounts owed by consumers; (C) the commission structure offered to third party agencies and law firms; (D) its current practice of recalling of accounts from law firms; or (E) the commission structure of Company Employees acting as collection agents;
- (xvi) enter into or amend or modify in any material respect, or consent to the termination of (other than at its stated expiry date), any (A) Material Contract or any other Contract that, if in effect as of the date hereof, would constitute a Material Contract, (B) Contract with a third party agency or law firm; (C) Contract with a data vendor; or (D) existing portfolio purchase and sale Contracts;
- (xvii) (A) make or change any material Tax election, (B) file any material amended Tax Return, (C) agree to any material adjustment of any Tax attribute, (D) change (or make a request to any Governmental Authority to change) any of its methods of reporting income or deductions for federal income Tax purposes, (E) file any claim for a material refund of Taxes, (F) consent to any extension or waiver of the limitation period applicable to any material Tax claim or assessment that would reasonably be expected to adversely affect Parent's Tax liability, (G) settle or compromise any suit, claim, action, investigation, proceeding or audit pending against or with respect to the Company or any of its Subsidiaries in respect of any material amount of Tax or enter into any material closing agreement that would reasonably be expected to adversely affect Parent's Tax liability for any taxable period beginning on or after the Effective Time or otherwise, or (H) enter into a Tax sharing, indemnification or allocation agreement (other than any Contract the primary purpose of which is not the sharing of Taxes);

(xviii) settle or enter into any settlement agreement with respect to any outstanding litigation or other dispute, except that, notwithstanding the foregoing, the Company may settle or enter into any settlement agreement with respect to any outstanding litigation or other dispute with any party (other than a Governmental Authority) where (A) the aggregate cash amount payable by the Company or its

Subsidiaries as a result of such settlement does not exceed \$100,000 and (B) the terms of such settlement or settlement agreement (not involving the payment of money) do not require the taking (or omission) of any action by the Company or any of its Subsidiaries which is materially inconsistent with past practice regarding actions (not involving the payment of money) taken or omitted to be taken by any of them in connection with settlements and settlement agreements relating to substantially similar litigation, claims, suits, proceedings or other disputes;

- (xix) enter into any new line of business; or
- (xx) agree in writing to take any of the foregoing actions.
- (b) Parent shall not, and shall not permit any of its Subsidiaries to, take, or agree or commit to take, any action that would reasonably be expected to, individually or in the aggregate, prevent, materially delay or materially impede, interfere with, hinder or delay the consummation by Parent or any of its Subsidiaries of the Transactions.

### Section 5.2 Preparation of the Proxy Statement/Prospectus; Stockholders Meeting.

(a) As promptly as reasonably practicable following the date of this Agreement (and in any event within fifteen (15) Business Days of the date hereof), Parent and the Company shall prepare and file with the SEC the Proxy Statement/Prospectus, and Parent shall prepare and file with the SEC the Form S-4, in which the Proxy Statement/Prospectus will be included as Parent's prospectus. Each of Parent and the Company shall use reasonable best efforts to have the Form S-4 declared effective under the Securities Act as promptly as reasonably practicable after such filing and to keep the Form S-4 effective as long as necessary to consummate the Transactions. The Company will cause the Proxy Statement/Prospectus together with the Election Form to be mailed to the Company's stockholders as promptly as reasonably practicable (and in no event more than five (5) Business Days) after the Form S-4 is declared effective under the Securities Act. No filing of, or amendment or supplement to, or correspondence with the SEC or its staff with respect to, the Form S-4 or the Proxy Statement/Prospectus will be made by Parent or the Company, as applicable, without providing the other party a reasonable opportunity to review and comment thereon and without the consent of the other party, which shall not be unreasonably withheld, conditioned or delayed; provided, however, that the foregoing shall not apply to any filings with the SEC deemed to supplement the Form S-4 or any document which forms a part thereof through its incorporation by reference therein or with respect to a Takeover Proposal, a Superior Proposal, a Company Adverse Recommendation Change or any matters relating thereto. Parent or the Company, as applicable, will advise the other party promptly after it receives oral or written notice of the time when the Form S-4 has become effective or any supplement or amendment has been filed, the issuance of any stop order, the suspension of the qualification of the Parent Common Stock issuable in connection with the Merger for offering or sale in any jurisdiction, or any oral or written request by the SEC for amendment of the Proxy Statement/Prospectus or the Form S-4 or comments thereon and responses thereto or requests by the SEC for additional information, and will promptly provide the other with copies of any written communication from the SEC or any state securities commission. If, at any time prior to the Effective Time, any information relating to

Parent or the Company, or any of their respective Affiliates, officers or directors, should be discovered by Parent or the Company which should be set forth in an amendment or supplement to any of the Form S-4 or the Proxy Statement/Prospectus, so that any of such documents would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the party which discovers such information shall promptly notify the other parties hereto and an appropriate amendment or supplement describing such information shall be promptly filed with the SEC and, to the extent required by Law, disseminated to the stockholders of the Company.

- (b) The Company shall, as soon as reasonably possible after the Form S-4 has been declared effective, duly give notice of, convene and hold a meeting of its stockholders to consider the approval of the Transactions and such other matters as may be then legally required (including any adjournment or postponement thereof, the "Company Stockholders Meeting"); provided, however, that the Company shall be permitted to delay or postpone convening the Company Stockholders Meeting if the Company Board (or a duly authorized committee thereof), after consultation with outside legal counsel, reasonably determines that failure to delay or postpone the Company Stockholders Meeting would reasonably be expected to be a breach of its fiduciary duties under applicable Law.
- (c) The Company shall, through the Company Board (or the Review Committee or other duly authorized committee of the Company Board), but subject to the right of the Company Board (or the Review Committee or other duly authorized committee of the Company Board) to make a Company Adverse Recommendation Change pursuant to Section 5.3, (i) recommend to its stockholders that the Company Stockholder Approval be given (the "Company Board Recommendation") and (ii) include the Company Board Recommendation in the Proxy Statement/Prospectus.

## Section 5.3 Solicitation; Change in Recommendation.

- (a) Subject to Section 5.3(c), the Company shall, and shall cause its and its Subsidiaries, directors, officers and employees to, and shall instruct and use its reasonable best efforts to cause its other Representatives to, (i) immediately cease and cause to be terminated any and all existing activities, discussions or negotiations with any Person (other than Parent and its Representatives) conducted heretofore with respect to any Takeover Proposal, (ii) request the prompt return or destruction of all confidential information previously furnished to any Person prior to the date hereof with respect to any Takeover Proposal or potential Takeover Proposal, and (iii) terminate the access granted prior to the date hereof of any Persons (other than Parent and its Representatives) to any "data room" hosted by the Company or any of its Subsidiaries or Representatives relating to any Takeover Proposal.
- (b) Subject to Section 5.3(c) and except as otherwise permitted by this Section 5.3, from the date of this Agreement until the earlier of the Effective Time or the date, if any, on which this Agreement is terminated pursuant to Section 7.1, the Company shall not, and shall cause its and its Subsidiaries, directors, officers and employees not to, and shall instruct and use its reasonable best efforts to cause its other Representatives not to, directly or indirectly, (i) solicit, initiate, knowingly encourage or knowingly facilitate the making, submission or

consummation of any Takeover Proposal or (ii) enter into, or otherwise participate in any discussions (except to notify such Person of the existence of the provisions of this <u>Section 5.3</u>) or negotiations regarding, or furnish to any Person any material non-public information in connection with, any Takeover Proposal.

(c) Notwithstanding anything to the contrary in Section 5.3(a) or Section 5.3(b), Parent and Merger Sub hereby agree that, from the date hereof until 11:59 p.m. New York City time on the later of (x) March 27, 2013 (the "Go Shop End Date") and (y) if an Excluded Party submits a Superior Proposal on or before the Go Shop End Date, 11:59 p.m. New York City time on the date on which the SP Notice Period with respect to that Superior Proposal expires (such later date, the "Excluded Party End Date"), the Company and its Subsidiaries and their respective Representatives shall have the right to (i) solicit, initiate, encourage and facilitate, whether publicly or otherwise, Takeover Proposals, including by way of providing access to non-public information pursuant to one or more Acceptable Confidentiality Agreements (provided that the Company shall promptly provide to Parent any non-public information concerning the Company or its Subsidiaries that is provided to any Person given such access which was not previously provided to Parent or its Representatives) and (ii) enter into, or otherwise participate in discussions or negotiations with respect to Takeover Proposals (or inquiries, proposals or offers or other efforts or attempts that could lead to a Takeover Proposal) or otherwise cooperate with or assist or participate in, or facilitate any such inquiries, proposals, offers, efforts, attempts, discussions or negotiations, in each case with any Excluded Party.

(d) Notwithstanding anything to the contrary contained in this Agreement, if at any time prior to obtaining the Company Stockholder Approval, the Company or any of its Subsidiaries, or any of its or their respective Representatives receives a written Takeover Proposal from any Person or group of Persons which Takeover Proposal was made on or after the date hereof and did not result from a material breach of this Section 5.3, the Company, the Company Board (or the Review Committee or other duly authorized committee of the Company Board) and their Representatives may engage in negotiations and discussions with, or furnish any information and other access to, any Person making such Takeover Proposal and any of its Representatives or potential sources of financing if the Company Board (or the Review Committee or other duly authorized committee of the Company Board) determines in good faith, after consultation with the Company's outside legal and financial advisors, that such Takeover Proposal is or could reasonably be expected to lead to a Superior Proposal and that the failure to engage in negotiations and discussions with, or furnish any information and other access to, such Person would reasonably be expected to be a breach of its fiduciary duties under applicable Law; provided that prior to furnishing any material non-public information to any such Person, the Company receives from the Person making such Takeover Proposal an Acceptable Confidentiality Agreement. The Company shall promptly (and in any event within 48 hours of receipt) notify Parent of any Takeover Proposal, together with a description of the material terms and conditions of such Takeover Proposal, the identity of the Person or group making any such Takeover Proposal and a copy of any written proposal, offer or draft agreement provided by such Person. The Company shall keep Parent reasonably informed on a timely basis (including within 48 hours after the occurrence of any material amendment, modification, development, discussion or negotiation relating to any such Takeover Proposal) of the status and material details of any such Takeover Proposal. The Company shall promptly provide to Parent any nonpublic

information concerning the Company or its Subsidiaries that is provided to such Person (to the extent such information was not previously provided to Parent or its Representatives). Notwithstanding anything to the contrary in this Agreement, but subject to  $\underline{Section 5.3(g)}$ , in no event shall Parent (whether pursuant to  $\underline{Section 7.1(c)(ii)(B)}$  or otherwise) make more than one (1) request to the Company Board (or the Review Committee or other duly authorized committee of the Company Board) to publicly reaffirm its approval and recommendation of the Merger and this Agreement in respect of (x) each Takeover Proposal and (y) each material amendment to or modification of a Takeover Proposal.

- (e) Except as otherwise provided in <u>Section 5.3(f)</u>, the Company Board shall not (i)(A) withdraw or modify, or publicly propose to withdraw or modify, in a manner adverse to Parent, the Company Board Recommendation, or (B) approve or recommend, or publicly propose to approve or recommend to the stockholders of the Company a Takeover Proposal (any action described in this <u>clause (i)</u> being referred to as a "<u>Company Adverse Recommendation Change</u>") or (ii) authorize the Company or any of its Subsidiaries to enter into any letter of intent, merger, acquisition or similar agreement with respect to any Takeover Proposal (other than an Acceptable Confidentiality Agreement).
- (f) Notwithstanding anything to the contrary in this Agreement, at any time prior to obtaining the Company Stockholder Approval the Company Board (or the Review Committee or other duly authorized committee of the Company Board) may make a Company Adverse Recommendation Change in response to either (x) a Superior Proposal or (y) an Intervening Event (and, solely with respect to a Superior Proposal, terminate this Agreement pursuant to Section 7.1(d)(ii)), if:
  - (i) the Company Board (or the Review Committee or other duly authorized committee of the Company Board) determines in good faith, after consultation with its outside legal counsel, that the failure to take such action would reasonably be expected to result in a breach of its fiduciary duties under applicable Law (after taking into account all adjustments to the terms of this Agreement that may be offered by Parent pursuant to this <u>Section 5.4(f)</u>);
  - (ii) solely with respect to a Superior Proposal: (A) the Company provides Parent prior written notice of its intent to make any Company Adverse Recommendation Change or terminate this Agreement pursuant to Section 7.1(d)(ii), together with a copy of the acquisition agreement (and any other relevant transaction documents) with respect to such Superior Proposal, if any, at least three (3) Business Days (such period, as it may be extended in accordance with this Section 5.3(f), the "SP Notice Period") prior to taking such action, to the effect that the Company Board (or the Review Committee or other duly authorized committee of the Company Board) has received a Superior Proposal and, absent any revision to the terms and conditions of this Agreement, the Company Board (or the Review Committee or other duly authorized committee of the Company Board) has resolved to effect a Company Adverse Recommendation Change or terminate this Agreement pursuant to Section 7.1(d)(ii), which notice shall specify the basis for such Company Adverse Recommendation Change or termination, including the material terms of the Superior Proposal (a "Notice of Superior Proposal") (it being understood that such Notice of Superior Proposal shall not

in itself be deemed a Company Adverse Recommendation Change and that any material revision or amendment to the terms of such Superior Proposal shall require a new notice, which notice shall cause the SP Notice Period to run for three (3) Business Day from the date of such new notice); (B) during the SP Notice Period, if requested by Parent, the Company shall have made its Representatives reasonably available to discuss with Parent's Representatives any proposed modifications to the terms and conditions of this Agreement so that such Takeover Proposal would cease to constitute a Superior Proposal; (C) Parent has not, within the SP Notice Period, made a written, binding and irrevocable offer capable of being accepted by the Company to alter the terms or conditions of this Agreement such that such Takeover Proposal would cease to constitute a Superior Proposal and (D) the Company Board (or the Review Committee or other duly authorized committee of the Company Board), after taking into account any modifications to the terms of this Agreement and the Merger agreed to by Parent and Merger Sub after receipt of such notice, continues to believe that such Takeover Proposal constitutes a Superior Proposal; and

(iii) solely with respect to any Intervening Event: (A) at least three (3) Business Days (such period, the "IE Notice Period." and, together with the SP Notice Period, the "Notice Period.") before making a Company Adverse Recommendation Change with respect to such Intervening Event, the Company notifies Parent in writing of its intention to do so, specifies the reasons therefor and provides a description of such Intervening Event; and (B) if Parent makes a written, binding and irrevocable offer capable of being accepted by the Company during the IE Notice Period to alter the terms or conditions of this Agreement, the Company Board (or the Review Committee or other duly authorized committee of the Company Board), after taking into consideration the adjusted terms and conditions of this Agreement as proposed by Parent, continues to determine in good faith (after consultation with outside counsel) that the failure to make such Company Adverse Recommendation Change would reasonably be expected to result in a breach of its fiduciary duties under applicable Law.

During any Notice Period prior to its effecting a Company Board Recommendation Change or terminating this Agreement as referred to above, the Company shall, and shall instruct and use its reasonable best efforts to cause its financial and legal advisors to, negotiate with Parent in good faith (to the extent Parent seeks to negotiate) regarding any revisions to the terms of the Transactions proposed by Parent.

(g) Nothing contained in this Agreement shall prohibit the Company or the Company Board (or the Review Committee or other duly authorized committee of the Company Board) from (i) taking and disclosing to the stockholders of the Company a position contemplated by Rule 14e-2(a) under the Exchange Act or making a statement contemplated by Item 1012(a) of Regulation M-A or Rule 14d-9 under the Exchange Act, (ii) making any disclosure to the stockholders of the Company if the Company Board (or the Review Committee or other duly authorized committee of the Company Board) determines in good faith, after consultation with its outside legal counsel, that the failure to make such disclosure would be reasonably likely to be inconsistent with applicable Law, (iii) informing any Person of the existence of the provisions contained in this Section 5.3 or (iv) making any "stop, look and listen" communication to the stockholders of the Company pursuant to Rule 14d-9(f) under the

Exchange Act (or any similar communication to the stockholders of the Company); provided, however, that notwithstanding anything in this Section 5.3(g), the Company Board shall not make a Company Adverse Recommendation Change except in accordance with Section 5.3(f); and provided, further, however, that if any such statement or disclosure made by the Company pursuant to clauses (i) or (ii) above contains anything beyond a factually accurate statement or a statement that would reasonably be expected to be materially adverse to Parent's or the Company's ability to consummate the Transactions, Parent shall be entitled to request in writing that the Company Board publicly reaffirms the Company Board Recommendation and it shall be deemed to be a Company Adverse Recommendation Change if the Company Board fails to do so within five (5) Business Days of the date of such request. Any requests for reaffirmation of the Company Board Recommendation pursuant to this Section 5.3(g) shall not count as a request by Parent for purposes of the last sentence of Section 5.3(d). No disclosures permitted under this Section 5.3(g) shall be, in themselves, a breach of Section 5.3 or a basis for Parent to terminate this Agreement pursuant to Article VII. Notwithstanding this Section 5.3 or any other provision of this Agreement to the contrary, at any time prior to the receipt of Company Stockholder Approval, the Company Board (or the Review Committee or other duly authorized committee of the Company Board) may grant a waiver or release under, or determine not to enforce any standstill agreement of the Company (other than a standstill agreement with any Excluded Party) if the Company Board (or the Review Committee or other duly authorized committee of the Company Board) determines in good faith (after consultation with its outside legal counsel) that the failure to take such action would reasonably be expected to result in a breach of their fiduciary duties under applicable Law; provided, however, that the Company shall give Parent notice of the grant by the Company Board (or the Review Committee or other duly authorized committee of the Company Board) at least 24 hours before granting any such waiver or release.

(h) As used in this Agreement, "Takeover Proposal" shall mean any bona fide inquiry, proposal or offer from any Person (other than Parent, Merger Sub and any of its Affiliates thereof) or "group" (as such term is defined in the Exchange Act) to purchase or otherwise acquire, in a single transaction or series of related transactions, (i) assets of the Company and its Subsidiaries (including securities of Subsidiaries, but excluding sales of assets in the ordinary course of business) that account for 20% or more of the Company's consolidated assets (based on the fair market value thereof) or from which 20% or more of the Company's revenues, earnings or cash flows on a consolidated basis are derived or (ii) 20% or more of the outstanding Company Common Stock pursuant to a merger, consolidation or other business combination, sale of shares of capital stock, tender offer, exchange offer or similar transaction.

(i) As used in this Agreement, "Superior Proposal" shall mean any bona fide written Takeover Proposal on terms which the Company Board (or the Review Committee or other duly authorized committee of the Company Board) determines in good faith, after consultation with the Company's outside legal counsel and independent financial advisors, (i) to be more favorable to the holders of Company Common Stock than the Transactions from a financial point of view (after taking into account any revisions to this Agreement set forth in any written, binding and irrevocable offer by Parent capable of being accepted by the Company), taking into account, to the extent applicable, the legal, financial, regulatory, timing and other aspects of such proposal and this Agreement that the Company Board (or the Review Committee or other duly authorized committee of the Company Board) considers relevant, (ii) the financing of which is fully committed or reasonably likely to be obtained and (iii) if accepted, is

reasonably likely to be consummated (in the case of each of the foregoing <u>clauses (ii)</u> and <u>(iii)</u>, as determined in good faith by the Company Board (or the Review Committee or other duly authorized committee of the Company Board)); <u>provided</u> that for purposes of the definition of Superior Proposal, the references to "20%" in the definition of Takeover Proposal shall be deemed to be references to "50%."

## Section 5.4 Reasonable Best Efforts.

(a) Subject to the terms and conditions of this Agreement, each of the Company, Parent and Merger Sub shall use its respective reasonable best efforts to (i) cause the Transactions to be consummated as soon as practicable (including by taking all actions set forth on Section 5.4(a) of the Parent Disclosure Schedule), (ii) make promptly any required submissions and filings under applicable Antitrust Laws with respect to the Transactions, (iii) promptly furnish information required in connection with such submissions and filing under such Antitrust Laws, (iv) keep the other parties reasonably informed with respect to the status of any such submissions and filings under Antitrust Laws, including with respect to, (A) the receipt of any non-action, action, clearance, consent, approval or waiver, (B) the expiration of any waiting period, (C) the commencement or proposed or threatened commencement of any investigation, litigation or administrative or judicial action or proceeding under Antitrust Laws, and (D) the nature and status of any objections raised or proposed or threatened to be raised under Antitrust Laws with respect to the Transactions and (v) make all necessary notices to and obtain all actions or non-actions, approvals, consents, waivers, registrations, permits, authorizations and other confirmations from any Governmental Authority or third party necessary, proper or advisable to consummate the Transactions as soon as practicable (but in any event on or prior to the End Date). For purposes hereof, "Antitrust Laws" means the Sherman Act, the Clayton Act, the HSR Act, the Federal Trade Commission Act and all other applicable Laws issued by a Governmental Authority that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening of competition through merger or acquisition.

(b) In furtherance and not in limitation of the foregoing, (i) each party hereto agrees to (A) make an appropriate filing of a Notification and Report Form pursuant to the HSR Act with respect to the Transactions as soon as practicable and in any event within ten (10) Business Days of the date hereof (unless the parties otherwise agree to a different date), (B) supply as soon as practicable any additional information and documentary material that may be requested pursuant to the HSR Act and (C) use its reasonable best efforts to take, or cause to be taken, all other actions consistent with this Section 5.4(b) necessary to cause the expiration or termination of the applicable waiting periods under the HSR Act (including any extensions thereof) as soon as practicable and (ii) each party agrees to (A) supply as soon as practical any additional information and documentary material that may be required or requested by any Governmental Authority and (B) use its reasonable best efforts to take or cause to be taken all other actions consistent with this Section 5.4(b) as necessary to obtain any necessary approvals, consents, waivers, permits, authorizations or other actions or non-actions from each Governmental Authority as soon as practicable.

(c) The Company, Parent and Merger Sub shall, (i) promptly notify the other parties hereto of, and if in writing, furnish the others with copies of (or, in the case of

oral communications, advise the others of the contents of) any communication to such Person from a Governmental Authority and permit the others to review and discuss in advance (and to consider in good faith any comments made by the others in relation to) any proposed written communication to a Governmental Authority, (ii) keep the others reasonably informed of any developments, meetings or discussions with any Governmental Authority in respect of any filings, investigation, or inquiry concerning the Transactions and (iii) not independently participate in any meeting or discussions with a Governmental Authority in respect of any filings, investigation or inquiry concerning the Transactions without giving the other party prior notice of such meeting or discussions and, unless prohibited by such Governmental Authority, the opportunity to attend or participate. However, each of Parent and Company may designate any non-public information provided to any Governmental Authority as restricted to "Outside Antitrust Counsel" only and any such information shall not be shared with employees, officers or directors or their equivalents of the other party without approval of the party providing the non-public information.

(d) In furtherance and not in limitation of the foregoing, Parent and Merger Sub agree to take promptly any and all steps necessary to avoid, eliminate or resolve each and every impediment and obtain all clearances, consents, approvals and waivers under Antitrust Laws that may be required by any Governmental Authority, so as to enable the parties to close the Transactions as soon as practicable (and in any event no later than three (3) Business Days prior to the End Date), including committing to and effecting, by consent decree, hold separate orders, trust, or otherwise, the termination, relinquishment, modification or waiver of existing relationships, ventures, contractual rights, obligations or other arrangements of Parent, the Company, the Surviving Corporation or their respective Subsidiaries (each a "Remedial Action"); provided, however, that any Remedial Action may, at the discretion of Parent and the Company, be conditioned upon consummation of the Transactions; provided, further, that none of Parent, Merger Sub or any of their Subsidiaries shall be required to, and the Company and its Subsidiaries may not, without the prior written consent of Parent, become subject to, consent to or agree to, or otherwise take any action with respect to, any requirement, condition, limitation, understanding, agreement or order to sell, license, assign, transfer, divest, hold separate or otherwise dispose of any assets, business or portion of business of the Surviving Corporation, Parent, Merger Sub or any of their respective Subsidiaries.

(e) In furtherance and not in limitation of the foregoing, in the event that any litigation or other administrative or judicial action or proceeding is commenced, threatened or is foreseeable challenging any of the Transactions and such litigation, action or proceeding seeks, or would reasonably be expected to seek, to prevent, materially impede or materially delay the consummation of the Transactions, Parent shall use reasonable best efforts, including a Remedial Action, to avoid or resolve any such litigation, action or proceeding and each of the Company, Parent and Merger Sub shall cooperate with each other and shall use its respective reasonable best efforts to contest and resist any such litigation, action or proceeding and to have vacated, lifted, reversed or overturned any decree, judgment, injunction or other order, whether temporary, preliminary or permanent, that is in effect and that prohibits, prevents or restricts consummation of the Transactions as promptly as practicable and in any event no later than three (3) Business Days prior to the End Date.

(f) Neither Parent nor Merger Sub shall, nor shall they permit their respective Subsidiaries to, acquire or agree to acquire any rights, assets (other than portfolios of defaulted receivables), business, Person or division thereof (through acquisition, license, joint venture, collaboration or otherwise), if such acquisition, would reasonably be expected to increase the risk of not obtaining any applicable clearance, consent, approval or waiver under Antitrust Laws with respect to the Transactions.

Section 5.5 <u>Public Announcements</u>. The initial press release with respect to the execution of this Agreement shall be a joint press release to be reasonably agreed upon by Parent and the Company. Following such initial press release, Parent and the Company shall consult with each other before issuing, and give each other the opportunity to review, comment upon and approve, any press release or other public statements with respect to the Transactions and shall not issue any such press release or make any such public statement prior to such consultation, except as such party may reasonably conclude may be required by applicable Law, court process or by obligations pursuant to any listing agreement with any national securities exchange or national securities quotation system (and then only after as much advance notice and consultation as is feasible); <u>provided</u>, <u>however</u>, that the restrictions set forth in this <u>Section 5.5</u> shall not apply to any release or public statement (a) made or proposed to be made by the Company in connection with a Takeover Proposal, a Superior Proposal or a Company Adverse Recommendation Change or any action taken pursuant thereto or (b) in connection with any dispute between the parties regarding this Agreement or the Transactions. In addition, the Company and its Subsidiaries shall not make any internal announcements to their employees regarding the Transactions without the prior review and approval of Parent (such approval not to be unreasonable withheld, conditioned or delayed) except to the extent such announcements are consistent in all material respects with the prior public disclosures regarding the Transactions.

### Section 5.6 Access to Information; Confidentiality.

(a) Subject to applicable Laws relating to the exchange of information, from the date hereof until the earlier of the Effective Time or the date on which this Agreement is terminated in accordance with its terms, the Company and Parent shall afford to the other party and its Representatives reasonable access during normal business hours to the Company's or Parent's, as applicable, properties, books, Contracts and records and the Company and Parent shall furnish promptly to the other party such information concerning its business and properties as such party may reasonably request (other than any publicly available document filed by it pursuant to the requirements of federal or state securities Laws); provided that the Company, Parent and their respective Representatives shall conduct any such activities in such a manner as not to interfere unreasonably with the business or operations of the other party; and provided, further, that the Company and Parent shall not be obligated to provide such access or information if the Company or Parent, as applicable, determines, in its reasonable judgment, that doing so would violate applicable Law or a Contract or obligation of confidentiality owing to a third-party (provided that the Company or Parent, as applicable, shall use commercially reasonable efforts to obtain any consents of third parties that are necessary to allow such information to be disclosed to the other) or jeopardize the protection of the attorney-client privilege (provided that the Company or Parent, as applicable, shall use commercially reasonable efforts to allow for such access or disclosure in a manner that does not result in loss of the attorney-client privilege). Until the Effective Time, the information provided will be subject to the terms of the

Confidentiality Agreement, dated as of December 12, 2012, between Parent and the Company (the "Confidentiality Agreement"), and, without limiting the generality of the foregoing, Parent and Company shall not, and Parent and Company shall cause their respective Representatives not to, use such information for any purpose unrelated to the consummation of the Transactions. No investigation shall affect the Company's, Parent's or Merger Sub's representations and warranties contained herein, or limit or otherwise affect the remedies available to Parent, Merger Sub or the Company pursuant to this Agreement.

(b) The Company and Parent acknowledge and agree that each (i) had an opportunity to discuss the business of other party with the management of the other party, (ii) has had access to the books and records, facilities, contracts and other assets of the other party which it and its affiliates have requested to review, (iii) has been afforded the opportunity to ask questions of and receive answers from officers of the other party and (iv) has conducted its own independent investigation of the other party, its businesses and the Transactions.

# Section 5.7 Notification of Certain Matters.

- (a) The Company shall give prompt notice to Parent, and Parent shall give prompt notice to the Company, of any notice or other communication received by such party from any Governmental Authority in connection with the Transactions or from any Person alleging that the consent of such Person is or may be required in connection with the Transactions, if the subject matter of such communication or the failure of such party to obtain such consent would reasonably be expected to be material to the Company, the Surviving Corporation or Parent or its ability to consummate the Transactions.
- (b) From and after the date of this Agreement until the Effective Time, each party hereto shall promptly notify the other party hereto of (i) the occurrence or non-occurrence of any event that would be likely to cause any condition to the obligations of any party to effect the Merger or any other Transaction not to be satisfied, or (ii) the failure of the Company, Parent or Merger Sub, as the case may be, to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it pursuant to this Agreement which would reasonably be expected to result in any condition to the obligations of any party to effect the Merger or any other Transaction not to be satisfied; provided, however, that the delivery of any notice pursuant to this Section 5.7(b) shall not cure any breach of any representation, warranty, covenant or agreement contained in this Agreement or otherwise limit or affect the remedies available hereunder to the party receiving such notice.

## Section 5.8 Indemnification and Insurance .

(a) For a period beginning at the Effective Time and ending on the sixth (6 th) anniversary of the Effective Time, the Surviving Corporation shall, and Parent shall cause the Surviving Corporation to, (i) indemnify, defend and hold harmless each current and former director, officer and employee of the Company and any of its Subsidiaries and each person who served as a director, officer, member, trustee or fiduciary of another corporation, partnership, joint venture, trust, pension or other employee benefit plan or enterprise if such service was at the request or for the benefit of the Company or any of its Subsidiaries (each, an "Indemnitee" and, collectively, the "Indemnitees") against all claims, liabilities, losses, damages,

judgments, fines, penalties, costs (including amounts paid in settlement or compromise) and expenses (including fees and expenses of legal counsel) in connection with any actual or threatened claim, suit, action, proceeding or investigation (whether civil, criminal, administrative or investigative) (each, a "Claim"), whenever asserted, arising out of, relating to or in connection with any action or omission relating to their position with the Company or its Subsidiaries occurring or alleged to have occurred before or at the Effective Time (including any Claim relating in whole or in part to the Agreement or the Transactions), to the fullest extent permitted under applicable Law and (ii) assume all obligations of the Company and its Subsidiaries to the Indemnitees in respect of limitation of liability, exculpation, indemnification and advancement of expenses as provided in (A) the Company Charter Documents and the respective organizational documents of each of the Company's Subsidiaries as currently in effect and (B) any indemnification agreements with an Indemnitee that are in effect as of the date hereof and set forth on Section 5.8 of the Company Disclosure Schedule, which shall in each case survive the Transactions and continue in full force and effect to the extent permitted by applicable Law. Without limiting the foregoing, at the Effective Time, the Surviving Corporation shall, and Parent shall cause the Surviving Corporation to, cause the certificate of incorporation and by-laws of the Surviving Corporation to include provisions for limitation of liabilities of directors and officers, indemnification, advancement of expenses and exculpation of the Indemnitees no less favorable to the Indemnitees than as set forth in the Company Charter Documents in effect on the date of this Agreement, which provisions shall not be amended, repealed or otherwise modified in a manner that would adversely affect the rights thereunder of the Indemnitees except as required by applicable Law for a period of six (6) years foll

(b) For a period beginning at the Effective Time and ending on the sixth (6 th) anniversary of the Effective Time, the Surviving Corporation shall, and Parent shall cause the Surviving Corporation to, pay and advance to an Indemnitee any expenses (including fees and expenses of legal counsel) in connection with any Claim relating to any acts or omissions covered under this Section 5.8 or the enforcement of an Indemnitee's rights under this Section 5.8 as and when incurred to the fullest extent permitted under applicable Law; provided that the person to whom expenses are advanced provides an undertaking to repay such expenses (but only to the extent required by applicable Law, the Company Charter Documents, applicable organizational documents of Subsidiaries of the Company Disclosure Schedule). Any determination required to be made with respect to whether an Indemnitee's conduct complies with an applicable standard under applicable Law, the Company Charter Documents, applicable organizational documents of Subsidiaries of the Company, or applicable indemnification agreements, as the case may be, shall be made by independent legal counsel selected by the Indemnitee and reasonably acceptable to Parent.

(c) For a period of six (6) years from the Effective Time, the Surviving Corporation shall, and Parent shall cause the Surviving Corporation to, maintain in effect the coverage provided by the policies of directors' and officers' liability insurance and fiduciary liability insurance in effect as of the date hereof maintained by the Company and its Subsidiaries with respect to matters arising on or before the Effective Time either through the Company's existing insurance provider or another provider reasonably selected by Parent; provided, however, that, after the Effective Time, Parent shall not be required to pay annual premiums in excess of 200% of the last annual premium paid by the Company prior to the date

hereof in respect of the coverages required to be obtained pursuant hereto (which premium is set forth on <u>Section 3.21</u> of the Company Disclosure Schedule), but in such case shall purchase as much coverage as reasonably practicable for such amount; and <u>provided</u>, <u>further</u>, that in lieu of the foregoing insurance coverage, the Company may purchase "tail" insurance coverage, at a cost no greater than the aggregate amount which the Surviving Corporation would be permitted to spend during the six–year period provided for in this <u>Section 5.8(c)</u>, that provides coverage no materially less favorable than the coverage described above.

- (d) The provisions of this <u>Section 5.8</u> are (i) intended to be for the benefit of, and shall be enforceable by, each Indemnitee, his or her heirs and his or her representatives and (ii) in addition to, and not in substitution for or limitation of, any other rights to indemnification or contribution that any such Person may have by Contract or otherwise. The obligations of Parent and the Surviving Corporation under this <u>Section 5.8</u> shall not be terminated or modified in such a manner as to adversely affect the rights of any Indemnitee to whom this <u>Section 5.8</u> applies unless (A) such termination or modification is required by applicable Law or (B) the affected Indemnitee shall have consented in writing to such termination or modification (it being expressly agreed that the Indemnitees to whom this <u>Section 5.8</u> applies shall be third party beneficiaries of this <u>Section 5.8</u>).
- (e) In connection with any claim for indemnification pursuant to any policy, agreement or governing document contemplated by this <u>Section 5.8</u>, Parent shall, and shall cause the Surviving Corporation to, provide such information and updates regarding such claim (including regarding the status of, and material developments arising in connection with, the defense thereof) as reasonably requested from time to time by Indemnitees party to such claim.
- (f) In the event that Parent, the Surviving Corporation or any of their respective successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers or conveys all or substantially all of its properties and assets to any Person, then, and in each such case, proper provision shall be made so that the successors and assigns of Parent and the Surviving Corporation shall assume all of the obligations thereof set forth in this <u>Section 5.8</u>.
- Section 5.9 <u>Transaction Litigation</u>. Each party hereto shall keep the other parties hereto reasonably informed of, and cooperate with the other parties hereto in connection with, any litigation or claim against any party hereto and/or its directors, officers or employees relating to the Transactions; <u>provided</u>, <u>however</u>, that the foregoing shall not require any party hereto to take any action if it may result in a waiver of any attorney/client or any other similar privilege. The Company shall cooperate with Parent in connection with, and shall give Parent the opportunity to participate in the defense or settlement of, any such litigation, and no settlement in connection with any such litigation shall be agreed to prior to the Effective Time without Parent's prior written consent (which consent shall not be unreasonably withheld, delayed or conditioned); provided, however, that no settlement requiring a payment or an admission of wrongdoing by a director shall be agreed without such director's consent.

Section 5.10 Section 16. The Company and Parent shall take all steps reasonably necessary to cause the Transactions, including any dispositions of equity securities of

the Company (including derivative securities with respect to such equity securities of the Company) by each individual who is or will be subject to the reporting requirements of Section 16(a) of the Exchange Act with respect to the Company or Parent, to be exempt under Rule 16b-3 promulgated under the Exchange Act.

# Section 5.11 Employee Matters.

- (a) For a period beginning at the Effective Time and ending on December 31, 2013 (the "Continuation Period"), Parent shall provide, or shall cause to be provided, to each individual who, immediately prior to the Effective Time, is an employee of the Company and its Subsidiaries and who, during the Continuation Period, continues such employment with the Surviving Corporation and its Subsidiaries ("Company Employees"), annual base salary and base wages and employee benefits (but excluding equity incentive compensation), in each case, that are no less favorable in the aggregate than such annual compensation and employee benefits provided to the Company Employees immediately prior to the Effective Time. In addition, (i) Parent shall or shall cause the Surviving Corporation to provide Company Employees whose employment terminates during the one (1) year period following the Effective Time with severance benefits at levels no less than and pursuant to the terms of the Company's severance policies as set forth on Section 5.11(a) of the Company Disclosure Schedule, but in no event more than ten (10) weeks' severance shall be paid and (ii) such severance benefits shall be determined taking into account the service crediting provisions set forth in Section 5.11(b).
- (b) For all purposes (including purposes of vesting, eligibility to participate and level of severance and other benefits based upon credited service, but not for purposes of defined benefit pension accrual) under the employee benefit plans of Parent and its Subsidiaries providing benefits to any Company Employee after the Effective Time (including any Company Plans that are continued) (the "New Plans"), each Company Employee shall be credited with his or her years of service with the Company and its Subsidiaries and their respective predecessors before the Effective Time, to the same extent as such service was credited to him or her under the terms of a corresponding Company Plan before the Effective Time, in which such Company Employee participated or was eligible to participate immediately prior to the Effective Time; provided that the foregoing shall not apply to the extent that its application would result in a duplication of benefits with respect to the same period of service. In addition, and without limiting the generality of the foregoing, (i) each Company Employee shall be immediately eligible to participate, without any waiting time, in any and all New Plans to the extent coverage under such New Plan is replacing comparable coverage under a Company Plan in which such Company Employee participated immediately before the Effective Time (such plans, collectively, the "Old Plans"), and (ii) for purposes of each New Plan providing medical, dental, pharmaceutical and/or vision benefits to any Company Employee, Parent shall use reasonable best efforts to cause any pre-existing condition exclusions and actively-at-work requirements of such New Plan to be waived for such Company Employee and his or her covered dependents, to the extent such conditions were inapplicable or waived under the comparable Old Plan of the Company or its Subsidiaries in which such Company Employee participated immediately prior to the Effective Time.

- (c) From and after the Effective Time, Parent shall cause the Surviving Corporation and its Subsidiaries to honor all obligations under the Company Plans and compensation and severance arrangements and agreements in accordance with their terms as in effect immediately before the Effective Time, other than the Company's 2013 Annual Incentive Compensation Plan for Management attached hereto as part of Section 5.11 of the Company Disclosure Schedule (the "2013 MIP") to the extent such Company Plans, arrangements and agreements are set forth on Section 5.11 of the Company Disclosure Schedule. In addition, upon the earlier of (i) a Company Employee's employment being terminated by the Company or a Subsidiary of the Company other than for "cause" or (ii) March 15, 2014, Parent shall cause the Surviving Corporation and its Subsidiaries to pay to each such Company Employee (other than a Company Employee listed on Section 8.13(b) of the Company Disclosure Schedule) a pro-rata portion of any bonus in respect of the 2013 performance year that such Company Employee would have been entitled to receive under the 2013 MIP accrued for the period from January 1, 2013 through the Closing Date. For the avoidance of doubt, neither any Company Employee listed on Section 8.13(b) of the Company Disclosure Schedule nor any Company Employee who voluntarily leaves the employ of the Surviving Corporation or any of its Subsidiaries, or whose employment is terminated for "cause", in each case prior to March 15, 2014, shall be entitled to any portion of the bonus accrued under the 2013 MIP. The parties hereto agree that, for purposes of this Section 5.11, the Transactions shall be deemed to constitute a "change in control," "change of control" or "corporate transaction" under such Company Plans, arrangements or agreements.
- (d) Nothing contained in this Section 5.11 shall (i) be treated as the adoption of or an amendment to any employee benefit or compensation plan, program, agreement, policy or arrangement, (ii) subject to compliance with the other provisions of this Section 5.11, alter or limit the ability of Parent, the Surviving Corporation or any of its or their Affiliates to amend or terminate any employee benefit or compensation plan, program, policy, agreement or arrangement (including any Company Plan and any New Plan) at any time sponsored, maintained or contributed to by any of them, (iii) confer upon any Person (including any current or former employee of the Company or any of its Subsidiaries and any Company Employee) any right to employment or continued employment for any period of time, or any right to a particular term or condition of employment, or shall in any way limit the ability of Parent, the Surviving Corporation or any of its or their affiliates to terminate the employment of any employee (including any Company Employee) at any time and for any or no reason, or (iv) confer upon any Person (including any current or former Company employee, any Company Employee and any of their respective dependents and beneficiaries) any rights as a third-party beneficiary of this Agreement. Parent's current intention is to transition Company Employees to the New Plans effective January 1, 2014.
- Section 5.12 <u>Merger Sub and Surviving Corporation</u>. Parent shall take all actions necessary to (a) cause Merger Sub and the Surviving Corporation to perform promptly their respective obligations under this Agreement and (b) cause Merger Sub to consummate the Merger on the terms and conditions set forth in this Agreement.

Section 5.13 No <u>Control of Other Party's Business</u>. Nothing contained in this Agreement is intended to give Parent or Merger Sub, directly or indirectly, the right to control or direct the Company's or its Subsidiaries' operations prior to the Effective Time. Prior to the

Effective Time, the Company shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its and its Subsidiaries' respective operations.

**Section 5.14** <u>Takeover Laws</u>. The Company and the Company Board shall each (a) use its reasonable best efforts to ensure that no state takeover statute or similar statute or regulation is or becomes applicable to the Transactions and (b) if any state takeover statute or similar statute becomes applicable to the Transactions, use its reasonable best efforts to ensure that such Transactions may be consummated as promptly as practicable on the terms contemplated by this Agreement and otherwise to minimize the effect of such statute or regulation on the Transactions.

Section 5.15 Reservation of Shares; Stock Exchange Listings. Parent shall (a) cause the shares of Parent Common stock to be issued in the Merger to be reserved for issuance prior to the Closing Date and (b) use its reasonable best efforts to cause the shares of Parent Common stock to be issued in the Merger to be approved for listing on Nasdaq, subject to official notice of issuance, prior to the Closing Date. Prior to the Closing Date, the Company shall use its reasonable best efforts to cause the shares of Company Common Stock to be delisted from Nasdaq and deregistered under the Exchange Act as soon as practicable following the Effective Time.

#### ARTICLE VI

# **CONDITIONS PRECEDENT**

**Section 6.1** Conditions to Each Party's Obligation to Effect the Merger. The respective obligations of each party hereto to effect the Merger shall be subject to the satisfaction (or waiver by Parent and the Company, if permissible under applicable Law) on or prior to the Closing Date of the following conditions:

- (a) Company Stockholder Approval . The Company Stockholder Approval shall have been obtained.
- (b) <u>Regulatory Approvals</u>. All waiting periods (and any extensions thereof) applicable to the Merger under the HSR Act shall have been terminated or shall have expired.
- (c) <u>No Injunctions or Restraints</u>. No Law, injunction, judgment or ruling enacted, promulgated, issued, entered, amended or enforced by any Governmental Authority (collectively, "Restraints") shall be in effect enjoining, restraining, preventing or prohibiting consummation of the Merger or making the consummation of the Merger illegal.
- (d) <u>Nasdaq Listing</u>. The shares of Parent Common Stock to be issued in the Merger and such other shares of Parent Common Stock to be reserved for issuance in connection with the Merger shall have been approved for listing on Nasdaq, subject to official notice of issuance.
- (e)  $\underline{\text{Form S-4}}$ . The Form S-4 shall have been declared effective by the SEC under the Securities Act and no stop order suspending the effectiveness of the Form S-4

shall have been issued by the SEC and remain in effect and no proceedings for that purpose shall have been initiated or threatened by the SEC.

Section 6.2 <u>Conditions to Obligations of Parent and Merger Sub</u>. The obligations of Parent and Merger Sub to effect the Merger are further subject to the satisfaction (or waiver by Parent, if permissible under applicable Law) on or prior to the Closing Date of the following conditions:

- (a) Representations and Warranties. The representations and warranties of the Company set forth in this Agreement, disregarding all qualifications and exceptions contained therein relating to materiality or Company Material Adverse Effect (other than the reference to "Company Material Adverse Effect" in Section 3.6(b) hereof), shall be true and correct as of the date of this Agreement and as of the Effective Time with the same effect as though made on and as of the Effective Time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date), except where the failure to be true and correct would not have a Company Material Adverse Effect; provided, however, that, notwithstanding the foregoing, (i) each of the representations and warranties of the Company set forth in Section 3.2 and Section 3.6(b)(i) shall be true and correct (other than de minimis inaccuracies) in all respects as of the date of this Agreement and as of the Effective Time as though made at and as of the Effective Time and (ii) each of the representations and warranties of the Company set forth in Section 3.3(a), Section 3.3(b)(i), Section 3.13 and Section 3.18 shall be true and correct in all material respects as of the date of this Agreement and as of the Effective Time as though made at and as of the Effective Time. Parent shall have received a certificate signed on behalf of the Company by an executive officer of the Company to such effect.
- (b) <u>Performance of Obligations of the Company</u>. The Company shall have performed in all material respects all obligations required to be performed by it under this Agreement at or prior to the Closing Date, and Parent shall have received a certificate signed on behalf of the Company by an executive officer of the Company to such effect.
- (c) <u>Absence of Company Material Adverse Effect</u>. Since the date of this Agreement, there shall not have occurred and be continuing a Company Material Adverse Effect or any fact, change, event, circumstance, occurrence or effect that would, individually or in the aggregate, reasonably be expected to result in a Company Material Adverse Effect.

**Section 6.3** <u>Conditions to Obligations of the Company</u>. The obligation of the Company to effect the Merger is further subject to the satisfaction (or waiver by the Company, if permissible under applicable Law) on or prior to the Closing Date of the following conditions:

(a) <u>Representations and Warranties</u>. The representations and warranties of Parent and Merger Sub set forth in this Agreement, disregarding all qualifications and exceptions contained therein relating to materiality or Parent Material Adverse Effect (other than the reference to "Parent Material Adverse Effect" in <u>Section 4.6(b)</u>

hereof), shall be true and correct as of the date of this Agreement and as of the Effective Time with the same effect as though made on and as of the Effective Time (except to the extent that any such representation and warranty expressly speaks as of an earlier date, in which case such representation and warranty shall be true and correct as of such earlier date) except where such failures to be so true and correct would not have a Parent Material Adverse Effect; provided, however, that, notwithstanding the foregoing, (i) each of the representations and warranties of Parent and Merger Sub set forth in Section 4.2, and Section 4.6(b) shall be true and correct (other than de minimis inaccuracies) in all respects as of the date of this Agreement and as of the Effective Time as though made at and as of the Effective Time and correct in all material respects as of the date of this Agreement and as of the Effective Time as though made at and as of the Effective Time. The Company shall have received a certificate signed on behalf of Parent and Merger Sub by an executive officer of Parent and Merger Sub to such effect.

- (b) <u>Performance of Obligations of Parent and Merger Sub</u>. Parent and Merger Sub shall have performed in all material respects all obligations required to be performed by them under this Agreement at or prior to the Closing Date, and the Company shall have received a certificate signed on behalf of Parent by an executive officer of Parent to such effect.
- (c) <u>Absence of Parent Material Adverse Effect</u>. Since the date of this Agreement, there shall not have occurred and be continuing a Parent Material Adverse Effect or any fact, change, event, circumstance, occurrence or effect that would, individually or in the aggregate, reasonably be expected to result in a Parent Material Adverse Effect.

**Section 6.4** <u>Frustration of Closing Conditions</u>. None of the Company, Parent or Merger Sub may rely on the failure of any condition set forth in <u>Section 6.1</u>, <u>Section 6.2</u> or <u>Section 6.3</u>, as the case may be, to be satisfied if such failure was caused by such party's failure to use its reasonable best efforts to consummate the Merger and the other Transactions or due to the failure of such party to perform in all material respects any of its other obligations under this Agreement.

#### ARTICLE VII

## **TERMINATION**

**Section 7.1** <u>Termination</u>. This Agreement may be terminated and the Transactions abandoned at any time prior to the Effective Time, whether before or after receipt of the Company Stockholder Approval:

- (a) by the mutual written consent of the Company and Parent; or
- (b) by either of the Company or Parent:
- (i) if the Merger shall not have been consummated on or before August 7, 2013 (as such date is extended pursuant to the following proviso, as applicable, the "End Date"); provided, however, that if the Closing has not occurred by such date and

on such date the conditions set forth in Section 6.1(b) or Section 6.1(c) with respect to the matters contemplated by Section 6.1(b) have not been satisfied or waived and each of the other conditions to consummation of the Merger set forth in Article VI has been satisfied, waived or remains capable of satisfaction, then the End Date shall automatically be extended to October 7, 2013; and provided, further, that the right to terminate this Agreement pursuant to this Section 7.1(b)(i) shall not be available to a party if the failure of the Merger to have been consummated on or before the End Date was primarily due to the failure of such party to timely perform any of its obligations under this Agreement;

- (ii) if any Restraint having the effect set forth in <u>Section 6.1(c)</u> shall be in effect and shall have become final and non-appealable; <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this <u>Section 7.1(b)(ii)</u> shall not be available to a party if the issuance of such final, non-appealable Restraint was primarily due to the failure of such party to timely perform any of its obligations under this Agreement or if such party shall have failed to comply with its obligations under <u>Section 5.3(a)</u> to seek to cause such Restraint to be vacated or lifted or to ameliorate the effects thereof; or
- (iii) if the Company Stockholders Meeting (including any adjournments or postponements thereof) shall have concluded and the Company Stockholder Approval contemplated by this Agreement shall not have been obtained; or
  - (c) by Parent:
- (i) if the Company shall have breached or failed to perform any of its representations, warranties, covenants or agreements set forth in this Agreement, which breach or failure to perform (A) would give rise to the failure of a condition set forth in Section 6.2(a) or Section 6.2(b), respectively, and (B) cannot be cured by the Company by the End Date or, if capable of being cured, shall not have been cured within thirty (30) calendar days following receipt of written notice from the Parent stating the Parent's intention to terminate this Agreement pursuant to this Section 7.1(c)(i) and the basis for such termination; provided that Parent shall not have the right to terminate this Agreement pursuant to this Section 7.1(c)(i) if it is then in material breach of any representation, warranties, covenants or other agreements hereunder; or
- (ii) (A) the Company Board (or the Review Committee or other duly authorized committee of the Company Board) shall have effected a Company Adverse Recommendation Change, (B) the Company, upon the written request of Parent, fails to publicly restate its recommendation of the Merger within five (5) Business Days after the date any Takeover Proposal is first commenced, published or delivered to the Company's stockholders or, in the case of the commencement, publication or delivery of any Takeover Proposal pursuant to Regulation 14D under the Exchange Act, prior to the later of (1) the fifth (5th) Business Day following any written request by Parent or (2) ten (10) Business Days following such commencement, publication or delivery or (C) the Company fails to include the Company Board Recommendation in the Proxy Statement/Prospectus; or

- (iii) the Company shall have committed a Willful Breach of any of its obligations under Section 5.3(a) or Section 5.3(b); or
- (iv) the Company shall have (A) granted a written waiver or release of, or (B) determined (through an act authorized by the Company Board (or the Review Committee or other duly authorized committee thereof)) not to enforce the terms of, in each case, any standstill agreement in effect between the Company and any Person listed on Section 7.1(c)(iv) of the Company Disclosure Schedule; or
  - (d) by the Company:
- (i) if Parent or Merger Sub shall have breached or failed to perform any of its representations, warranties, covenants or agreements set forth in this Agreement, which breach or failure to perform (A) would give rise to the failure of a condition set forth in Section 6.3(a) or Section 6.3(b), respectively, and (B) cannot be cured by Parent or Merger Sub by the End Date or, if capable of being cured, shall not have been cured within thirty (30) calendar days following receipt of written notice from the Company stating the Company's intention to terminate this Agreement pursuant to this Section 7.1(d)(i) and the basis for such termination; provided that, the Company shall not have the right to terminate this Agreement pursuant to this Section 7.1(d)(i) if it is then in material breach of any representation, warranties, covenants or other agreements hereunder; or
- (ii) in order to enter into a transaction that is a Superior Proposal, in accordance with Section 5.3; provided that the right to terminate this Agreement pursuant to this Section 7.1(d)(ii) shall not be available to the Company unless the Company pays or has paid the Termination Fee to Parent or caused the Termination Fee to be paid to Parent in accordance with Section 7.3; it being understood that the Company may enter into any transaction that is a Superior Proposal simultaneously with the termination of this Agreement pursuant to this Section 7.1(d)(ii).

Section 7.2 Effect of Termination . In the event of the termination of this Agreement as provided in Section 7.1, written notice thereof shall be given to the other parties, specifying the provision hereof pursuant to which such termination is made, and this Agreement shall forthwith become null and void (other than Section 7.2, Section 7.3 and Article VIII, all of which shall survive termination of this Agreement), and there shall be no liability on the part of Parent, Merger Sub or the Company or their respective directors, officers and Affiliates hereunder; provided, however, that, subject to Section 7.3 (including the limitations on liability contained therein), no party shall be relieved or released from any liabilities or damages arising out of (A) any breach of its obligations under this Agreement, (B) its fraud or willful misconduct or (C) willful or material breach of its representations or warranties set forth in this Agreement.

#### **Section 7.3 Termination Fee.**

(a) In the event that this Agreement is terminated by the Company pursuant to <u>Section 7.1(d)(ii)</u>, prior to or substantially simultaneously with (and in no event later

than the day of such termination) such termination the Company shall pay or cause to be paid as directed by Parent the Termination Fee.

- (b) In the event that this Agreement is terminated by Parent pursuant to Section 7.1(c)(ii), Section 7.1(c)(iii) or Section 7.1(c) (iv), the Company shall pay or cause to be paid as directed by Parent the Termination Fee within two (2) Business Days of such termination.
- (c) In the event that this Agreement is terminated (i) by Parent or the Company pursuant to Section 7.1(b)(i) or Section 7.1(b) (iii), (ii) a bona fide Takeover Proposal shall have been publicly disclosed after the date hereof and prior to the End Date or date of the Company Stockholder Meeting, as applicable, and (iii) within twelve (12) months of the date this Agreement is terminated, the Company consummates the transactions contemplated by such Takeover Proposal (provided that for purposes of clause (iii) of this Section 7.3(c), the references to "20%" in the definition of Takeover Proposal shall be deemed to be references to "50%"), then, the Company shall pay or cause to be paid as directed by Parent the Termination Fee on the date of consummation of such transaction.
- (d) For purposes of this Agreement, "<u>Termination Fee</u>" shall mean an amount equal to \$7.4 million; <u>provided</u> that, in the event that this Agreement is terminated by the Company pursuant to <u>Section 7.1(d)(ii)</u> on or before the Excluded Party End Date (applicable to such Excluded Party) in connection with entering into an agreement relating to a transaction that is a Superior Proposal with an Excluded Party, "Termination Fee" shall mean an aggregate amount equal to (i) \$4.25 million.
- (e) In the event this Agreement is terminated by the Company pursuant to Section 7.1(d)(ii) and a Termination Fee is due to Parent in accordance with the terms hereof, then the Company shall, in addition to the payment of the Termination Fee, reimburse Parent upon demand up to an aggregate amount (not to exceed \$2,000,000) equal to all documented out-of-pocket fees and expenses reasonably incurred or paid by or on behalf of Parent on or after November 29, 2012 in connection with the Transactions or related to the authorization, preparation, negotiation, execution and performance of this Agreement and the Voting Agreement, in each case including all such fees and expenses of counsel, financial advisors, accountants, experts and consultants retained by Parent and any commitment fees paid by Parent in connection with the financing for the Transactions.
- (f) Notwithstanding the foregoing, in no event shall the Company be required to pay the fees referred to in this Section 7.3 (i) on more than one occasion or (ii) if, at the time this Agreement is terminated by Parent, this Agreement could have been terminated by the Company pursuant to Section 7.1(b)(ii), Section 7.1(b)(iii) (unless on or before the date of the Company Stockholders Meeting a bona fide Takeover Proposal shall have been publicly disclosed, in which event Section 7.3(c) shall govern whether a fee shall be payable) or Section 7.1 (d)(i). Notwithstanding anything to the contrary in this Agreement, the parties agree that the payment of the Termination Fee and, if applicable in accordance with Section 7.3(e), Parent expenses, shall be the sole and exclusive remedy available to Parent and Merger Sub with respect to this Agreement and the Transactions in the event any such payment becomes due and payable, and, upon payment of the Termination Fee and, if applicable in accordance with Section 7.3(e),

Parent expenses, the Company (and the Company's Affiliates and its and their respective directors, officers, employees, stockholders and Representatives) shall have no further liability to Parent and Merger Sub under this Agreement.

- (g) Any amount that becomes payable pursuant to this  $\underline{\text{Section 7.3(a)}}$  shall be paid by wire transfer of immediately available funds to the account designated in  $\underline{\text{Section 7.3(g)}}$  of the Parent Disclosure Schedule and shall be reduced by any amounts required to be deducted or withheld therefrom under applicable Law in respect of Taxes.
- (h) The Company acknowledges and hereby agrees that the provisions of this <u>Section 7.3(a)</u> are an integral part of the transactions contemplated by this Agreement (including the Merger), and that, without such provisions, Parent and Merger Sub would not have entered into this Agreement. If the Company shall fail to pay in a timely manner the amounts due pursuant to this <u>Section 7.3(a)</u>, and, in order to obtain such payment, Parent makes a claim against the Company that results in a judgment against the Company, the Company shall pay to Parent the reasonable costs and expenses of Parent (including its reasonable attorneys' fees and expenses) incurred or accrued in connection with such suit.

## ARTICLE VIII

## MISCELLANEOUS

- **Section 8.1** No Survival of Representations and Warranties. None of the representations and warranties in this Agreement or in any instrument delivered pursuant to this Agreement shall survive the Effective Time. This Section 8.1 shall not limit any covenant or agreement of the parties that by its terms contemplates performance in whole or in part after the Effective Time. The Confidentiality Agreement shall (a) survive termination of this Agreement in accordance with its terms and (b) terminate as of the Effective Time.
- **Section 8.2** <u>Fees and Expenses</u>. Except as provided in <u>Section 7.3</u>, whether or not the Merger is consummated, all fees and expenses incurred in connection with the Merger, this Agreement and the Transactions shall be paid by the party incurring or required to incur such fees or expenses.
- **Section 8.3** <u>Amendment or Supplement</u>. At any time prior to the Effective Time, this Agreement may be amended or supplemented in any and all respects, whether before or after receipt of the Company Stockholder Approval, by written agreement of the parties hereto and delivered by duly authorized officers of the respective parties; <u>provided</u>, <u>however</u>, that following approval of the Transactions by the stockholders of the Company, there shall be no amendment or change to the provisions hereof which by Law would require further approval of this Agreement or the Transactions by the stockholders of the Company.
- **Section 8.4** Waiver. At any time prior to the Effective Time, any party may, subject to applicable Law, (a) waive any inaccuracies in the representations and warranties of any other party hereto, (b) extend the time for the performance of any of the obligations or acts of any other party hereto or (c) waive compliance by the other party with any of the agreements contained herein or, except as otherwise provided herein, waive any of such party's conditions.

Notwithstanding the foregoing, no failure or delay by the Company, Parent or Merger Sub in exercising any right hereunder shall operate as a waiver thereof nor shall any single or partial exercise thereof preclude any other or further exercise thereof or the exercise of any other right hereunder. Any agreement on the part of a party hereto to any such extension or waiver shall be valid only if set forth in an instrument in writing signed on behalf of such party.

Section 8.5 <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned, directly or indirectly, in whole or in part, by operation of Law or otherwise, by any of the parties without the prior written consent of the other parties; provided that Merger Sub, upon prior written notice to the Company, may assign (in its sole discretion) any of or all its rights, interests and obligations under this Agreement to Parent or to any direct or indirect wholly-owned Subsidiary of Parent, but no such assignment shall relieve Parent or Merger Sub of any of its obligations hereunder. Subject to the preceding sentence, this Agreement shall be binding upon, inure to the benefit of, and be enforceable by, the parties hereto and their respective successors and permitted assigns. Any purported assignment not permitted under this Section shall be null and void.

**Section 8.6** <u>Counterparts</u>. This Agreement may be executed in counterparts (each of which shall be deemed to be an original but all of which taken together shall constitute one and the same agreement) and shall become effective when one or more counterparts have been signed by each of the parties and delivered (by electronic communication, facsimile or otherwise) to the other parties.

Section 8.7 Entire Agreement; Third-Party Beneficiaries. This Agreement, including the Company Disclosure Schedule and Parent Disclosure Schedule, and the exhibits hereto, together with the other instruments referred to herein, including the Voting Agreement and the Confidentiality Agreement (a) constitute the entire agreement, and supersede all other prior agreements and understandings, both written and oral, among the parties, or any of them, with respect to the subject matter hereof and thereof and (b) except for (i) the rights of the Company's stockholders, holders of Options and Company Stock Units to receive the Merger Consideration and consideration in respect of Options and Company Stock Units, respectively, at the Effective Time, and (ii) the rights of any third party beneficiaries under Section 5.8, is not intended to and shall not confer upon any Person other than the parties hereto any rights or remedies hereunder.

## Section 8.8 Governing Law; Jurisdiction.

- (a) This Agreement and all actions and proceedings (whether based on contract, tort or otherwise) arising out of or relating to this Agreement, any of the Transactions or the actions of Parent, Merger Sub or the Company in the negotiation, administration, performance and enforcement hereof or thereof shall be governed by, and construed in accordance with, the Laws of the State of Delaware, without giving effect to any choice or conflict of Laws provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware.
- (b) Each of the parties hereto hereby agrees that (i) all actions and proceedings arising out of or relating to this Agreement, any of the Transactions or the actions of

Parent, Merger Sub or the Company in the negotiation, administration, performance and enforcement hereof or thereof shall be heard and determined in the Chancery Court of the State of Delaware and any state appellate court therefrom sitting in New Castle County in the State of Delaware (or, if such Chancery Court of the State of Delaware declines to accept jurisdiction over a particular matter, any state or federal court within the State of Delaware and any appellate court therefrom), (ii) it will not attempt to deny or defeat such personal jurisdiction by motion or other request for leave from any such court, (iii) agrees that it will not bring any action relating to this Agreement or the Transactions (including the Merger) in any court other than the aforesaid courts and (iv) a final judgment in any action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law. Each of the parties hereto hereby waives any objection that it may now or hereafter have to the venue of any such action or proceeding in any such court or that such action or proceeding was brought in an inconvenient court and agrees not to plead or claim the same.

(c) Each party irrevocably consents to the service of process outside the territorial jurisdiction of the courts referred to in this <u>Section 8.8</u> in any such action or proceeding by mailing copies thereof by registered or certified United States mail, postage prepaid, return receipt requested, to its address as specified in or pursuant to this <u>Article VIII</u>. However, the foregoing shall not limit the right of a party to effect service of process on the other party by any other legally available method.

Section 8.9 Specific Enforcement. The parties agree that immediate, extensive and irreparable damage would occur for which monetary damages would not be an adequate remedy in the event that any of the provisions of this Agreement are not performed in accordance with their specific terms or are otherwise breached. Accordingly, the parties agree that, if for any reason Parent, Merger Sub or the Company shall have failed to perform its obligations under this Agreement or otherwise breached this Agreement, then the party seeking to enforce this Agreement against such nonperforming party under this Agreement shall be entitled to specific performance and the issuance of immediate injunctive and other equitable relief without the necessity of proving the inadequacy of money damages as a remedy, and the parties further agree to waive any requirement for the securing or posting of any bond in connection with the obtaining of any such injunctive or other equitable relief, this being in addition to and not in limitation of any other remedy to which they are entitled at Law or in equity.

Section 8.10 WAIVER OF JURY TRIAL. EACH OF THE PARTIES HERETO IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING BETWEEN THE PARTIES HERETO ARISING OUT OF OR RELATING TO THIS AGREEMENT, ANY OF THE TRANSACTIONS OR THE ACTIONS OF PARENT, MERGER SUB OR THE COMPANY IN THE NEGOTIATION, ADMINISTRATION, PERFORMANCE OR ENFORCEMENT HEREOF OR THEREOF.

**Section 8.11 Notices**. All notices, requests and other communications to any party hereunder shall be in writing and shall be deemed given if delivered personally, facsimiled (which is confirmed) or sent by overnight courier (providing proof of delivery) to the parties at the following addresses:

If to Parent or Merger Sub, to:

Encore Capital Group, Inc. 3111 Camino Del Rio North

**Suite 1300** 

San Diego, California Attention: General Counsel Facsimile: 858-309-1546

with a copy (which shall not constitute notice) to:

Fulbright & Jaworski L.L.P.

300 Convent Street

**Suite 2100** 

San Antonio, Texas 78205-3792 Attention: Daryl Lansdale Facsimile: (210) 270-7205

If to the Company, to:

Asset Acceptance Capital Corp. 28405 Van Dyke Avenue Warren, Michigan 48093 Attention: Rion Needs

Reid E. Simpson Edwin L. Herbert Facsimile: (586) 983-7081

(586) 446-7832 (586) 446-1783

with a copy (which shall not constitute notice) to:

Kirkland & Ellis LLP 601 Lexington Avenue New York, NY 10022

Attention: Michael Movsovich

Jeffrey Symons

Facsimile: (212) 446-6460

or such other address or facsimile number as such party may hereafter specify by like notice to the other parties hereto. All such notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5 p.m. in the place of receipt and such day is a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed not to have been received until the next succeeding Business Day in the place of receipt.

**Section 8.12** Severability. If any term or other provision of this Agreement is determined by a court of competent jurisdiction to be invalid, illegal or incapable of being enforced by any rule of Law or public policy, all other terms, provisions and conditions of this Agreement shall nevertheless remain in full force and effect. Upon such determination that any

term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible to the fullest extent permitted by applicable Law in an acceptable manner to the end that the Transactions are fulfilled to the extent possible.

Section 8.13 <u>Definitions</u>. As used in this Agreement, the following terms shall have the meanings ascribed to them below:

- "2013 Company Plan" means the Company's long term business plan made available to Parent prior to the date hereof and which is attached hereto as Section 5.1(a) of the Company Disclosure Schedule.
  - "2013 MIP" shall have the meaning set forth in Section 5.11(c).
- "<u>Acceptable Confidentiality Agreement</u>" shall mean a customary confidentiality agreement (which need not prohibit the making of a Takeover Proposal) that contains provisions that are no less favorable in the aggregate to the Company than those contained in the Confidentiality Agreement (other than that it need not prohibit the making of a Takeover Proposal), with any changes thereto as reasonably necessary to allow the Company to comply with its obligations under <u>Section 5.3</u>.
- "Affiliate" shall mean, as to any Person, any other Person that, directly or indirectly, controls, or is controlled by, or is under common control with, such Person. For this purpose, "control" (including, with its correlative meanings, "controlled by" and "under common control with") shall mean the possession, directly or indirectly, of the power to direct or cause the direction of management or policies of a Person, whether through the ownership of securities or partnership or other ownership interests, by Contract or otherwise.
  - "Agreement" shall have the meaning set forth in the preamble.
  - "Antitrust Laws" shall have the meaning set forth in Section 5.4(a).
  - "Balance Sheet Date" shall have the meaning set forth in Section 3.5(g).
  - "Bankruptcy and Equity Exception" shall have the meaning set forth in Section 3.3(a).
  - "Book-Entry Shares" shall have the meaning set forth in Section 2.1(e).
- "Business Day" shall mean a day except a Saturday, a Sunday or other day on which the SEC or banks in the City of New York are authorized or required by Law to be closed.
  - "Cash Consideration" shall have the meaning set forth in Section 2.1(c)(i).
  - "Cash Election" shall have the meaning set forth in Section 2.1(c)(i).
  - "Certificate" shall have the meaning set forth in Section 2.1(e).
  - "Certificate of Merger" shall have the meaning set forth in Section 1.3.

- "Claim" shall have the meaning set forth in Section 5.8(a).
- "Clayton Act" shall mean the Clayton Act of 1914.
- "Closing" shall have the meaning set forth in Section 1.2.
- "Closing Date" shall have the meaning set forth in Section 1.2.
- "Code" shall have the meaning set forth in Section 2.7.
- "Company" shall have the meaning set forth in the preamble.
- "Company Adverse Recommendation Change" shall have the meaning set forth in Section 5.3(e).
- "Company Board" shall mean the board of directors of the Company.
- "Company Board Recommendation" shall have the meaning set forth in Section 5.2(c).
- "Company Charter Documents" shall have the meaning set forth in Section 3.1(c).
- "Company Common Stock" shall have the meaning set forth in Section 2.1.
- "Company Disclosure Schedule" shall have the meaning set forth in the Article III preamble.
- "Company DSUs" shall have the meaning set forth in Section 2.4(b).
- "Company Employees" shall have the meaning set forth in Section 5.11(a).
- "Company Insurance Policies" shall have the meaning set forth in Section 3.21.
- "Company IP" shall have the meaning set forth in Section 3.12.
- "Company IP Agreements" shall have the meaning set forth in Section 3.12.

<sup>&</sup>quot;Company Material Adverse Effect" shall mean any fact, change, event, circumstance, occurrence or effect that, individually or in the aggregate, (a) is materially adverse to the business, results of operations or financial condition of the Company and its Subsidiaries taken as a whole, other than any fact, change, event, circumstance, occurrence or effect, directly or indirectly, arising out of, resulting from or relating to the following: (i) any condition, change, event, occurrence or effect in any of the industries or markets in which the Company or its Subsidiaries operates; (ii) any enactment of, change in, or change in interpretation or enforcement of, any Law or GAAP or government policy; (iii) general economic, regulatory or political conditions (or changes therein) or conditions (or changes therein) in the financial, credit or securities markets (including changes in interest or currency exchange rates) in any country or region in which the Company, any of its Subsidiaries or any of their respective business relations conducts business; (iv) any acts of God, natural disasters, terrorism, armed hostilities, sabotage, war or any escalation or worsening of acts of terrorism, armed hostilities or war; (v) the

announcement, pendency of or performance of the Transactions, including by reason of the identity of Parent or any communication by Parent regarding the plans or intentions of Parent with respect to the conduct of the business of the Company or any of its Subsidiaries and including the impact of any of the foregoing on any relationships, contractual or otherwise, with customers, suppliers, distributors, collaboration partners, stockholders, employees or regulators; (vi) any action taken pursuant to the terms of this Agreement or with the express written consent or at the express written direction of Parent or Merger Sub (or any action not taken as a result of the failure of Parent to consent to any action requiring Parent's consent pursuant to Section 5.1, including changes resulting from any Remedial Action, if required); (vii) any change in the market price, or change in trading volume, of the capital stock of the Company, (viii) any failure by the Company or its Subsidiaries to meet internal, analysts' or other earnings estimates or financial projections or forecasts for any period, or any changes in credit ratings and any changes in any analysts recommendations or ratings with respect to the Company or any of its Subsidiaries; and (ix) any pending, initiated or threatened legal or administrative proceeding, claim, suit or action against the Company, any of its Subsidiaries or any of their respective officers or directors, in each case, arising out of or relating to the execution of this Agreement or the Transactions; provided that (1) any fact, change, event, circumstance, occurrence or effect set forth in clauses (a)(i) through (a)(iv) may be taken into account in determining whether there has been, is or would reasonably be expected to be a Company Material Adverse Effect to the extent (and only to the extent) such fact, change, event, circumstance, occurrence or effect has a disproportionate adverse effect on business, results of operations or financial condition of the Company and its Subsidiaries taken as a whole, in relation to other participants in the business and industries in which the Company and its Subsidiaries operate, and (2) the underlying cause of any failure referred to in clauses (a)(vii) and (a)(viii) may be taken into account in determining whether there has been, is or would reasonably be expected to be a Company Material Adverse Effect; or (b) would, individually or in the aggregate, reasonably be expected to prevent or materially impede, interfere with, hinder or delay the consummation by the Company of the Transactions.

- "Company Pension Plan" shall have the meaning set forth in Section 3.9(a).
- "Company Permits" shall have the meaning set forth in Section 3.8(b).
- "Company Plan" shall mean each "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) and each other employment (other than employment agreements terminable at will without any severance obligation in excess of the amount set forth on Section 5.11 of the Company Disclosure Schedule), incentive (equity or otherwise), severance, retention, change in control, deferred compensation, material fringe benefit, perquisite or other material compensatory plan, program, policy, agreement or arrangement that is maintained or contributed to by the Company or any of its Subsidiaries or with respect to which the Company or any of its Subsidiaries is obligated to contribute for the benefit of any current or former employee or director (or any of their respective beneficiaries or dependents) of the Company or any of its Subsidiaries or otherwise with respect to which the Company or any of its Subsidiaries has or would reasonably be expected to incur any material liability.

"Company Preferred Stock" shall have the meaning set forth in Section 3.2.

- "Company RSUs" shall have the meaning set forth in Section 2.4(b).
- "Company SEC Documents" shall have the meaning set forth in Section 3.5(a).
- "Company Securities" shall have the meaning set forth in Section 3.2.
- "Company Stock Plans" shall mean the Company's 2012 Stock Incentive Plan and the Company's 2004 Stock Incentive Plan.
- "Company Stock Units" shall have the meaning set forth in Section 2.4(b).
- "Company Stockholder Approval" shall have the meaning set forth in Section 3.18.
- "Company Stockholders Meeting" shall have the meaning set forth in Section 5.2(b).
- "Confidentiality Agreement" shall have the meaning set forth in Section 5.6(a).
- "Consent Decree" shall mean that certain Consent Decree dated on or about January 30, 2012 between the United States of America and Asset Acceptance, LLC.
- "Contract" means any loan or credit agreement, debenture, note, bond, mortgage, indenture, deed of trust, lease, license, contract, commitment, instrument, obligation, binding arrangement, understanding or other agreement, whether written or oral.
  - " DGCL" shall have the meaning set forth in the recitals.
  - "<u>Dissenting Shares</u>" shall have the meaning set forth in <u>Section 2.6</u>.
  - "<u>Dissenting Stockholders</u>" shall have the meaning set forth in <u>Section 2.6</u>.
  - " Effective Time" shall have the meaning set forth in Section 1.3.
  - "Election Deadline" shall have the meaning set forth in Section 2.2(b).
  - "Election Form" shall have the meaning set forth in Section 2.2(a).
- "Encumbrances" shall mean any mortgage, deed of trust, lease, license, restriction, hypothecation, option to purchase or lease or otherwise acquire any interest, right of first refusal or offer, conditional sales or other title retention agreement, adverse claim of ownership or use, easement, encroachment, right of way or other title defect, or encumbrance of any kind or nature.
  - "End Date" shall have the meaning set forth in Section 7.1(b)(i).
- "Environmental Laws" means all Laws relating to workplace safety or health, pollution or protection of natural resources, endangered or threatened species or the environment (including ambient air, soil, surface water or groundwater, or subsurface strata), including Laws relating to the presence of, the exposure to or the management, manufacture, use, containment, storage, recycling, reclamation, reuse, treatment, generation, discharge, transportation,

processing, production, disposal or remediation of, or Releases or threatened Releases of, Hazardous Materials as the foregoing are enacted or in effect on or prior to Closing.

- "ERISA" means the Employee Retirement Income Security Act of 1974.
- "Exchange Act" shall have the meaning set forth in Section 3.4.
- "Exchange Agent" shall have the meaning set forth in Section 2.3(a).
- "Excluded Party" shall mean any Person (or group of Persons that includes any such Person) listed on Section 1A of the Company Disclosure Schedule.
  - "Excluded Party End Date" has the meaning set forth in Section 5.3(c).
  - "Federal Trade Commission Act" shall mean the Federal Trade Commission Act of 1914.
  - "Form S-4" shall have the meaning set forth in Section 3.4.
  - "GAAP" shall mean generally accepted accounting principles in the United States, consistently applied.
  - "Go Shop End Date" shall have the meanings set forth in Section 5.3(c).
- "Governmental Authority" shall mean any federal, state or local, domestic, foreign or multinational government, court, regulatory or administrative agency, commission, authority or other governmental instrumentality.
- "<u>Hazardous Materials</u>" means any materials or substances or wastes as to which liability or standards of conduct may be imposed under any Environmental Law.
  - "HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976.
  - "IE Notice Period" shall have the meaning set forth in Section 5.3(f)(iii).
- "Indebtedness" shall mean (i) any indebtedness for borrowed money (including the issuance of any debt security), (ii) any capital lease obligations, (iii) any obligation as an account party in respect of letters of credit, (iv) any obligation under interest rate swap agreements, interest rate cap agreements and interest rate collar agreements, and all other agreements or arrangements designed to protect such Person against fluctuations in interest rates or currency exchange rates, and (v) any agreement, undertaking or arrangement by which a Person guarantees, endorses or otherwise becomes or is contingently liable upon (by direct or indirect agreement, contingent or otherwise, to provide funds for payment, to supply funds to, or otherwise to invest in, a debtor, or otherwise to assure a creditor against loss) the indebtedness, obligation or any other liability of any other Person (other than by endorsements of instruments in the course of collection and other than the Company or any Subsidiary).
  - "Indemnitee" and "Indemnitees" shall have the meanings set forth in Section 5.8(a).

- "Intellectual Property" shall mean, in any and all jurisdictions throughout the world, all (a) patents and patent applications, (b) registered trademarks, trade names, service marks, logos, corporate names, internet domain names, and any applications for registration of any of the foregoing, together with all goodwill associated with each of the foregoing, (c) registered and material unregistered copyrights, including copyrights in computer software, mask works and databases and (d) trade secrets and other proprietary know-how.
- "Intervening Event" means an event, development or change in circumstance that was not actually known to the Company Board (or the Review Committee or other duly authorized committee of the Company Board) prior to the execution of this Agreement (or if so known, the consequences of which, when taken together with all other events, developments or circumstances then known, were not actually known or reasonably foreseeable based on information actually known to the members of the Company Board (or the Review Committee or other duly authorized committee of the Company Board) as of the date of this Agreement), which event, development or change in circumstance, or any consequence thereof, becomes known to the Company Board (or the Review Committee or other duly authorized committee of the Company Board) prior to the receipt of the Company Stockholder Approval; provided, that in no event shall the receipt, existence or terms of a Takeover Proposal or any matter relating thereto or consequence thereof constitute an Intervening Event.
  - "IRS" means the U.S. Internal Revenue Service.
- "Knowledge" shall mean, (a) in the case of the Company, the actual knowledge, as of the date of this Agreement, of the individuals listed on Section 8.13 of the Company Disclosure Schedule and (b) in the case of Parent and Merger Sub, the actual knowledge, as of the date of this Agreement, of the individuals listed on Section 8.13 of the Parent Disclosure Schedule.
  - "Laws" shall have the meaning set forth in Section 3.8.
  - "Leased Real Property" shall have the meaning set forth in Section 3.14.
  - "Leases" shall have the meaning set forth in Section 3.14.
- "Liens" shall mean any pledges, liens, charges, encumbrances, options to purchase or lease or otherwise acquire any interest, and security interests of any kind or nature whatsoever.
  - "Mailing Date" shall have the meaning set forth in Section 2.2(a).
  - "Material Contract" shall have the meaning set forth in Section 3.15(a).
  - "Maximum Stock Election" shall have the meaning set forth in Section 2.1(c)(ii).
  - "Merger" shall have the meaning set forth in the recitals.
  - "Merger Consideration" shall have the meaning set forth in Section 2.1(c).
  - "Merger Sub" shall have the meaning set forth in the preamble.

- "Merger Sub Board" shall mean the board of directors of Merger Sub.
- "Nasdaq" shall mean The Nasdaq Stock Market.
- "New Plans" shall have the meaning set forth in Section 5.11(a).
- "Non-Electing Company Share" shall have the meaning set forth in Section 2.1(c)(iii).
- "Notice of Superior Proposal" shall have the meaning set forth in Section 5.3(f)(ii).
- "Notice Period" shall have the meaning set forth in Section 5.3(f)(iii).
- "Old Plans" shall have the meaning set forth in Section 5.11(a).
- "Option" shall have the meaning set forth in Section 2.4(a).
- "Optionholder" shall have the meaning set forth in Section 2.4(a).
- "Parent" shall have the meaning set forth in the preamble.
- "Parent Board" shall mean the board of directors of Parent.
- "Parent Charter Documents" shall have the meaning set forth in Section 4.1(b).
- "Parent Common Stock" means the common stock of Parent, par value \$0.01 per share.
- "Parent Disclosure Schedule" shall have the meaning set forth in the Article IV preamble.
- "Parent Material Adverse Effect" shall mean any fact, change, event, circumstance, occurrence or effect that, directly or indirectly, (a) is, or would reasonably be expected to become, materially adverse to the business, results of operations or financial condition of Parent and its Subsidiaries taken as a whole, other than any fact, change, event, circumstance, occurrence or effect, directly or indirectly, arising out of, resulting from or relating to the following: (i) any condition, change, event, occurrence or effect in any of the industries or markets in which Parent or its Subsidiaries operates; (ii) any enactment of, change in, or change in interpretation or enforcement of, any Law or GAAP or government policy; (iii) general economic, regulatory or political conditions (or changes therein) or conditions (or changes therein) in the financial, credit or securities markets (including changes in interest or currency exchange rates) in any country or region in which Parent, any of its Subsidiaries or their respective business relations conducts business; (iv) any acts of God, natural disasters, terrorism, armed hostilities, sabotage, war or any escalation or worsening of acts of terrorism, armed hostilities or war; (v) the announcement, pendency of or performance of the Transactions, including by reason of any communication by Parent regarding the plans or intentions of Parent with respect to the conduct of the business of the Company or any of its Subsidiaries and including the impact of any of the foregoing on any relationships, contractual or otherwise, with customers, suppliers, distributors, collaboration partners, stockholders, employees or regulators; (vi) any action taken pursuant to the terms of this Agreement, including any changes resulting from any Remedial Action, if required; (vii) any change in the market price, or change in trading

volume, of the capital stock of Parent, (viii) any failure by Parent or its Subsidiaries to meet internal, analysts' or other earnings estimates or financial projections or forecasts for any period, or any changes in credit ratings and any changes in any analysts recommendations or ratings with respect to Parent or any of its Subsidiaries; and (ix) any pending, initiated or threatened legal or administrative proceeding, claim, suit or action against Parent, any of its Subsidiaries or any of their respective officers or directors, in each case, arising out of or relating to the execution of this Agreement or the Transactions; provided that (1) any fact, change, event, circumstance, occurrence or effect set forth in clauses (a)(i) through (a)(iv) may be taken into account in determining whether there has been, is or would reasonably be expected to be a Company Material Adverse Effect to the extent (and only to the extent) such fact, change, event, circumstance, occurrence or effect has a disproportionate adverse effect on business, results of operations or financial condition of Parent and its Subsidiaries taken as a whole, in relation to other participants in the business and industries in which Parent and its Subsidiaries operate, and (2) the underlying cause of any failure referred to in clauses (a)(vii) and (a)(viii) may be taken into account in determining whether there has been, is or would reasonably be expected to be a Parent Material Adverse Effect; or (b) would, individually or in the aggregate, reasonably be expected to prevent or materially impede, interfere with, hinder or delay the consummation by Parent or Merger Sub of the Transactions.

- "Parent Permits" shall have the meaning set forth in Section 4.8(b).
- "Parent Preferred Stock" shall have the meaning set forth in Section 4.2.
- "Parent SEC Documents" shall have the meaning set forth in Section 4.5(a).
- "Parent Securities" shall have the meaning set forth in Section 4.2.
- "Parent Stock Plans" means Parent's 2005 Stock Incentive Plan.
- "Parent Subsidiary Securities" shall have the meaning set forth in Section 4.1(b).
- "Permitted Encumbrances" shall mean (a) zoning, building codes and other land use Laws regulating the use or occupancy of such real property or the activities conducted thereon which are imposed by any Governmental Authority having jurisdiction over such real property and (b) easements, rights-of-way, encroachments, restrictions, covenants, conditions and other similar Encumbrances that (i) are disclosed in the public records or (ii) would be set forth in a title policy, title report or survey with respect to the applicable real property and in each case, individually or in the aggregate, (A) are not substantial in character, amount or extent in relation to the applicable real property and (B) do not materially and adversely impact the current or contemplated use, utility or value of the applicable real property or otherwise materially and adversely impair the present or contemplated business operations at such location.
- "Permitted Liens" shall mean (a) statutory Liens for Taxes, assessments or other charges by Governmental Authorities not yet due and payable or the amount or validity of which is being contested in good faith and by appropriate proceedings and for which adequate reserves in accordance with GAAP have seen set aside on its books, (b) mechanics', materialmen's, carriers', workmen's, warehouseman's, repairmen's, landlords' and similar Liens granted or which arise in the ordinary course of business for amounts not overdue or being diligently

contested in good faith by appropriate proceedings and for which adequate reserves in accordance with GAAP have seen set aside on its books, (c) Liens reflected in the Company SEC Documents, (d) Liens arising under or in connection with applicable building and zoning Laws, codes, ordinances, and state and federal regulations which are not violated by the current use and operation of the Person's real property, (e) Liens which do not secure monetary liabilities of any Person and that, individually or in the aggregate, do not and would not materially detract from the value or marketability of any of the assets of the Person or materially interfere with the use thereof as currently used and (f) licenses of Intellectual Property.

- "Permitted Portfolio Acquisition" shall have the meaning set forth in Section 5.1(a)(iv).
- "Person" means an individual, corporation, limited liability company, partnership, association, trust, unincorporated organization, other entity or group (as defined in the Exchange Act), including a Governmental Authority.
  - "Proxy Statement/Prospectus" shall have the meaning set forth in Section 3.4.
- "Release" means any release, spill, emission, discharge, leaking, pumping, injection, deposit, disposal, dispersal, leaching or migration into the environment (including ambient air, surface water, groundwater and surface or subsurface strata) or into or out of any property, including the movement of Hazardous Materials through or in the soil, surface water or groundwater.
  - "Remedial Action" shall have the meaning set forth in Section 5.4(d).
- "Representatives" means, with respect to any Person, the advisors, attorneys, accountants, consultants or other representatives (acting in such capacity) retained by or on behalf of such Person or any of its controlled Affiliates, together with directors, officers and employees of such Person and its Subsidiaries.
  - "Restraints" shall have the meaning set forth in Section 6.1(c).
  - "Review Committee" shall have the meaning set forth in the recitals.
  - "Sarbanes-Oxley Act" shall have the meaning set forth in Section 3.5(a).
  - "SEC" means the U.S. Securities and Exchange Commission.
  - "Securities Act" shall have the meaning set forth in Section 3.1(b).
  - "Sherman Act" means the Sherman Antitrust Act of 1890.
  - "SP Notice Period" shall have the meaning set forth in Section 5.3(f)(ii).
  - "Stock Consideration" shall have the meaning set forth in Section 2.1(c)(ii).
  - "Stock Electing Company Share" shall have the meaning set forth in Section 2.1(c)(ii).
  - "Stock Election" shall have the meaning set forth in Section 2.1(c)(ii).

- "Subsidiary" when used with respect to any party, shall mean any corporation, limited liability company, partnership, association, trust, joint venture or other entity of which securities or other ownership interests (i) representing more than 50% of the equity and more than 50% of the ordinary voting power (or, in the case of a partnership, more than 50% of the general partnership interests) or (ii) having the right to vote for the election of a majority of the board of directors or any similar governing body of such corporation, partnership, limited liability company, association, trust, joint venture or other entity are, as of such date, owned by such party or one or more Subsidiaries of such party or by such party and one or more Subsidiaries of such party.
  - "Subsidiary Securities" shall have the meaning set forth in Section 3.1(b).
  - "Superior Proposal" shall have the meaning set forth in Section 5.3(i).
  - "Supporting Stockholder" has the meaning set forth in the recitals.
  - "Surviving Corporation" shall have the meaning set forth in Section 1.1.
  - "Takeover Proposal" shall have the meaning set forth in Section 5.3(h).
  - "Tax Returns" shall have the meaning set forth in Section 3.9(h).
  - "Taxes" shall have the meaning set forth in Section 3.9(h).
  - "Termination Fee" shall have the meaning set forth in Section 7.3(d).
  - "Transactions" shall have the meaning set forth in the recitals.
  - "Voting Agreement" shall have the meaning set forth in the recitals.
  - "WARN" shall have the meaning set forth in Section 3.20.
- "Willful Breach" means a material breach that is a direct consequence of an act knowingly undertaken by a member of the Company Board or a senior executive officer of the Company listed on Section 8.13(b) of the Company Disclosure Schedule, in each case, with the actual knowledge and intent that such act would cause a breach of any provision of this Agreement to which such concept applies.

# Section 8.14 Interpretation.

(a) When a reference is made in this Agreement to an Article, a Section, Exhibit or Schedule, such reference shall be to an Article of, a Section of, or an Exhibit or Schedule to, this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "include", "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation". The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular

provision of this Agreement. Unless the context otherwise requires, the word "or" is not exclusive and the words "will" and "will not" are expressions of command and not merely expressions of future intent or expectation. All terms defined in this Agreement shall have the defined meanings when used in any document made or delivered pursuant hereto unless otherwise defined therein. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neuter genders of such term. Any agreement, instrument or statute defined or referred to herein or in any agreement or instrument that is referred to herein means such agreement, instrument or statute as from time to time amended, modified or supplemented, including (in the case of agreements or instruments) by waiver or consent and (in the case of statutes) by succession of comparable successor statutes and references to all attachments thereto and instruments incorporated therein and the rules and regulations promulgated thereunder. References to a Person are also to its permitted assigns and successors.

- (b) The parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.
- (c) All capitalized terms not defined in the Company Disclosure Schedule or the Parent Disclosure Schedule shall have the meanings ascribed to them in this Agreement. The representations and warranties of Parent, Merger Sub and the Company are made and given, and the covenants are agreed to, subject to the disclosures and exceptions set forth in the Company Disclosure Schedule or the Parent Disclosure Schedule, as applicable. Disclosure of any matter in any Section of the Company Disclosure Schedule or Parent Disclosure Schedule shall be deemed to be disclosure of such matter with respect to any other Sections of the Company Disclosure Schedule or Parent Disclosure Schedule, respectively, to the extent it is readily apparent that such disclosure is applicable to such other Section. The inclusion of any item in the Company Disclosure Schedule or Parent Disclosure Schedule shall not be deemed to be an admission or evidence of materiality of such item, nor shall it establish any standard of materiality for any purpose whatsoever. No disclosure in the Company Disclosure Schedule or Parent Disclosure Schedule relating to any possible breach or violation of any Contract or Law shall be construed as an admission or indication that any such breach or violation exists or has actually occurred. In no event shall the listing of any matter in the Company Disclosure Schedule or the Parent Disclosure Schedule, as applicable, be deemed or interpreted to expand the scope the respective party's representations, warranties and/or covenants set forth in this Agreement. All attachments to the Company Disclosure Schedule and Parent Disclosure Schedule are incorporated by reference into the Company Disclosure Schedule or Parent Disclosure Schedule in which they are directly or indirectly referenced. The information contained in the Company Disclosure Schedule or Parent Disclosure Schedule is in all events provided subject to and on the terms of the Confidentiality Agreement as though it were Evaluation Material (as such term is defi

[ signature page follows ]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be duly executed and delivered as of the date first above written.

## ASSET ACCEPTANCE CAPITAL CORP.

By: /s/ Rion B. Needs

Name: Rion B. Needs

Title: Chief Executive Officer

PINNACLE SUB, INC.

By: /s/ Paul Grinberg

Name: Paul Grinberg

Title: Chief Financial Officer

ENCORE CAPITAL GROUP, INC.

By: /s/ Paul Grinberg

Name: Paul Grinberg

Title: Chief Financial Officer

[ Signature Page to Agreement and Plan of Merger ]

# **EXHIBIT A**

# CERTIFICATE OF INCORPORATION & BY-LAWS OF MERGER SUB

See attached.

[ Signature Page to Agreement and Plan of Merger ]

#### EXHIBIT A TO AGREEMENT AND PLAN OF MERGER

#### CERTIFICATE OF INCORPORATION

**OF** 

PINNACLE SUB, INC.

ARTICLE I NAME

The name of the corporation (which is hereinafter referred to as the "Corporation") is:

Pinnacle Sub, Inc.

#### ARTICLE II REGISTERED OFFICE AND AGENT

The address of the Corporation's registered office in the State of Delaware is 1675 S. State Street, Suite B, in the City of Dover, County of Kent, Delaware 19901. The name of the Corporation's registered agent for service of process at such address is Capitol Services, Inc.

#### ARTICLE III PURPOSE

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized and incorporated under the General Corporation Law of the State of Delaware (the "DGCL").

#### ARTICLE IV CAPITAL STOCK

The total number of shares of capital stock that the Corporation will have the authority to issue is one thousand (1,000) shares of common stock of the par value of \$0.01 per share.

#### ARTICLE V INCORPORATOR

The name of the incorporator is Bryan C. Wittman, whose mailing address is c/o Fulbright & Jaworski L.L.P., 300 Convent Street, Suite 2100, San Antonio, Texas 78205.

#### ARTICLE VI BOARD OF DIRECTORS

The business and affairs of the Corporation will be managed by or under the direction of a board of directors (the "Board of Directors"). The Board of Directors may exercise all such authority and powers of the Corporation and do all such lawful acts and things as are not by law or this Certificate of Incorporation directed or required to be exercised or done by the stockholders. Directors need not be elected by written ballot unless the bylaws of the Corporation otherwise provide.

# ARTICLE VII ELIMINATION OF CERTAIN LIABILITY; INDEMNIFICATION AND INSURANCE

SECTION 1. <u>Elimination of Certain Liability of Directors</u>. A director of the Corporation will not be personally liable to the Corporation or its stockholders for monetary damages for breach of fiduciary duty as a director, except for liability (a) for any breach of the director's duty of loyalty to the Corporation or its stockholders, (b) for any acts or omissions not in good faith or that involve intentional misconduct or a knowing violation of law, (c) under Section 174 of the DGCL or (d) for any transaction from which the director derived an improper personal benefit. If the DGCL is amended hereafter to authorize further elimination or limitation of the personal liability of directors, then the liability of a director of the Corporation, in addition to the limitation on personal liability provided herein, shall be limited to the fullest extent permitted by the DGCL as so amended. Any repeal or modification of this section shall be prospective only, and shall not adversely affect any limitation on the personal liability of a director of the Corporation existing at the time of such repeal or modification.

#### SECTION 2. Indemnification and Insurance.

(a) Right to Indemnification. Each person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (hereinafter a "proceeding"), by reason of the fact that the person is or was a director or officer of the Corporation, or, while a director or officer of the Corporation, is or was serving at the request of the Corporation (as defined in Section 145(i) of the DGCL) as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise (as defined in Section 145(i) of the DGCL), both as to action in such person's official capacity and as to action in another capacity while holding such office, shall be indemnified and held harmless by the Corporation against expenses (including attorneys' fees), judgments, fines (as defined in Section 145(i) of the DGCL) and amounts paid in settlement actually and reasonably incurred by the person in connection with such proceeding to the fullest extent authorized by the DGCL, as the same exists or may hereafter be amended (but, in the case of any such amendment, only to the extent that such amendment permits the Corporation to provide broader indemnification rights than said law permitted the Corporation to provide prior to such amendment); provided, however, that the Corporation shall indemnify any such person seeking indemnification in connection with a proceeding (or part thereof) initiated by such person only if such proceeding (or part thereof) was authorized by the Board of Directors. The indemnification granted pursuant to this SECTION 2 shall continue as to a

person who has ceased to be a director or officer of the Corporation and shall inure to the benefit of the heirs, executors and administrators of such a person. The right to indemnification conferred in this SECTION 2 shall be a contract right. The Corporation may, by action of the Board of Directors, provide indemnification to employees and agents of the Corporation with the same scope and effect as the foregoing indemnification of directors and officers. For purposes of this SECTION 2, references to "the Corporation" shall include, in addition to the resulting corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director, officer, employee or agent of such constituent corporation, or is or was serving at the request of such constituent corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under this SECTION 2 with respect to the resulting or surviving corporation as such person would have with respect to such constituent corporation if its separate existence had continued.

- (b) <u>Advancement of Expenses</u>. The Corporation shall pay expenses (including attorneys' fees) incurred by a director or officer in defending any proceeding in advance of the final disposition of such proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that such person is not entitled to be indemnified by the Corporation as authorized in this SECTION 2. Such expenses (including attorneys' fees) incurred by former directors and officers or other employees and agents of the Corporation may be so paid upon such terms and conditions, if any, as the Corporation, by action of the Board of Directors, may deem appropriate.
- (c) Non-Exclusivity of Rights. The indemnification and advancement of expenses provided by, or granted pursuant to, this SECTION 2 shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under any statute, bylaw, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in such person's official capacity and as to action in another capacity while holding such office.
- (d) <u>Insurance</u>. The Corporation shall have power to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against any liability asserted against such person and incurred by such person in any such capacity, or arising out of such person's status as such, whether or not the Corporation would have the power to indemnify such person against such liability under Section 145 of the DGCL.

# ARTICLE VIII AMENDMENT OF CERTIFICATE OF INCORPORATION

The Corporation reserves the right at any time from time to time to amend, alter, change or repeal any provision contained in this Certificate of Incorporation, and any other provisions authorized by the laws of the State of Delaware at the time in force may be added or inserted, in the manner now or hereafter prescribed by law. All rights, preferences and privileges of whatsoever nature conferred upon stockholders, directors or any other persons or entities

whomsoever by and pursuant to this Certificate of Incorporation in its present form or as hereafter amended are granted subject to the right reserved in this ARTICLE VIII.

#### ARTICLE IX BYLAWS

In furtherance and not in limitation of the powers conferred by law, and for the purpose of the orderly management of the business and the conduct of the affairs of the Corporation, the Board of Directors is expressly authorized and empowered to adopt, amend and repeal the bylaws of the Corporation from time to time at any regular or special meeting of the Board of Directors or by written consent, subject to the right of the stockholders of the Corporation to adopt, amend or repeal any bylaws of the Corporation.

THE UNDERSIGNED, being the incorporator hereinbefore named, for the purpose of forming a corporation pursuant to the DGCL, does make this Certificate, hereby declaring and certifying that this is my act and deed and the facts herein stated are true. Accordingly, I have hereunto set my hand this 4 th day of March, 2013.

Ву:	
	Bryan C. Wittman, Incorporator

#### **BYLAWS**

OF

#### P INNACLE S UB, I NC.

Adopted as of March 4, 2013

#### ARTICLE I OFFICES

Section 1. <u>Registered Office</u>. The registered office of the corporation will be located in the City of Dover, County of Kent, State of Delaware.

Section 2. Other Offices. The corporation may have offices at such other places both within and without the State of Delaware as the Board of Directors may from time to time determine or as the business of the corporation may require.

#### ARTICLE II MEETINGS OF THE STOCKHOLDERS

Section 1. <u>Place of Meetings</u>. All meetings of the stockholders for the election of directors or for any other proper purpose will be held at such place either within or without the State of Delaware as the Board of Directors may from time to time designate, as stated in the notice of such meeting or a duly executed waiver of notice thereof. The Board of Directors may, in its sole discretion, determine that any meeting not be held at any place, but may instead be held solely by means of remote communication pursuant to Section 211(a)(2) of the Delaware General Corporation Law.

Section 2. <u>Annual Meeting</u>. Unless directors are elected by written consent of stockholders in lieu of an annual meeting, an annual meeting of stockholders will be held at such date and time as may be designated by the Board of Directors and stated in the notice of the meeting. At such meeting, the stockholders entitled to vote thereat will elect a Board of Directors and may transact such other business as may be properly brought before the meeting.

Section 3. <u>Special Meetings</u>. Unless otherwise provided by law or the certificate of incorporation of the corporation (as amended or restated from time to time, the "Certificate of Incorporation"), special meetings of stockholders may be called by the Chief Executive Officer or the President, and will be called by the Secretary at the request, in writing, of either a majority of the Board of Directors or stockholders owning outstanding shares of capital stock of the corporation entitled to cast ten percent (10%) or more of the votes that could be cast at such meeting. Such request will state the purpose or purposes of the proposed meeting.

Section 4. <u>Notice of Annual or Special Meeting</u>. Unless otherwise required by law, written or printed notice stating the location, day and hour of the meeting and, in case of a special meeting, the purpose or purposes for which the meeting is called, must be delivered either personally or by mail, by or at the direction of the Secretary to each stockholder of record entitled to vote at such meeting, not less than ten (10) nor more than sixty (60) days before the date of the meeting, in accordance with Article VIII of these Bylaws.

Section 5. <u>Business at Special Meeting</u>. The business transacted at any special meeting of the stockholders will be limited to the purpose or purposes stated in the notice thereof.

Section 6. Quorum of Stockholders . The holders of outstanding shares of capital stock of the corporation entitled to cast a majority of the votes that could be cast at any meeting of stockholders, represented in person or by proxy, will constitute a quorum at such meeting for the transaction of business. When a separate vote by a class or classes is required, a majority of the outstanding shares of each such class or classes, represented in person or by proxy, will constitute a quorum entitled to take action with respect to that vote on that matter.

Section 7. Adjournment. At any meeting of stockholders, if less than a quorum is present, the holders of outstanding shares of capital stock of the corporation entitled to cast a majority of the votes that could be cast by the holders of capital stock then present in person or by proxy will have the power to adjourn the meeting from time to time without notice other than announcement at the meeting until a quorum is present. Any business may be transacted at the adjourned meeting that might have been transacted at the meeting originally noticed. If the adjournment is for more than thirty (30) days, or if after the adjournment a new record date (as provided for in Article IX, Section 5 of these Bylaws) is fixed for the adjourned meeting, a notice of the adjourned meeting will be given to each stockholder of record entitled to vote at the adjourned meeting.

Section 8. Act of Stockholders. Directors will be elected by a plurality of the votes cast by holders of shares of capital stock of the corporation outstanding and entitled to vote in the election of directors represented in person or by proxy at a meeting of stockholders at which a quorum is present, and all elections of directors must be by written ballot. Any other action or matter that is approved by a majority of the votes cast by holders of shares of capital stock of the corporation outstanding and entitled to vote on such action or matter, represented in person or by proxy at a meeting at which a quorum is present, will be the act of the stockholders, unless a higher proportion of such votes is required by law or the Certificate of Incorporation. When a separate vote by a class or classes is required, a majority of the outstanding shares of such class or classes and represented in person or by proxy, will be the act of such class.

Section 9. Proxies. At any meeting of the stockholders, each stockholder having the right to vote will be entitled to vote either in person or by proxy executed in writing by the stockholder or by his duly authorized attorney-in-fact. No proxy will be valid after one (1) year from its date of execution unless otherwise provided in the proxy. Each proxy will be revocable unless expressly provided therein to be irrevocable and the proxy is coupled with an interest sufficient in law to support an irrevocable proxy or otherwise made irrevocable by law. Proxies must be filed with the secretary of the meeting, or of any adjournment thereof, before being voted. Except as otherwise provided by a proxy, proxies will not be valid after final adjournment of such meeting. A proxy with respect to stock held in the name of two or more persons will be valid if executed by any one of them unless at or prior to its exercise the corporation receives a specific written notice to the contrary from any one of them. A proxy purporting to be executed by or on behalf of a stockholder will be deemed valid unless challenged at or prior to its exercise, and the burden of proving invalidity will rest on the challenger.

Section 10. <u>Voting List</u>. At least ten (10) days before each meeting of stockholders, a complete list of stockholders entitled to vote at each such meeting or in any adjournment thereof, arranged in alphabetical order, with the address of and the number of shares of each class of capital stock held by each, will be prepared by the Secretary or the officer or agent having charge of the stock transfer ledger of the corporation. Such list will be open to the examination of any stockholder, for any purpose germane to the meeting, during ordinary business hours, for such ten (10) day period at the principal executive office of the corporation. Such list will also be produced and kept open at the time and place of

the meeting. The stock ledger will be the only evidence as to who are the stockholders entitled to vote in person or by proxy at any meeting of stockholders.

Section 11. Consent of Stockholders in Lieu of Meeting. Any action that may be taken at a special or annual meeting of the stockholders may be taken without a meeting, without prior notice and without a vote if a consent in writing, setting forth the action so taken, is signed by all of the holders of outstanding shares having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted. Every written consent will bear the date of signatures of each stockholder and no written consent will be effective to take the corporate action referred to therein unless, within sixty (60) days after the earliest dated consent, written consents signed by a sufficient number of holders to take such action are delivered to the Secretary. Prompt notice of the taking of corporate action without a meeting by less than unanimous written consent will be given to those stockholders who did not consent in writing.

Section 12. <u>Presiding Officer and Conduct of Meetings</u>. The Chairman of the Board will preside at all meetings of the stockholders and will automatically serve as chairman of such meetings. In the absence of the Chairman of the Board, the Board of Directors may elect a person to act as chairman at such meeting. In the absence of the Chairman of the Board, and if the Board of Directors fails to elect a chairman, then the President will preside at the meeting of the stockholders and will automatically be the chairman of such meeting, unless and until a different person is elected by a majority of the shares entitled to vote at such meeting. The Secretary will act as secretary at all meetings of the stockholders. In the absence or disability of the Secretary, an Assistant Secretary (if one has been elected) appointed by the chairman of the meeting will act as secretary at such meetings and in the absence or disability of an Assistant Secretary, the chairman of the meeting will appoint another person to act as secretary at such meetings.

Section 13. <u>Inspectors</u>. The Board of Directors may, in advance of any meeting of stockholders, appoint one or more inspectors to act at such meeting or any adjournment thereof. If any of the inspectors so appointed fail to appear or act, or if inspectors have not been appointed, the chairman of the meeting must appoint one or more inspectors. The inspectors will determine the number of shares of capital stock of the corporation outstanding and the voting power of each, the number of votes represented at the meeting, the existence of a quorum, the validity and effect of proxies, and will receive votes, ballots or consents, hear and determine all challenges and questions arising in connection with the right to vote, count and tabulate all votes, ballots or consents, determine the results, and do such acts as are proper to conduct the election or vote with fairness to all stockholders. On request of the chairman of the meeting, the inspectors will make a report in writing of any challenge, request or matter determined by them and will execute a certificate of any fact found by them. No director or candidate for the office of director may act as an inspector of an election of directors.

## ARTICLE III BOARD OF DIRECTORS

Section 1. <u>Powers</u>. The business and affairs of the corporation will be managed by or under the direction of its Board of Directors. The Board of Directors may exercise all such powers of the corporation and do all such lawful acts and things as are not by law or by the Certificate of Incorporation or by these Bylaws directed or required to be exercised and done by the stockholders.

Section 2. <u>Compensation of Directors</u>. As specifically prescribed from time to time by resolution of the Board of Directors, directors may be paid their expenses of attendance at each meeting of the Board of Directors and may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated compensation for their services in their capacity as directors. This provision will not

preclude any director from serving the corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation for attending committee meetings.

- Section 3. <u>Chairman of the Board</u>. The Board of Directors, at its first meeting after each annual meeting of stockholders, may elect one of its members as Chairman of the Board. The Chairman of the Board will preside at all meetings of the Board of Directors and will have such other powers and duties as usually pertain to such position or as may be delegated by the Board of Directors.
- Section 4. <u>Vacancies and Newly Created Directorships</u>. Vacancies and newly created directorships resulting from any increase in the authorized number of directors may be filled by a majority of the directors then in office, although less than a quorum, and the directors so elected will hold office for the unexpired term of their predecessor in office until the next annual meeting and until their successors are elected and have qualified. A vacancy will be deemed to exist by reason of the death, resignation or removal from office of any director.
- Section 5. <u>Resignation; Removal</u>. Any director may resign at any time by giving written notice thereof to the Board of Directors. Any such resignation will take effect as of its date unless some other date is specified therein, in which event it will be effective as of that date. The acceptance of such resignation will not be necessary to make it effective. The holders of shares of capital stock of the corporation having the right to cast a majority of the votes that could be cast in an election of directors may remove any director or the entire Board of Directors, with or without cause, either by a vote at a special meeting or annual meeting, or by written consent.

# ARTICLE IV MEETINGS OF THE BOARD

- Section 1. <u>First Meeting</u>. The first meeting of each newly-elected Board of Directors will be held immediately following the annual meeting of the stockholders and no notice of such meeting will be necessary to the newly-elected directors in order legally to constitute the meeting, provided a quorum is present.
- Section 2. <u>Regular Meetings</u>. Regular meetings of the Board of Directors may be held with or without notice at such time and at such place either within or without the State of Delaware as from time to time is prescribed by the Board of Directors.
- Section 3. <u>Special Meetings</u>. Special meetings of the Board of Directors may be called by the Chairman of the Board, the Chief Executive Officer, the President or by a majority of the Board of Directors. Written notice of special meetings of the Board of Directors must be given to each director at least 24 hours before the time of the meeting.
- Section 4. <u>Business at Regular or Special Meeting</u>. Neither the business to be transacted at nor the purpose or purposes of any regular or special meeting of the Board of Directors need be specified in the notice or waiver of notice of such meeting.
- Section 5. <u>Quorum of Directors</u>. A majority of the Board of Directors will constitute a quorum for the transaction of business, unless a greater number is required by law or the Certificate of Incorporation. If a quorum is not present at any meeting of the Board of Directors, the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum is present.

Section 6. Act of Meeting of the Board of Directors. The act of a majority of the directors present at a meeting at which a quorum is present will be the act of the Board of Directors, unless the act of a greater number is required by law or the Certificate of Incorporation.

Section 7. <u>Action by Unanimous Written Consent Without a Meeting</u>. Any action required or permitted to be taken at a meeting of the Board of Directors may be taken without a meeting if a consent in writing setting forth the action so taken is signed by all members of the Board of Directors and such consent is filed with the minutes of proceedings of the Board of Directors. Such consent will have the same force and effect as a unanimous vote of the Board of Directors.

#### ARTICLE V COMMITTEES

The Board of Directors, by resolution adopted by a majority of the full Board of Directors, may designate from among its members an executive committee and one or more other committees, each of which, to the extent provided in such resolution, will have and may exercise all the authority of the Board of Directors, subject to the limitations imposed by applicable law. The Board of Directors, by resolution adopted by a majority of the full Board of Directors, may designate one or more of its members as alternate members of any committee, who may, subject to any limitations imposed by the Board of Directors, replace absent or disqualified members at any meeting of that committee. Vacancies in the membership of any such committee will be filled by resolution adopted by the majority of the full Board of Directors at a regular or special meeting of the Board. All committees will keep regular minutes of their proceedings and report the same to the Board of Directors when required. To the extent applicable, the provisions of Article IV of these Bylaws governing the meetings of the Board of Directors will likewise govern the meetings of any committee thereof. Any member of the executive committee or any other committee may be removed by the Board of Directors by the affirmative vote of a majority of the full Board of Directors, whenever, in its judgment, the best interests of the corporation will be served thereby.

#### ARTICLE VI DIRECTORS' ACTION BY USE OF CONFERENCE TELEPHONE

Members of the Board of Directors or members of any committee designated by such Board of Directors may participate in and hold a meeting of such Board of Directors or committee by conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and participation in such a meeting will constitute presence in person at such meeting, except when a person participates in the meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business on the ground that the meeting is not lawfully called or convened.

## ARTICLE VII OFFICERS

Section 1. <u>Executive Officers</u>. The officers of the corporation will consist of a President and a Secretary and may also include a Chief Executive Officer, one or more Vice Presidents, a Chief Financial Officer and such other officers as are provided for in this Article VII, each of whom will be elected by the Board of Directors as provided in Section 2 of this Article. Any two or more offices may be held by the same person.

Section 2. <u>Election and Qualification</u>. The Board of Directors, at its first meeting held immediately after each annual meeting of stockholders, will choose a President and a Secretary. The

Board of Directors also may elect a Chief Executive Officer, one or more Vice Presidents, a Chief Financial Officer and such other officers, including assistant officers and agents as may be deemed necessary, who will hold their offices for such terms and exercise such powers and perform such duties as are determined from time to time by the Board of Directors.

- Section 3. Salaries. The compensation of all officers and agents of the corporation will be determined by the Board of Directors.
- Section 4. <u>Term, Removal, and Vacancies</u>. Each officer of the corporation will hold office until his successor is chosen and qualified or until his death, resignation, or removal. Any officer may resign at any time upon giving written notice to the corporation. Any officer or agent may be removed by the Board of Directors, with or without cause, but such removal will be without prejudice to the contract rights, if any, of the person so removed. Election or appointment of an officer or agent will not of itself create contract rights. Any vacancy occurring in any office of the corporation by death, resignation, removal or otherwise will be filled by the Board of Directors.
- Section 5. <u>Chief Executive Officer</u>. The Chief Executive Officer will, subject to the oversight and the direction of the Board of Directors, be the chief executive officer of the corporation, and will see that all orders and resolutions of the Board of Directors are carried into effect. The Chief Executive Officer will have such other powers and duties as usually pertain to such office or as may be prescribed by the Board of Directors. Unless the Board of Directors shall elect different persons to be Chief Executive Officer and President, any person that is elected to be President shall also be deemed to have been elected to be Chief Executive Officer.
- Section 6. President. The President will, subject to the oversight and the direction of the Board of Directors and the Chief Executive Officer (if other than the President), have general powers of oversight, supervision and management of the business and affairs of the corporation, and will see that all orders and resolutions of the Board of Directors are carried into effect. The President will have such other powers and duties as usually pertain to such office or as may be prescribed by the Board of Directors. The President, or such other officer or agent of the corporation to which the President may delegate such power, will execute bonds, mortgages, instruments, contracts, agreements and other documentation. Unless the Board of Directors shall elect different persons to be Chief Executive Officer and President, any person that is elected to be Chief Executive Officer shall also be deemed to have been elected to be President.
- Section 7. <u>Vice Presidents</u>. Unless otherwise determined by the Board of Directors, the Vice Presidents, in the order of their seniority as such seniority may from time to time be designated by the Board of Directors, will perform the duties and exercise the powers of the President in the absence or disability of the President. They will perform such other duties and have such other powers as the Board of Directors or the President may from time to time prescribe.

Section 8. Secretary. The Secretary will attend all meetings of the stockholders and all meetings of the Board of Directors, will record all the proceedings of the meetings of the stockholders and of the Board of Directors in books to be kept for that purpose, and will perform like duties for the standing committees when required. The Secretary will give, or cause to be given, notice of all meetings of the stockholders and special meetings of the Board of Directors, and will perform such other duties as may be prescribed by the Board of Directors or the President. The Secretary will keep in safe custody the seal of the corporation, and, when authorized by the Board of Directors, affix the same to any instrument requiring it. When so affixed, such seal will be attested by his or her signature or by the signature of an Assistant Secretary.

Section 9. <u>Assistant Secretaries</u>. Unless otherwise determined by the Board of Directors, the Assistant Secretaries, in the order of their seniority as such seniority may from time to time be designated by the Board of Directors, will perform the duties and exercise the powers of the Secretary in the absence or disability of the Secretary. They will perform such other duties and have such other powers as the Board of Directors, the President or the Secretary may from time to time prescribe.

Section 10. <u>Chief Financial Officer</u>. The Chief Financial Officer will have the custody of the corporate funds and securities and will keep full and accurate accounts of receipts and disbursements in books belonging to the corporation and will deposit all moneys and other valuable effects in the name and to the credit of the corporation in such depositories as may be designated by the Board of Directors or the President. The Treasurer will disburse the funds of the corporation as may be ordered by the Board of Directors or the President, taking proper vouchers for such disbursements, and will perform such other duties and have such other powers as usually pertain to the office of treasurer or as may be prescribed by the Board of Directors or the President from time to time.

Section 11. <u>Assistant Treasurer</u>. Unless otherwise determined by the Board of Directors, the Assistant Treasurer will perform the duties and exercise the powers of the Chief Financial Officer in the absence or disability of the Chief Financial Officer. The Assistant Treasurer will perform such other duties and have such other powers as the Board of Directors, the President or the Chief Financial Officer may from time to time prescribe.

Section 12. Officers' Bond. If required by the Board of Directors, any officer will give the corporation a bond, which must be renewed as the Board of Directors may require, in such sum and with such surety or sureties as may be satisfactory to the Board of Directors for the faithful performance of the duties of his or her office and for the restoration to the corporation, in case of his or her death, resignation, retirement or removal from office, of any and all books, papers, vouchers, money and other property of whatever kind in his or her possession or under his or her control belonging to the corporation.

#### ARTICLE VIII NOTICES

Section 1. <u>Methods of Giving Notice</u>. Whenever any notice is required to be given to any stockholder or director under the provisions of any law, the Certificate of Incorporation or these Bylaws, it must be given in writing and delivered personally or mailed to such stockholder or director at such address as appears on the books of the corporation, and such notice will be deemed to be given two (2) business days after the day the same is deposited in the United States mail with sufficient postage thereon prepaid. To the extent provided in Section 4 of this Article VIII, notice to stockholders and directors may also be given by any form of communication, not directly involving the physical transmission of paper, that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process ("electronic transmission").

Section 2. <u>Waiver of Notice</u>. Whenever any notice is required to be given to any stockholder or director under the provisions of any law, the Certificate of Incorporation or these Bylaws, a waiver thereof in writing signed by the person or persons entitled to said notice, whether before or after the time stated therein, will be deemed equivalent to the giving of such notice.

Section 3. <u>Attendance as Waiver</u>. Attendance of a stockholder or director at a meeting will constitute a waiver of notice of such meeting, except when a stockholder or director attends a meeting for the express purpose of objecting, at the beginning of the meeting, to the transaction of any business on the

ground that the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, a meeting need be specified in any written waiver.

Section 4. <u>Electronic Transmission</u>. Without limiting the manner by which notice otherwise may be given effectively to stockholders and directors, any notice to stockholders or directors given by the corporation under any provision of the Delaware General Corporation Law (except as otherwise provided below), the Certificate of Incorporation or these Bylaws will be effective if given by a form of electronic transmission consented to by the stockholder or director to whom the notice is given. Any such consent will be revocable by the stockholder or director by written notice to the corporation. Any such consent will be deemed revoked if (a) the corporation is unable to deliver by electronic transmission two (2) consecutive notices given by the corporation in accordance with such consent and (b) such inability becomes known to the Secretary or an Assistant Secretary or to the transfer agent of the corporation, or other person responsible for the giving of notice; provided, however, the inadvertent failure to treat such inability as a revocation will not invalidate any meeting or other action. Notice given by electronic transmission will be deemed given: (1) if by facsimile telecommunication, when directed to a number at which the recipient has consented to receive notice; (2) if by electronic mail, when directed to an electronic mail address at which the recipient has consented to receive notice; (3) if by a posting on an electronic network together with separate notice to the recipient of such specific posting, upon the later of (A) such posting and (B) the giving of such separate notice; and (4) if by any other form of electronic transmission, when directed to the recipient. An affidavit of the Secretary or an Assistant Secretary or of the transfer agent or other agent of the corporation that the notice has been given by a form of electronic transmission will, in the absence of fraud, be prima facie evidence of the facts stated therein. Notice to stockholders may not be given by electronic transmission if such

# ARTICLE IX CERTIFICATES FOR SHARES

Section 1. Certificates Representing Shares. The corporation will deliver certificates representing all shares to which stockholders are entitled. Such certificates will be numbered, entered in the books of the corporation as they are issued and will be signed by the Chief Executive Officer, the President or a Vice President, and by the Secretary or an Assistant Secretary, and may be sealed with the seal of the corporation or a facsimile thereof. Any or all signatures on the certificate may be a facsimile. In case any officer who has signed or whose facsimile signature has been placed upon such certificate has ceased to be such officer before such certificate is issued, it may be issued by the corporation with the same effect as if he or she were such officer at the date of its issuance. If the corporation is authorized to issue shares of more than one class, there will be set forth upon the face or back of the certificate a statement that the corporation will furnish to any stockholder upon request and without charge a full statement of all of the powers, designations, preferences, limitations and relative, participating, optional, or other special rights of the shares of each class authorized to be issued and the qualifications, limitations or restrictions of such preferences and/or rights and, if the corporation is authorized to issue any preferred or special class in series, the variations in the relative rights and preferences between the shares of each such series so far as the same have been fixed and determined and the authority of the Board of Directors to fix and determine the relative rights and preferences of subsequent series. Each certificate representing shares must state upon the face thereof that the corporation is organized under the laws of the State of Delaware, the name of the person to whom issued, the number and the class and the designation of the series, if any, that such certificate represents and the par value of each share represented by such certificate or a statement that the share

Section 2. <u>Restriction on Transfer of Shares</u>. If any restriction on the transfer, or registration of the transfer, of shares is imposed or agreed to by the corporation, as permitted by law, the Certificate of Incorporation or these Bylaws, such restriction must be noted conspicuously on the face or back of each certificate representing shares in accordance with applicable law.

Section 3. <u>Transfer of Shares</u>. Subject to the provisions of Section 6 of this Article IX, upon surrender to the corporation or the transfer agent of the corporation of a certificate for shares duly endorsed or accompanied by proper evidence of succession, assignment, or authority to transfer, it will be the duty of the corporation to issue a new certificate to the person entitled thereto, cancel the old certificate, and record the transaction upon its books. The corporation will not, however, be required to pay any tax that may be payable in respect of any transfer of shares of capital stock of the corporation, and no such transfer will be made or reflected on the books of the corporation unless and until the person requesting such transfer has paid to the corporation the amount of any such tax, or has established, to the satisfaction of the corporation, that such tax has been paid.

Section 4. <u>Lost, Stolen, or Destroyed Certificate</u>. The corporation may issue a new certificate or certificates in place of any certificate or certificates theretofore issued by the corporation alleged to have been lost, stolen, or destroyed upon the making of an affidavit of that fact by the person claiming the certificate to be lost, stolen or destroyed. The corporation, as a condition precedent to the issuance thereof, may require the owner of such lost, stolen or destroyed certificates, or his legal representative, to advertise the same in such manner as it may require and/or to give the corporation a bond in such sum as it may direct to indemnify the corporation against any claim that may be made against the corporation with respect to the certificate alleged to have been lost, stolen or destroyed or the issuance of such new certificate.

Section 5. Fixing Record Date. For the purpose of determining stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, or entitled to receive payment of any dividend or other distribution, or in order to make a determination of stockholders for any other proper purpose, the Board of Directors may fix a date as the record date for any such determination of stockholders, such date to not precede the date of adoption of the resolution fixing the record date, and such date to be not more than sixty (60) days, and, in case of a meeting of stockholders, not less than ten (10) days, prior to the date on which the particular action requiring such determination of stockholders is to be taken. If no record date is fixed for the determination of stockholders entitled to notice of, or to vote at, a meeting of stockholders, the close of business on the day next preceding the date on which notice of the meeting is mailed or, if notice is waived, the close of business on the day next preceding the day on which the meeting is held will be the record date for such determination of stockholders. If no record date is fixed for the determination of stockholders entitled to receive payment of a dividend or other distribution, or for any other proper purpose, the close of business on the day on which the resolution of the Board of Directors declaring such dividend or relating to such other proper purpose is adopted will be the record date for such determination of stockholders entitled to vote at any meeting of stockholders has been made as provided in this Section 5, such determination will apply to any adjournment thereof; provided that the Board of Directors may fix a new record date for the adjourned meeting.

Section 6. <u>Registered Stockholders</u>. The corporation may recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends, and to vote such shares as owner, and will not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of any other person, whether or not it has express or other notice thereof, except as otherwise provided by law.

# ARTICLE X INDEMNIFICATION

The indemnification of directors, officers and other persons shall be as provided in the Certificate of Incorporation.

## ARTICLE XI GENERAL PROVISIONS

- Section 1. <u>Dividends</u>. The Board of Directors from time to time may declare, and the corporation may pay, dividends on its outstanding shares of capital stock in cash, property, or its own shares, pursuant to law.
- Section 2. <u>Reserves</u>. The Board of Directors may by resolution create a reserve or reserves out of earned surplus for any proper purpose or purposes, and may abolish any such reserve in the same manner.
- Section 3. <u>Negotiable Instruments</u>. All bills, notes, checks, or instruments for the payment of money will be signed by such officer or officers or such other person or persons as permitted by these Bylaws or in such manner as the Board of Directors from time to time may designate.
- Section 4. <u>Fiscal Year</u>. The fiscal year of the corporation will end on December 31 of each year unless a different fiscal year is fixed by resolution of the Board of Directors.
- Section 5. <u>Seal</u>. The corporate seal will have inscribed thereon the name of the corporation and may be used by causing it or a facsimile thereof to be impressed or affixed or in any other manner reproduced.
- Section 6. <u>Books and Records</u>. The corporation will keep books and records of account and will keep minutes of the proceedings of the stockholders, the Board of Directors, and each committee of the Board of Directors. The corporation will keep at its registered office or principal place of business, or at the office of its transfer agent or registrar, a record of the original issuance of shares issued by the corporation and a record of each transfer of those shares that have been presented to the corporation for registration of transfer. Such records will contain the names and addresses of all past and current stockholders of the corporation and the number and class of shares issued by the corporation held by each of them. Any books, records, minutes, and share transfer records may be in written form or in any other form capable of being converted into written form within a reasonable time.

#### ARTICLE XII AMENDMENTS

The Bylaws may be amended or repealed or new bylaws adopted as provided by the Certificate of Incorporation.

#### **EXECUTION VERSION**

#### **VOTING AGREEMENT**

This VOTING AGREEMENT (this "<u>Agreement</u>"), is dated as of March 6, 2013, by and among Encore Capital Group, Inc. ("<u>Parent</u>") and the stockholder of Asset Acceptance Capital Corp. (the "<u>Company</u>") listed on the signature page hereto (the "<u>Stockholder</u>"). Capitalized terms used herein and not otherwise defined herein shall have the meanings ascribed to such terms in the Merger Agreement as of the date hereof.

#### WITNESSETH:

WHEREAS, Parent, Pinnacle Sub, Inc., a Delaware corporation and wholly owned subsidiary of Parent ("Merger Sub"), and the Company entered into an Agreement and Plan of Merger, dated as of March 6, 2013 (the "Merger Agreement"), providing for, among other things and subject to the terms and conditions of the Merger Agreement, the merger of Merger Sub with and into the Company (the "Merger"), with the Company surviving the Merger.

WHEREAS, as of the date hereof, the Stockholder is the record and beneficial owner of the number of shares of Company Common Stock set forth on Exhibit A hereto (together with such additional shares of Company Common Stock which become beneficially owned (within the meaning of Rule 13d-3 promulgated under the Exchange Act) by the Stockholder, whether upon the exercise of options, conversion of convertible securities or otherwise, after the date hereof, but excluding any shares sold or transferred on or after the date hereof in compliance with Section 4.1 or Section 4.4, the "Owned Shares"), which shares set forth on Exhibit A collectively represent approximately 35.6% of the voting power of the outstanding Company Common Stock.

WHEREAS, as a condition to Parent's willingness to enter into and perform its obligations under the Merger Agreement, Parent has required that the Stockholder agree, and the Stockholder has agreed, to enter into this Agreement.

NOW, THEREFORE, in consideration of the premises and for other good and valuable consideration given to each party hereto, the receipt of which is hereby acknowledged, the parties agree as follows:

## 1. Agreement to Vote; Agreement for Cash Election.

1.1 Agreement to Vote. The Stockholder hereby agrees that, from the date hereof until the earlier of (i) the time that the Company Stockholder Approval has been obtained and (ii) termination of this Agreement in accordance with Section 5.1, at any meeting of the stockholders of the Company at which the approval and adoption of the Merger Agreement and the transactions contemplated thereby is to be voted upon, however called, or any adjournment or postponement thereof, the Stockholder shall be present (in person or by proxy) and vote (or cause to be voted), to the extent entitled to vote thereon, all of its Owned Shares at such time (a) in favor of (i) approval and adoption of the Merger Agreement and the transactions contemplated

thereby and (ii) any proposal to adjourn or postpone any meeting of the stockholders of the Company to a later date if there are not sufficient votes to approve the adoption of the Merger Agreement on the date on which such meeting is held and (b) against (A) any Takeover Proposal, (B) any merger agreement or merger (other than the Merger Agreement), consolidation, combination, business transaction, sale of assets, reorganization, recapitalization, dissolution, liquidation or winding up of the Company, and (C) any amendment of the Company's organizational documents that, in the case of (B) or (C), would reasonably be expected to impair the ability of Parent or Merger Sub to complete the Merger, or that would reasonably be expected to prevent, impede, interfere with, delay, or inhibit the timely consummation of the Merger or the fulfillment of Parent's, the Company's or Merger Sub's conditions under the Merger Agreement. Notwithstanding anything herein to the contrary, this Section 1.1 shall not require any Stockholder to be present (in person or by proxy) or vote (or cause to be voted) any of its Owned Shares to amend the Merger Agreement or take any action that could result in the amendment or modification, or a waiver of a provision therein, in any such case, in a manner that (x) decreases the amount or changes the form of the consideration or imposes any restrictions or additional conditions on the receipt of the consideration to the stockholders of the Company or (y) is otherwise materially adverse to the Stockholder. Notwithstanding anything herein to the contrary, in the event there is a Company Adverse Recommendation Change due to an Intervening Event, the Stockholder shall vote the Owned Shares for and against approval and adoption of the Merger Agreement in the same proportion as the shares of Company Common Stock owned by Persons other than the Stockholders vote such shares at the Company Stockholders Meeting.

- 1.2 <u>Agreement for Cash Election</u>. The Stockholder hereby agrees to make a Cash Election pursuant to the Merger Agreement for the entirety of the Owned Shares.
- 1.3 Other Voting Rights. Except as permitted by this Agreement, the Stockholder will continue to hold and shall have the right to exercise all voting rights related to the Stockholder's Owned Shares. For the avoidance of doubt, except as expressly set forth in Section 1.1, nothing in this Agreement shall limit the right of the Stockholder to vote in favor of, against, or abstain with respect to any matter presented to the Company's stockholders not in connection with a Takeover Proposal proposed by such third party.
- 2. <u>Representations and Warranties of Stockholder</u>. The Stockholder hereby represents and warrants to Parent as of the date of this Agreement and as of the date of the Company Stockholders Meeting as follows:
- 2.1 Power; Due Authorization; Binding Agreement. The Stockholder is duly organized, validly existing and in good standing under the laws of its jurisdiction of formation. The Stockholder has the requisite power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation by the Stockholder of the transactions contemplated hereby have been duly and validly authorized by all necessary corporate, partnership or other applicable action on the part of the Stockholder, and no other proceedings on the part of the Stockholder are necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by the Stockholder and, assuming the due and valid authorization, execution and delivery hereof by the other parties hereto, constitutes a valid and binding agreement of the Stockholder, enforceable against the Stockholder in accordance with its terms,

except that (i) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, relating to creditors' rights generally and (ii) equitable remedies of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the court before which any proceeding therefor may be brought.

- 2.2 Ownership of Shares. On the date hereof, the Owned Shares set forth opposite the Stockholder's name on Exhibit A hereto are owned of record and beneficially by the Stockholder. Other than restrictions in favor of Parent pursuant to this Agreement and except for such transfer restrictions of general applicability as may be provided under the Securities Act or the "blue sky" Laws of the various states of the United States, as of the date hereof the Stockholder has, and at any stockholder meeting of the Company held during the term of this Agreement to vote regarding approval and adoption of the Merger Agreement, the Stockholder will have (except as otherwise permitted by this Agreement, including in connection with the permitted Transfer of any Owned Shares), sole voting power and sole dispositive power with respect to the matters set forth in Section 1.1 in respect of all of the then Owned Shares of the Stockholder. There are no options, warrants or other rights, agreements, arrangements or commitments of any character to which the Stockholder is a party relating to the pledge, disposition or voting of any of the Owned Shares and there are no voting trusts, proxies or voting agreements with respect to the Owned Shares. Except as set forth on Exhibit A, the Stockholder does not beneficially own any shares of Company Common Stock, or any options, warrants or other rights to acquire any additional shares of Company Common Stock or any security exercisable for or convertible into shares of Company Common Stock. For the avoidance of doubt, Parent agrees that any shares, shares subject to options, restricted stock units and deferred stock units owned or held by Mt. T. Daniels and Mr. A. Ignaczak are not Owned Shares and are not subject to this Agreement.
- 2.3 No Conflict; Consents. None of the execution and delivery of this Agreement by the Stockholder, the consummation by the Stockholder of the transactions contemplated hereby or compliance by the Stockholder with any of the provisions hereof will conflict with or result in a breach, or constitute a default (with or without notice of lapse of time or both), under any provision of any organizational document, trust agreement, loan or credit agreement, note, bond, mortgage, indenture, lease or other agreement, instrument or Law applicable to the Stockholder or to the Stockholder's property or assets (including the Owned Shares), in each case, except for any conflict, breach or default that would not reasonably be expected to prevent or materially delay or otherwise impair the ability of the Stockholder to perform its obligations hereunder or to consummate the transactions contemplated hereby. No consent, approval or authorization of, or designation, declaration or filing with, any Governmental Authority or other Person on the part of the Stockholder is required in connection with the valid execution, delivery and performance of this Agreement, except for any of the foregoing that would not reasonably be expected to prevent or materially delay or otherwise impair the ability of the Stockholder to perform its obligations hereunder or to consummate the transactions contemplated hereby.
- 2.4. <u>No Litigation</u>. There is no suit, action, proceeding, judgment, order, decree, ruling, charge, or settlement pending or, to the knowledge of the Stockholder, threatened against the Stockholder or any of the Stockholder's properties or assets (including the Owned Shares) that would reasonably be expected to prevent or materially delay or otherwise impair the

ability of the Stockholder to perform its obligations hereunder or to consummate the transactions contemplated hereby.

- 2.5 <u>Acknowledgment</u>. The Stockholder has had the opportunity to review the Merger Agreement and this Agreement with counsel of the Stockholder's own choosing. The Stockholder understands and acknowledges that the Merger Agreement governs the terms of the Merger and the other matters specified therein. The Stockholder understands, acknowledges and agrees that Parent is entering into the Merger Agreement in reliance upon the Stockholder's execution, delivery and performance of this Agreement.
- 3. Representations and Warranties of Parent . Parent hereby represents and warrants to the Stockholder that Parent has the requisite corporate power and authority to execute and deliver this Agreement, to perform its obligations hereunder and to consummate the transactions contemplated hereby. The execution and delivery of this Agreement and the consummation by Parent of the transactions contemplated hereby have been duly and validly authorized by all necessary corporate action on the part of Parent, and no other proceedings on the part of Parent are necessary to authorize this Agreement or to consummate the transactions contemplated hereby. This Agreement has been duly and validly executed and delivered by Parent and, assuming the due and valid authorization, execution and delivery hereof by the other parties hereto, constitutes a valid and binding agreement of Parent, enforceable against Parent in accordance with its terms, except that (i) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, relating to creditors' rights generally and (ii) equitable remedies of specific performance and injunctive and other forms of equitable relief may be subject to equitable defenses and to the discretion of the court before which any proceeding therefor may be brought.

#### 4. Certain Covenants of the Stockholder.

4.1 Restriction on Transfer, Proxies and Non-Interference. The Stockholder hereby agrees, except as permitted by Section 4.4, from the date hereof until the termination of this Agreement in accordance with Section 5.1, not to (a) sell, transfer, pledge, encumber, assign or otherwise dispose of, or enter into any contract, option or other arrangement or understanding with respect to the sale, transfer, pledge, encumbrance, assignment or other disposition of, or limitation on the voting rights of, any of the Owned Shares of the Stockholder (any such action, a "Transfer"), other than to an Affiliate of the Stockholder if, as a precondition to such Transfer, such Affiliate agrees in a writing, reasonably satisfactory in form and substance to Parent, to be bound by all of the terms of this Agreement, (b) grant any proxies or powers of attorney with respect to the Owned Shares of the Stockholder, deposit any such Owned Shares into a voting trust or enter into a voting agreement with respect to any such Owned Shares, in each case with respect to any vote on the approval and adoption of the Merger Agreement or any other matters set forth in Section 1.1 of this Agreement, or (c) commit or agree to take any of the foregoing actions during the term of this Agreement. If any involuntary Transfer of any of the Owned Shares shall occur (including, but not limited to, a sale by a Stockholder's trustee in any bankruptcy, or a sale to a purchaser at any creditor's or court sale), the transferee (which term, as used herein, shall include any and all transferees and subsequent transferees of the initial transferee) shall, to the extent permitted by applicable Law, take and hold such Owned Shares subject to all of the restrictions, liabilities and rights under this Agreement, which shall continue in full force and effect until the valid termination of this Agreement. Any attempted Transfer of Owned Shares or any interest therein in violation of this Section 4.1 shall be null and void.

- 4.2 No Limitations on Actions. Parent expressly acknowledges that the Stockholder is entering into this Agreement solely in its capacity as the beneficial owner of the applicable Owned Shares and this Agreement shall not limit or otherwise affect the actions or fiduciary duties of any trustee, beneficiary, settlor, employee or designee of the Stockholder or any of its affiliates (collectively, "Affiliates") in its capacity, if applicable, as a director of the Company, including but not limited to a director of the Company taking any action in compliance with Section 5.3 of the Merger Agreement and making any statements (public or otherwise) in connection therewith. Parent shall not assert any claim that any action taken by any of Stockholder's Affiliates in its capacity as a director or officer of the Company violates any provision of this Agreement. For the avoidance of doubt, (i) nothing in this Agreement shall limit in any way the Company or its officers or directors from taking actions permitted or required by the Merger Agreement and (ii) Parent hereby agrees and acknowledges that the Company is not and will not be deemed to be an "Affiliate" of the Stockholder for the purposes of this Agreement.
- 4.3 <u>Further Assurances</u>. From time to time, at the reasonable request of Parent and without further consideration, the Stockholder shall use its commercially reasonable efforts to execute and deliver such additional documents and take all such further action as may be reasonably necessary to comply with its obligations under this Agreement.
- 4.4 <u>Permitted Transfers</u>. The Stockholder hereby agrees not to Transfer any Owned Shares other than to a Permitted Transferee. For purposes of this Agreement, "<u>Permitted Transferee</u>" means a person who has (i) executed a signature page to this Agreement pursuant to which such person agrees to be a "Stockholder" pursuant to this Agreement with respect to all Owned Shares proposed to be Transferred to such person and (ii) provided the requisite contact information for such person as contemplated by <u>Section 5.5</u> of this Agreement.
- 4.5 Appraisal Rights . The Stockholder hereby irrevocably and unconditionally (a) waives, and agrees not to assert or perfect, any and all rights that may arise with respect to the Merger or any of the transactions contemplated by the Merger Agreement to demand appraisal of any Owned Shares (including, without limitation, under Section 262 of the DGCL) or any rights that the Stockholder may have to dissent from the Merger and (b) agrees not to commence or join in, and agrees to take all actions necessary to opt out of, any class in any class action with respect to any claim, derivative or otherwise, against Parent, Merger Sub, the Company or any of their respective successors (i) challenging the validity of, or seeking to enjoin the operation of, any provision of this Agreement or (ii) alleging breach of any fiduciary duty of any Person in connection with the negotiation and entry into the Merger Agreement.
- 4.6 No Solicitation. The Stockholder (solely in its capacity as such) agrees that, during the term of this Agreement, the Stockholder will not, and will cause its directors, officers and employees, not to, and will instruct and use its reasonable best efforts to cause its other Affiliates, and its and their Representatives, not to, directly or indirectly (i) solicit, initiate, knowingly encourage or knowingly facilitate the making, submission or consummation of any Takeover Proposal or (ii) enter into, or otherwise participate in any discussions (except to notify such Person of the existence of the provisions of this Section 4.6) or negotiations regarding, or furnish to any Person any non-public information in connection with, any Takeover Proposal, or (iii) enter into any letter of intent, agreement in principle, merger agreement or other similar agreement with any person relating to an Takeover Proposal; provided that the foregoing shall not restrict the Stockholder from taking any of the foregoing actions specified in clauses (i)

through (iii) to the extent that the Company is permitted to take any such actions with such Person pursuant to Section 5.3(c) and Section 5.3(d) of the Merger Agreement. For the avoidance of doubt, nothing in this Section 4.6 will limit or affect any actions or omissions taken by any of Stockholder's Representatives in their capacity as a director or officer of the Company, including in exercising rights under the Merger Agreement (including, without limitation, under Section 5.3 thereof), and no such actions or omissions shall be deemed a breach of this Section 4.6.

- 4.7. Additional Shares. The Stockholder agrees that all shares of Company Common Stock that Stockholder purchases, acquires the right to vote or otherwise acquires beneficial ownership (as defined in Rule 13d-3 under the Exchange Act) of after the execution of this Agreement shall be subject to the terms of this Agreement and shall constitute Owned Shares for all purposes of this Agreement. The Stockholder shall notify Parent promptly in writing of any additional shares of Company Common Stock acquired by the Stockholder.
- 4.8. <u>Documentation and Information</u>. The Stockholder consents to and hereby authorizes Parent, Merger Sub and the Company to publish and disclose in all documents and schedules filed with the SEC, and any press release or other disclosure document that Parent, Merger Sub or the Company determines to be necessary in connection with the Merger and any transactions contemplated by the Merger Agreement, the Stockholder's identity and ownership of the Owned Shares, the existence of this Agreement and the nature of the Stockholder's commitments and obligations under this Agreement, and the Stockholder acknowledges that Parent, Merger Sub and the Company may, in their sole discretion, file this Agreement or a form hereof with the SEC or any other Governmental Authority.

#### 5. Miscellaneous.

- 5.1 <u>Termination of this Agreement</u>. This Agreement, and all obligations, terms and conditions contained herein, shall automatically terminate without any further action required by any person upon the earliest to occur of: (i) the termination of the Merger Agreement in accordance with its terms; (ii) the Effective Time; or (iii) the making of any change, by amendment, waiver or other modification to any provision of the Merger Agreement that (x) decreases the amount or changes the form of the consideration or imposes any restrictions or additional conditions on the receipt of the consideration to the stockholders of the Company or (y) is otherwise materially adverse to the Stockholder (either directly or indirectly through Parent or Merger Sub).
- 5.2 <u>Effect of Termination</u>. In the event of termination of this Agreement pursuant to <u>Section 5.1</u>, this Agreement shall become void and of no effect with no liability on the part of any party hereto; <u>provided</u>, <u>however</u>, no such termination shall relieve any party hereto from any liability for any intentional breach of this Agreement occurring prior to such termination and the provisions of this <u>Article 5</u>, including <u>Section 5.11</u>, shall survive any such termination.
- 5.3 Entire Agreement; Assignment. This Agreement constitutes the entire agreement among the parties with respect to the subject matter hereof and supersedes all other prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof. Other than as set forth in Section 5.4, nothing in this Agreement, express or implied, is intended to or shall confer upon any person other than the parties hereto any right,

benefit or remedy of any nature whatsoever under or by reason of this Agreement. Other than in connection with a Transfer to a Permitted Transferee in accordance with <u>Section 4.4</u>, this Agreement shall not be assigned by operation of law or otherwise and shall be binding upon and inure solely to the benefit of each party hereto.

- 5.4 Amendments; Third Party Beneficiary . This Agreement may not be modified, amended, altered or supplemented, except upon the execution and delivery of a written agreement executed by each of the parties hereto. The parties hereto expressly agree that the Company is intended to, and shall, be a third party beneficiary of the covenants and agreements of the parties hereto, which covenants and agreements shall not be amended, modified or waived without the prior written consent of the Company.
- 5.5 Notices. Any notice required to be given hereunder shall be sufficient if in writing, and sent by facsimile transmission (provided that any notice received by facsimile transmission or otherwise at the addressee's location on any Business Day after 5:00 p.m. (addressee's local time) shall be deemed to have been received at 9:00 a.m. (addressee's local time) on the next Business Day), by reliable overnight delivery service (with proof of service), hand delivery or certified or registered mail (return receipt requested and first-class postage prepaid), addressed as follows:

#### If to any Stockholder:

AAC Quad-C Investors LLC 230 East High Street Charlottesville, Virginia Attn.: Anthony R. Ignaczak Email: ari@qc-inc.com

#### with a copy to:

White & Case LLP 1155 Avenue of the Americas New York NY 10036 Attention: Matthew Kautz

Email: <u>mkautz@whitecase.com</u>

## If to the Company:

Asset Acceptance Capital Corp. 28405 Van Dyke Avenue Warren, Michigan 48093

Attn.: Rion Needs Reid E. Simpson

Email: <u>rneeds@assetacceptance.com</u> <u>rsimpson@assetacceptance.com</u>

with a copy to:

Kirkland & Ellis 601 Lexington Ave.

New York, New York 10022

Attn.: Michael Movsovich

Jeffrey Symons

Email: michael.movsovich@kirkland.com

jeffrey.symons@kirkland.com

#### If to Parent:

Encore Capital Group, Inc. 3111 Camino Del Rio North Suite 1300 San Diego, California Attn.: General Counsel

Fax: 858-309-1546

#### with a copy to:

Fulbright & Jaworski L.L.P. 300 Convent Street, Suite 2200 San Antonio, Texas 78205-3792 Attn.: Daryl Lansdale, Esq. Email: dlansdale@fulbright.com

or to such other address as any party shall specify by written notice so given, and such notice shall be deemed to have been delivered as of the date so telecommunicated, personally delivered or received. Any party to this Agreement may notify any other party of any changes to the address or any of the other details specified in this paragraph; <u>provided</u>, <u>however</u>, that such notification shall only be effective on the date specified in such notice or two (2) Business Days after the notice is given, whichever is later. Rejection or other refusal to accept or the inability to deliver because of changed address of which no notice was given shall be deemed to be receipt of the notice as of the date of such rejection, refusal or inability to deliver.

#### 5.6 Governing Law; Venue; Waiver of Jury Trial.

- (a) This Agreement and all disputes or controversies arising out of or relating to this Agreement or the transactions contemplated hereby shall be governed by, and construed in accordance with, the internal Laws of the State of Delaware, without regard to the Laws of any other jurisdiction that might be applied because of the conflicts of Laws principles of the State of Delaware.
- (b) Each of the parties irrevocably agrees that any legal action or proceeding arising out of or relating to this Agreement brought by any party against any other party shall be brought and determined in the Court of Chancery of the State of Delaware; <u>provided</u> that if jurisdiction is not then available in the Court of Chancery of the State of Delaware, then any such

legal action or proceeding may be brought in any federal court located in the State of Delaware. Each of the parties hereby irrevocably submits to the jurisdiction of the aforesaid courts for itself and with respect to its property, generally and unconditionally, with regard to any such action or proceeding arising out of or relating to this Agreement and the transactions contemplated hereby. Each of the parties agrees not to commence any action, suit or proceeding relating thereto except in the courts described above in Delaware, other than actions in any court of competent jurisdiction to enforce any judgment, decree or award rendered by any such court in Delaware. Each of the parties further agrees that notice as provided herein shall constitute sufficient service of process and the parties further waive any argument that such service is insufficient. Each of the parties hereby irrevocably and unconditionally waives, and agrees not to assert, by way of motion or as a defense, counterclaim or otherwise, in any action or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby, (a) any claim that it is not personally subject to the jurisdiction of the courts in Delaware as described herein for any reason, (b) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in any such court (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (c) that (i) the suit, action or proceeding in any such court is brought in an inconvenient forum, (ii) the venue of such suit, action or proceeding is improper or (iii) this Agreement, or the subject matter hereof, may not be enforced in or by any such court.

- (c) EACH OF THE PARTIES TO THIS AGREEMENT HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO A TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY. EACH PARTY MAKES THIS WAIVER VOLUNTARILY AND SUCH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS CONTAINED IN THIS SECTION 5.6(c).
- 5.7 Specific Performance; Exclusive Remedy. The parties agree that, in the event of any breach or threatened breach of any covenant or obligation contained in this Agreement, the parties would be irreparably harmed and that money damages would not provide an adequate remedy. Accordingly, each of the parties agrees that the parties to this Agreement shall be entitled to seek and obtain (a) a decree or order of specific performance to enforce the observance and performance of such covenant or obligation, and (b) an injunction restraining such breach or threatened breach. Each of the parties further agrees that (i) no party shall be required to obtain, furnish or post any bond or similar instrument in connection with or as a condition to obtaining any remedy referred to in this Section 5.7, and each party irrevocably waives any right it may have to require the obtaining, furnishing or posting of any such bond or similar instrument, and (ii) it will not oppose the granting of an injunction, specific performance and other equitable relief on the basis that the other party has an adequate remedy at law. Parent hereby agrees that specific performance or injunctive relief pursuant to this Section 5.7 shall be its sole and exclusive remedy with respect to breaches or threatened breaches by any Stockholder in connection with this Agreement, and neither Parent nor any of its Affiliates may pursue or accept any other form of relief (including monetary damages) that may be available for breach of this Agreement.
- 5.8 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same instrument and shall become effective when one or more counterparts have been signed by each of the parties and delivered to

the other parties. This Agreement may be executed by facsimile or electronic transmission signature and a facsimile or electronic transmission signature shall constitute an original for all purposes.

- 5.9 <u>Descriptive Headings</u>. The descriptive headings used herein are inserted for convenience of reference only and are not intended to be part of or to affect the meaning or interpretation of this Agreement.
- 5.10 Severability. Any term or provision of this Agreement that is invalid or unenforceable in any situation in any jurisdiction shall not affect the validity or enforceability of the remaining terms and provisions hereof or the validity or enforceability of the offending term or provision in any other situation or in any other jurisdiction. If the final judgment of a court of competent jurisdiction declares that any term or provision hereof is invalid or unenforceable, the parties hereto agree that the court making such determination shall have the power to limit the term or provision, to delete specific words or phrases, or to replace any invalid or unenforceable term or provision with a term or provision that is valid and enforceable and that comes closest to expressing the intention of the invalid or unenforceable term or provision, and this Agreement shall be enforceable as so modified. In the event such court does not exercise the power granted to it in the prior sentence, the parties hereto agree to replace such invalid or unenforceable term or provision with a valid and enforceable term or provision that will achieve, to the extent possible, the economic, business and other purposes of such invalid or unenforceable term.

## 5.11 Non-Recourse.

- (a) No past, present or future director, officer, employee, incorporator, member, partner, stockholder, trustee, beneficiary, settlor, agent, attorney, representative or affiliate of any party hereto or of any of their respective affiliates shall have any liability (whether in contract or in tort) for any obligations or liabilities of such party arising under, in connection with or related to this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby; <u>provided</u>, <u>however</u>, that nothing in this <u>Section 5.11</u> shall limit any liability of the parties hereto for breaches of the terms and conditions of this Agreement.
- (b) Nothing contained herein, and no action taken by any Stockholder pursuant hereto, shall be deemed to constitute the parties as a partnership, an association, a joint venture or any other kind of entity, or create a presumption that the parties are in any way acting in concert or as a group with respect to the obligations or the transactions contemplated by this Agreement.
- 5.12 <u>Fees and Expenses</u>. All costs and expenses incurred in connection with this Agreement shall be paid by the party incurring such expenses.
- 5.13 No Ownership Interest. Nothing contained in this Agreement shall be deemed to vest in Parent any direct or indirect ownership or incidence of ownership of or with respect to any Owned Shares. All rights, ownership and economic benefits of and relating to the Owned Shares shall remain vested in and belong to the Stockholder, and Parent shall have no authority to direct the Stockholder in the voting or disposition of any of the Owned Shares, except as otherwise provided herein.
- 5.14 <u>Agreement Negotiated</u>. The form of this Agreement has been negotiated by or on behalf of Parent and the Stockholder, each of which was represented by attorneys who

have carefully negotiated the provisions hereof. No Law or rule relating to the construction or interpretation of contracts against the drafter of any particular clause should be applied with respect to this Agreement.

[ remainder of page intentionally blank ]

IN WITNESS WHEREOF, the parties hereto have caused this Voting Agreement to be duly executed as of the day and year first above written.

# ENCORE CAPITAL GROUP, INC.

By: /s/ Paul Grinberg
Name: Paul Grinberg

Title: Chief Financial Officer

## **STOCKHOLDER:**

AAC QUAD-C INVESTORS LLC

By: /s/ Thad M. Jones
Name: Thad M. Jones
Title: Vice President

SIGNATURE PAGE TO VOTING AGREEMENT

# EXHIBIT A

# STOCKHOLDER STOCK OWNERSHIP

Number of Common

Stockholder	Shares Owned
AAC Quad-C Investors LLC	10,932,051

## **Encore Capital Group Announces Acquisition of Asset Acceptance**

- Combination Will Lead to Greater Operational Efficiencies and Shareholder Value -- Conference call scheduled for 9:00 a.m. Eastern -

SAN DIEGO and WARREN, March 6, 2013 — Encore Capital Group, Inc. (Nasdaq: ECPG) and Asset Acceptance Capital Corp. (Nasdaq: AACC) today announced that Encore has entered into an agreement to acquire Asset Acceptance Capital Corp. Based in Warren, Michigan, Asset Acceptance has decades of experience in buying consumer receivables across a broad range of asset classes, as well as a tenured workforce. Combined, Encore and Asset Acceptance have purchased over 60 million individual consumer accounts, including credit card, telecommunications, consumer loans and other related assets, with a face value of over \$130 billion.

Under the terms of the agreement, Encore will acquire Asset Acceptance for \$6.50 per share, which represents a total equity value of approximately \$200 million, representing a 24 percent premium to AACC's 30 day volume weighted average share price. Asset Acceptance shareholders will have the option to receive their consideration in cash or Encore stock or any combination of cash and Encore stock, at their election, with the aggregate stock consideration across all stockholders capped at 25 percent of the total equity consideration to be received. Subject to regulatory approval and Asset Acceptance shareholder approval, the acquisition is expected to close during the second quarter of this year.

"This acquisition moves our industry into a new phase of maturity defined by more efficient companies that are committed to operating ethically and treating consumers with respect," said Encore's President and Chief Executive Officer Brandon Black. "Encore's strong operating and cost advantages will allow Asset Acceptance's investments to be significantly more profitable and will deliver greater value to shareholders. Consumers also stand to benefit as we extend Encore's industry-leading Consumer Bill of Rights to millions more people who are on their way to financial recovery."

Encore has been acquiring competitor portfolios since 2001 and has developed expertise in valuing, managing and integrating large transactions. Encore made two significant portfolio acquisitions last year alone and has already successfully integrated them into its operational platform. The acquisition of Asset Acceptance provides Encore not only with an additional portfolio, but also with valuable operations capabilities and synergy opportunities.

"We believe Encore is a strong strategic fit for Asset as the transaction will enable us to maximize the full value of both our investments and our talented workforce," said Rion Needs, President and Chief Executive Officer of Asset

Acceptance. "The combination of our two companies will better position us for success in a rapidly changing and competitive marketplace."

"Our industry is highly fragmented, and we have long been anticipating consolidation as the regulatory and business climate grows more complex," said Mr. Black. "This transaction clearly signals that Encore will play a key role in driving this phase in our industry's development. Transactions like this are a component of our long-term growth plan, along with pursuing growth in new asset classes and geographies."

Encore and Asset Acceptance were advised respectively by Morgan Stanley and William Blair.

#### Transaction conference call

Encore also announced that it will host a conference call today at 9:00 a.m. ET to discuss the announcement. Members of the public are invited to listen to the event via a telephone conference call line or the Internet. To access the live listen-only telephone conference call, please dial (877) 670-9781 (domestic) or (408) 940-3818 (international). To access the live webcast via the Internet, log on at the Investor Relations page of Encore's website at <a href="https://www.encorecapital.com">www.encorecapital.com</a>

For those who cannot listen to the live broadcast, a telephonic replay will be available for seven days by dialing (800) 585-8367 or (404) 537-3406 and entering the conference number 19118734. A replay of the conference call will also be available shortly after the call on Encore's website.

#### About Encore Capital Group, Inc.

Encore Capital Group is a leading provider of debt management and recovery solutions for consumers and property owners across a broad range of assets. Through its subsidiaries, Encore Capital Group purchases portfolios of consumer receivables from major banks, credit unions, and utility providers, and partners with individuals as they repay their obligations and work toward financial recovery. Through its Propel Financial Services, LLC subsidiary, Encore Capital Group assists property owners who are delinquent on their property taxes by structuring affordable monthly payment plans.

Headquartered in San Diego, Encore Capital Group is a publicly traded NASDAQ Global Select company (ticker symbol: ECPG) and a component stock of the Russell 2000, the S&P SmallCap 600, and the Wilshire 4500. More information about Encore Capital Group can be found at www.encorecapital.com. Encore Capital Group's website and the information contained therein, is not incorporated into and is not a part of this press release.

#### **About Asset Acceptance Capital Corp.**

For 50 years, Asset Acceptance has provided credit originators, such as credit card issuers, consumer finance companies, retail merchants, utilities and others

an efficient alternative in recovering defaulted consumer debt. For more information, please visit www.AssetAcceptance.com

## **Forward Looking Statements**

The statements in this press release that are not historical facts, including, most importantly, those statements preceded by, or that include, the words "may," "believe," "projects," "expects," "anticipates" or the negation thereof, or similar expressions, may constitute "forward-looking statements" within the meaning of the Private Securities Litigation Reform Act of 1995 (the "Reform Act"). These statements may include, but are not limited to, statements regarding our future operating results, performance, business plans or prospects. For all "forward-looking statements," the Company claims the protection of the safe harbor for forward-looking statements contained in the Reform Act. Such forwardlooking statements involve risks, uncertainties, assumptions and other factors which may cause actual results, performance or achievements to be materially different from any future results, performance or achievements expressed or implied by such forward-looking statements. Encore and Asset Acceptance caution readers that any forward-looking statement is not a guarantee of future performance and that actual results could differ materially from those contained in the forward-looking statement. Readers should carefully consider the risks and uncertainties and other factors that may affect future results of the combined company described in the section entitled "Risk Factors" in the proxy statement/prospectus to be delivered to Encore's and Asset Acceptance's respective shareholders, and in Encore's and Asset Acceptance's respective filings with the SEC that are available on the SEC's web site located at www.sec.gov, including the sections entitled "Risk Factors" in Encore's and Asset Acceptance's most recent reports on Forms 10-K, 10-Q and 8-K, each as it may be amended from time to time. Readers should not place undue reliance on forward-looking statements, which speak only as of the date of this press release. Except for any obligations to disclose material information under the Federal securities laws, Encore undertakes no obligation to publicly update any forward-looking statements to reflect events or circumstances after the date of this press release.

#### Additional Information and Where to Find It

This communication does not constitute an offer to sell or the solicitation of an offer to buy any securities or a solicitation of any vote or approval. In connection with the proposed transaction, Encore and Asset Acceptance will file with the SEC a registration statement on Form S-4 that will include a proxy statement of Asset Acceptance and a prospectus of Encore, as well as other relevant documents concerning the proposed transaction. STOCKHOLDERS OF ASSET ACCEPTANCE ARE URGED TO READ THE REGISTRATION STATEMENT AND THE PROXY STATEMENT/PROSPECTUS REGARDING THE MERGER WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS

# FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT ENCORE, ASSET ACCEPTANCE AND THE PROPOSED TRANSACTION.

A free copy of the proxy statement/prospectus, as well as other filings containing information about Encore and Asset Acceptance, may be obtained at the SEC's internet site (http://www.sec.gov). You will also be able to obtain these documents, free of charge, from Encore at www.encorecapital.com under the tab "Investors," and then under the heading "SEC Filings" or from Asset Acceptance at www.assetacceptance.com under the tab "Investors," and then under the heading "SEC Filings."

Encore, Asset Acceptance and their respective directors and executive officers may be deemed to be participants in the solicitation of proxies in respect of the proposed transaction. Information about Encore directors and executive officers is available in Encore's proxy statement filed with the SEC on April 27, 2012. Information about Asset Acceptance's directors and executive officers is available in Asset Acceptance's proxy statement filed with the SEC on March 28, 2012. Other information regarding the participants in the proxy solicitation to be filed with the SEC and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the proxy statement/prospectus and other relevant materials to be filed with the SEC regarding the merger. Investors should read the proxy statement/prospectus carefully when it becomes available before making any voting or investment decisions.

This document shall not constitute an offer to sell or the solicitation of an offer to buy or subscribe for any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the U.S. Securities Act of 1933, as amended.

#### **Contacts:**

## **Encore Investor Relations**

Adam Sragovicz Tel 858-309-9509 adam.sragovicz@encorecapital.com

or

Paul Grinberg Tel 858-309-6904

# paul.grinberg@encorecapital.com

# **Encore Media Relations**

Katie Lilley, Hillenby Tel 703-722-3061 katielilley@hillenby.com

# **Asset Acceptance Investor Relations**

Mary Arraf Tel 586-983-7087 marraf@assetacceptance.com



#### Additional Information and Where to Find It

This communication does not constitute an offer to sell or the solicitation of an offer to buy any securities or a solicitation of any vote or approval. In connection with the proposed transaction, Encore and Asset Acceptance will file with the SEC a registration statement on Form S-4 that will include a proxy statement of Asset Acceptance and a prospectus of Encore, as well as other relevant documents concerning the proposed transaction. STOCKHOLDERS OF ASSET ACCEPTANCE ARE URGED TO READ THE REGISTRATION STATEMENT AND THE PROXY STATEMENT/PROSPECTUS REGARDING THE MERGER WHEN IT BECOMES AVAILABLE AND ANY OTHER RELEVANT DOCUMENTS FILED WITH THE SEC, AS WELL AS ANY AMENDMENTS OR SUPPLEMENTS TO THOSE DOCUMENTS, BECAUSE THEY WILL CONTAIN IMPORTANT INFORMATION ABOUT ENCORE, ASSET ACCEPTANCE AND THE PROPOSED TRANSACTION.

A free copy of the proxy statement/prospectus, as well as other filings containing information about Encore and Asset Acceptance, may be obtained at the SEC's internet site (<a href="http://www.sec.gov">http://www.sec.gov</a>). You will also be able to obtain these documents, free of charge, from Encore at <a href="http://www.sec.gov">www.encorecapital.com</a> under the tab "Investors," and then under the heading "SEC Filings" or from Asset Acceptance at <a href="http://www.assetacceptance.com">www.assetacceptance.com</a> under the tab "Investors," and then under the heading "SEC Filings."

Encore, Asset Acceptance and their respective directors and executive officers may be deemed to be participants in the solicitation of proxies in respect of the proposed transaction. Information about Encore directors and executive officers is available in Encore's proxy statement filed with the SEC on April 27, 2012. Information about Asset Acceptance's directors and executive officers is available in Asset Acceptance's proxy statement filed with the SEC on March 28, 2012. Other information regarding the participants in the proxy solicitation and a description of their direct and indirect interests, by security holdings or otherwise, will be contained in the proxy statement/prospectus and other relevant materials to be filed with the SEC regarding the merger when they become available. Investors should read the proxy statement/prospectus carefully when it becomes available before making any voting or investment decisions.

This document shall not constitute an offer to sell or the solicitation of an offer to buy any securities, nor shall there be any sale of securities in any jurisdiction in which such offer, solicitation or sale would be unlawful prior to registration or qualification under the securities laws of any such jurisdiction. No offering of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the U.S. Securities Act of 1933, as amended.



## Adam Sragovicz, Director, Finance & Treasury

Thank you, [Operator]. Good morning and welcome to Encore Capital Group's special transaction announcement call. With me on the call today is Brandon Black, our President and Chief Executive Officer and Paul Grinberg, our Chief Financial Officer. Brandon will make prepared remarks, and then we will be happy to take your questions.

Before we begin, we have a few housekeeping items. Throughout the call, we will use forward-looking statements, including predictions, expectations, estimates, or other information that might be considered forward-looking. While these forward-looking statements represent our current judgment, these statements are also subject to risks and uncertainties that may cause actual results to differ materially from statements being made today. As a result, we caution you against placing undue reliance on these forward-looking statements, which speak only as of the date they are made. Please be sure to see our Forms 10-K, 10-Q, and other SEC filings, including a press release issued as an exhibit to our Current Report on Form 8-K filed today, which include a more complete discussion of our financial results and risk factors that pertain to our business. In addition, this communication does not constitute an offer to sell or the solicitation of an offer to buy any securities.

As a reminder, this conference call will also be made available for replay on the Investors section of our website, and we also plan to post the prepared remarks following the conclusion of this call.

With that, let me turn the call over to Brandon Black, our President and Chief Executive Officer.

Page 2 of 4



#### Brandon Black - President and CEO:

Thank you, Adam, and good morning everyone. I appreciate you joining us on such short notice.

Earlier today, we announced that Encore has reached an agreement to acquire Asset Acceptance Capital Corp. Together, the two companies have purchased over 60 million consumer accounts with a face value of more than 130 billion dollars.

Under the terms of the agreement, Encore will acquire Asset Acceptance for \$6.50 per share, which represents a total equity value of approximately 200 million dollars. Asset Acceptance shareholders will have the option to receive their consideration in cash or Encore stock or any combination of cash and stock, at their election, with the aggregate stock consideration across all shareholders capped at 25 percent of the total equity consideration to be received. Encore anticipates funding the equity purchase and additional assumed liabilities through Encore's current credit facility and existing banking relationships. Subject to shareholder and regulatory approval, the acquisition is expected to close during the second quarter of this year.

We believe this acquisition marks the beginning of a new phase in our highly fragmented industry – a phase defined by more efficient companies committed to operating ethically and treating consumers with respect. Transactions like this are an important component of our long-term growth plan, along with pursuing growth in new asset classes and geographies.

As we have consistently said, success in our industry depends on three factors: an efficient operating platform, access to low-cost capital, and a deep understanding of the financially distressed consumer. We have also said that companies that lack these factors will find it difficult to survive in this increasingly competitive environment, particularly as pricing continues to rise. This is the case with Asset as a standalone company. As we integrate their operations, we believe that our strong operating and



cost advantages will allow the portfolio investments made by Asset to be significantly more profitable.

We have spoken a number of times about our expectation of industry consolidation as the business and regulatory environments grow more complex. For the past few years, we have actively positioned ourselves for this environment, and we believe this acquisition sends a clear signal that Encore will play a key role in driving consolidation in a highly fragmented industry. We have a history of acquiring and managing large portfolios in the resale space. For example, last year alone we made two significant portfolio acquisitions and have successfully integrated them into our business. We expect to leverage this experience as we work to integrate Asset into Encore's operations.

We believe Asset's portfolios have around one billion dollars in estimated remaining collections and, as a result, this transaction will account for a significant portion of our purchasing activity in 2013. As we have experienced during complicated transactions in the past, integration and absorption can take time and attention, so we will be devoting a lot of time to ensure that we maximize the value of Asset's portfolio. The purchase price of the portfolio represents attractive IRRs when compared to opportunities we currently see in the marketplace.

Before we open the line for questions, I'd like to thank Asset's management team for their efforts in bringing this transaction together. We not only believe that it's the right thing to do for our combined shareholders and employees, we also believe it's the right thing to do for consumers. It allows us to extend to millions more consumers the industry's only Consumer Bill of Rights, which is our commitment to treat people with the respect they deserve as they regain their financial health.

Before we open up the line for questions, I'd like to let you know that I am at Asset's headquarters in Warren, Michigan while Paul is at Encore's offices in San Diego. We'll try to be as coordinated as we can be in answering your questions. Operator, please open the line for questions.